

Scuola Internazionale Superiore di Studi Avanzati

Mathematics Area - PhD course in Mathematical Analysis, Modelling, and Applications

A general splitting principle on the non-smooth setting and applications

Candidate: Fabio Marconi Advisor: Prof. Nicola Gigli

Academic Year 2022-23



Contents

\mathbf{In}	trod	uction	5	
	Hist	History and motivation of RCD spaces		
	Splitting-type theorems			
	Gen	eral splitting principle and applications	10	
	Plar	n of the work	13	
1	Preliminaries			
	1.1	Notation on metric measure spaces	17	
	1.2	Sobolev space on metric measure spaces	20	
	1.3	Normed modules and first order differential calculus	24	
		1.3.1 Cotangent and tangent modules	26	
		1.3.2 Divergence and Laplace operators	28	
	1.4	Pullback of 1-forms and speed of a test plan	30	
	1.5	RCD spaces	32	
		1.5.1 Wasserstein distance and CD conditions	33	
		1.5.2 Convergence of (pointed) metric measure spaces and		
		stability \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	40	
	1.6	Second order differential calculus	41	
	1.7	Warped product spaces	45	
	1.8	Regular Lagrangian Flows	48	
2	A g	eneral splitting principle on RCD spaces	55	
	2.1	Producing a coordinate function	55	
	2.2	Set up and statement of the splitting result	57	
	2.3	Behaviour of the measure under the flow	60	
	2.4	Behaviour of the distance under the flow	64	
	2.5	Isometry and RCD condition of the quotient space \ldots \ldots \ldots	70	
3	Applications to spaces with positive spectrum			
	3.1	Preliminaries: harmonic functions and		
		Bochner inequality	75	

3.2	Prelim	inaries: sets of finite perimeter and Gauss-Green formula 79	
	3.2.1	Capacity and quasi-continuous functions 80	
	3.2.2	$L^0(\text{Cap})$ -modules and quasi-continuous vector	
		fields	
	3.2.3	Tangent module over ∂E and Gauss-Green formula 83	
3.3	Volum	e of the ends and harmonic functions	
3.4	Rigidi	ty theorem for spaces with positive spectrum $\ldots \ldots 97$	
	3.4.1	Properties of the bounded harmonic function 97	
	3.4.2	$ \mathbf{d}u $ as a function of u	
	3.4.3	Proof of Theorem 3.4.1	
3.5	Rigidity theorem for spaces with maximal spectrum 10		
	3.5.1	Isomorphism with the warped product	
	3.5.2	RCD condition of the quotient space	

Bibliography

Introduction

History and motivation of RCD spaces

In the last decades there were several attempts to generalise Riemannian manifolds with curvature bounds: the main reason is to find a set of structures which contains Riemannian manifolds with curvature bounded from one side and is compact with respect to a reasonable convergence, the Gromov-Hausdorff one, which is induced by a distance between metric spaces which measures "how far two spaces are from being isomorphic".

The concept of metric space was enough to give a synthetic concept of space with sectional curvature bounded from one side via Toponogov comparison theorem (see [BBI] and its references for an introduction about the subject).

Theorem 0.0.1 (Toponogov). Let $k \in \mathbb{R}$ and let M be a complete, connected Riemannian manifold. Then the following are equivalent:

- i) the manifold has sectional curvature always greater than or equal to k;
- ii) indicating with \mathbb{M}^k the simply connected model space with constant sectional curvature k, for every $p \in M$ and $\gamma : [0,1] \to M$ geodesic the following holds: let $\tilde{p}, \tilde{\gamma}_0, \tilde{\gamma}_1 \in \mathbb{M}^k$ be such that the three edges of the triangle $\tilde{p}\tilde{\gamma}_0\tilde{\gamma}_1$ are equal to the ones of $p\gamma_0\gamma_1$. Then $\mathsf{d}_{\mathrm{M}}(p,\gamma_t) \geq \mathsf{d}_{\mathbb{M}^k}(\tilde{p},\tilde{\gamma}_t)$ for every $t \in [0,1]$.

Condition (ii) is purely metric and can be written in geodesic metric spaces. We can define then "metric spaces with sectional curvature bounded from below by k" (also known as Alexandrov spaces and denoted by $\mathsf{CBB}(k)$) as the spaces for which condition (ii) holds.

In trying to approach in a similar way manifolds with a lower bound on the Ricci curvature with essential stability properties we note that working with metric spaces is not enough: taking the limit (in the Gromov-Hausdorff sense) of a sequence of manifolds with Ricci curvature greater than K seen as metric spaces might fail to have Ricci curvature greater than K. This first problem is solved taking metric measure spaces and considering a variant of the Gromov-Hausdorff convergence (measured Gromov-Hausdorff convergence) which takes into account the measures too.

The first generalization of manifolds with Ricci curvature bounded from below is given by J. Cheeger and T. Colding in [CC96] (see also [CC97], [CC00a], [CC00b] and [CN]), where they studied the theory of Ricci limits, that are the measured Gromov-Hausdorff limits of sequences of Rimannian manifolds with Ricci curvature uniformly bounded from below. This theory has good stability properties, but it is based on an extrinsic argument: Ricci limits can only be characterized (and studied) as limits of manifolds.

In searching an intrinsic approach, as the one used for spaces with bounds on sectional curvature, ten years later, indipendently in the works [S06a] and [S06b] by K.-T. Sturm and in [LV] by J. Lott and C. Villani, it was introduced the concept of Curvature-Dimension condition CD(K, N), where $K \in \mathbb{R}$ is a synthetic lower bound on the Ricci curvature and $N \in [1, +\infty]$ is an upper bound on the dimension.

In the class of $\mathsf{CD}(K, N)$ spaces several properties of manifolds with Ricci curvature bounded from below hold, for instance the Bishop-Gromov volume comparison ([S06b]), the Brunn-Minkowski inequality ([S06b], [BaSt]), the Bonnet-Myers theorem ([K]), the Laplacian comparison ([G15], [CaMo20]), the Lévy-Gromov isoperimetric inequality ([CaMo17], [CMM]) and a weak local (1,1)-Poincaré inequality ([R12b]), but it also contains non-Riemannian Finsler manifolds. The definition of $\mathsf{RCD}(K, N)$ spaces, which stands for Riemannian Curvature-Dimension condition, is given in [AGS14b] (for N = $+\infty$) and in [G15] (for $N \in [1, +\infty)$) in order to exclude Finsler manifolds and keep only $\mathsf{CD}(K, N)$ spaces with a "Riemannian structure": in addition to the $\mathsf{CD}(K, N)$ condition, the $\mathsf{RCD}(K, N)$ condition requires that a sort of tangent bundle (or equivalently a generalization of the Sobolev space $W^{1,2}$ on metric measure spaces) is a Hilbert space.

The definition of $\mathsf{RCD}(K, N)$ spaces allows to introduce second order differential calculus and to prove other important results such as a weak formulation of the Bochner inequality and some rigidity theorems such as the Cheeger-Gromoll Spltting Theorem and the "volume cone implies metric cone" principle (see [G13a], [G13b] and [DPG]).

The aim of this work is to study splitting-type theorems in more generality: more precisely, we prove that if there exists a function with "good" Laplacian and Hessian then the space is isomorphic to the product of the real line \mathbb{R} with another space. We talk in detail about these theorems in the next sections, where we describe, as first thing, the known results of this type, and then the new general splitting principle we obtained in [GMa] and a couple of applications.

Splitting-type theorems

In this section we see some splitting-type theorems proved in the non-smooth setting.

The first one is the Cheeger-Gromoll Splitting Theorem.

Its first version on 2-dimensional surfaces with non-negative curvature was proved by S. Cohn-Vossen in [C-V], and about twenty years later Toponogov proved a version of the Splitting Theorem for manifolds with positive sectional curvature in [T]. Short after that A. D. Milka proved it in Alexandrov spaces (see [Mil]).

In [CG] J. Cheeger and D. Gromoll proved it for manifolds with positive Ricci curvature, then J. Cheeger and T. Colding proved in [CC96] an Almost Splitting Theorem, which leads directly to the Splitting Theorem for Ricci limit spaces.

The $\mathsf{CD}(0, N)$ class of metric measure spaces is too big, indeed it contains every space of the form $(\mathbb{R}^d, \mathsf{d}_{\|\cdot\|}, \mathcal{L}^d)$ where $\mathsf{d}_{\|\cdot\|}$ is a distance induced by a norm $\|\cdot\|$. It is easy to see that if d = 2 and the norm $\|\cdot\|$ does not come from a scalar product then the Splitting Theorem cannot hold.

Its proof in the RCD setting is due to N. Gigli and can be found in [G13a] (see also [G13b]). We recall here its statement in RCD(0, N) spaces.

Theorem 0.0.2 (Splitting Theorem for RCD spaces). Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(0, N)$ space containing a line, i.e. such that there exists a map $\bar{\gamma} : \mathbb{R} \to X$ such that

$$\mathsf{d}(\bar{\gamma}_s, \bar{\gamma}_t) = |t - s|$$

for every $t, s \in \mathbb{R}$.

Then $(X, \mathsf{d}, \mathfrak{m})$ is isomorphic to the product of the Euclidean line with the Lebesgue measure $(\mathbb{R}, \mathsf{d}_{\mathrm{Eucl}}, \mathcal{L}^1)$ and another space $(X', \mathsf{d}', \mathfrak{m}')$ where the product distance $\mathsf{d}_{\mathrm{Eucl}} \times \mathsf{d}'$ is defined by

$$\mathsf{d}_{\mathrm{Eucl}} \times \mathsf{d}'((t,x'),(s,y'))^2 := \mathsf{d}'(x',y')^2 + |t-s|^2$$

for every $x', y' \in X'$ and for every $t, s \in \mathbb{R}$.

Moreover if $N \ge 2$ then (X', d', \mathfrak{m}') is an $\mathsf{RCD}(0, N-1)$ space; otherwise, if $N \in [1, 2)$, the space X' is just a point.

The idea of the proof is, as in the smooth setting, to take the Busemann function $b: X \to \mathbb{R}$ defined as

$$\mathbf{b}(x) := \lim_{t \to +\infty} t - \mathsf{d}(x, \bar{\gamma}_t),$$

where $\bar{\gamma}$ is a fixed line in X. In a certain sense, the modulus of the differential of b is **m**-almost everywhere equal to 1, moreover Δb and Hess(b) are both 0.

Using these properties, it is proved that, denoting with Fl. the gradient flow of b, the map $(x', t) \mapsto \mathsf{Fl}_t(x')$ is a measure preserving isometry between $(X, \mathsf{d}, \mathfrak{m})$ and $(\mathbb{R} \times X', \mathsf{d}_{\mathrm{Eucl}} \times \mathsf{d}', \mathcal{L}^1 \times \mathfrak{m}')$, where X' is a level set of b, d' is the distance induced on X' and \mathfrak{m}' is the measure on the level set X' obtained disintegrating \mathfrak{m} . The proof of the isomorphism is reached by proving that the Sobolev spaces of X and the product space are isometric and passing to the metric isomorphism thanks to the Sobolev-to-Lipschitz property.

The second rigidity theorem we recall is the "volume cone to metric cone" principle: its proof in Ricci limit spaces is due to J. Cheeger and T. Colding (see [CC96], they proved an "almost rigidity" result for this too, which we do not recall here), while in the RCD setting it has been proved by G. De Philippis and N. Gigli in [DPG].

Theorem 0.0.3 (From volume cone to metric cone). Let $N \in [1, +\infty)$ and let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(0, N)$ space with $O \in X$ and R > r > 0 such that

$$\mathfrak{m}(B_R(O)) = \left(\frac{R}{r}\right)^N \mathfrak{m}(B_r(O)).$$

Then exactly one of the following holds:

- i) the sphere $S_{\frac{R}{2}}(O)$ contains only one point: in this case (X, d) is isometric to $[0, \operatorname{diam}(X)]$ $([0, +\infty)$ if X is unbounded) with an isometry which sends O in 0 and the measure $\mathfrak{m}|_{B_R(O)}$ to $N\mathfrak{m}(B_R(O))x^{N-1}dx$;
- ii) the sphere $S_{\frac{R}{2}}(O)$ contains exactly two points: in this case (X, d) is a 1-dimensional Riemannian manifold, possibly with boundary, moreover there is a bijective local isometry (in the sense of distance-preserving maps) from $B_R(O)$ to (-R, R) sending O to 0 and the measure $\mathfrak{m}|_{B_R(O)}$ to the measure $\frac{1}{2}N\mathfrak{m}(B_R(O))|x|^{N-1}dx$, and such local isometry is an isometry when restricted to $\overline{B}_{\frac{R}{2}}(O)$;
- iii) the sphere $S_{\frac{R}{2}}(O)$ contains more than two points: in this case $N \geq 2$ and there exists an $\mathsf{RCD}(N-2, N-1)$ space $(X', \mathsf{d}', \mathfrak{m}')$ with $\operatorname{diam}(X') \leq \pi$ such that the ball $B_R(O)$ is locally isometric to the ball $B_R(O_Y)$ of the cone Y built over X', and such local isometry is an isometry when restricted to $\overline{B}_{\frac{R}{2}}(O)$.

The strategy to prove this theorem is similar to the one of the Splitting Theorem, but it requires a more sophisticated calculus with a more precise concept of Hessian. The Busemann function in this case is $b(x) := \frac{1}{2} d(O, x)^2$. The modulus of the differential of b is $\sqrt{2b}$, and its Laplacian is constantly equal to N on $B_R(O)$. Moreover its Hessian is equal to the identity.

A variant of this result has been proved by N. Gigli and I. Y. Violo in [GV].

Theorem 0.0.4 (From outer functional cone to outer metric cone). Let $(X, \mathsf{d}, \mathfrak{m})$ be an $\mathsf{RCD}(0, N)$ space with $N \ge 2$ and let U be an open subset of X with ∂U bounded. Suppose that there exists a positive $b \in D(\Delta, U) \cap C(\overline{U})$ such that

- -) $\Delta \mathbf{b} = N$ holds \mathfrak{m} -a.e. in U;
- -) $|D\sqrt{2b}|^2 = 1$ holds **m**-a.e. in U;
- -) $b_0 := \limsup_{x \to \partial U} b(x) < +\infty;$
- -) $\{b > b_0\} \neq \emptyset$.

Then the following holds:

- i) there exists an $\mathsf{RCD}(N-2, N-1)$ space $(X', \mathbf{d}', \mathbf{m}')$ with $\operatorname{diam}(X') \leq \pi$ such that, denoting with $(Y, \mathbf{d}_Y, \mathbf{m}_Y)$ the N-cone built over X' with vertex in O_Y , there exists a bijective measure preserving local isometry $S : \{b > b_0\} \to Y \setminus \overline{B_r}(O_Y)$ with $r := \sqrt{2b_0}$;
- ii) if diam(X') = π then (X, d, m) is isomorphic to (Y, d_Y, m_Y), while if diam(X') < π then the local isometry S is an isometry between Y $\overline{B_{r'}}(O_Y)$ and $\{b > \frac{1}{2}r_Z^2\}$, where $r' := \sqrt{2b_0}\left(1 - \sin\left(\frac{(\operatorname{diam}(X')}{2}\right)\right)^{-1}$;
- *iii)* the function b has the following explicit form:

$$\mathbf{b}(x) = \frac{1}{2} \mathsf{d}_{\mathbf{Y}}(O_{\mathbf{Y}}, S(x))^2 = \frac{1}{2} (\mathsf{d}(x, \partial\{\mathbf{b} > \mathbf{b}_0\}) + \sqrt{2\mathbf{b}_0})^2$$

for every $x \in \{b > b_0\}$. In particular for every $t > b_0$ the level set $\left\{b > \frac{t^2}{2}\right\}$ is Lipschitz-path connected and isometric, with its induced intrinsic metric, to (X', d').

The main difference between the proof of Theorem 0.0.3 and Theorem 0.0.4 is the use of the theory of Regular Lagrangian Flows to build the flow of the "gradient" of the Busemann function b.

A similar approach is used by C. Connell, X. Dai, J. Nuñéz-Zimbrón, R. Perales, P. Suaréz-Serrato and G. Wei in [CD+] to prove the following theorem. **Theorem 0.0.5.** Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(-(N-1), N)$ space for a certain $N \in (1, +\infty)$. Assume that there exists a function $b \in D(\Delta_{loc})$ such that $|Db| = 1 \mathfrak{m}$ -a.e. and $\Delta b = N - 1$, then (X, d, \mathfrak{m}) is isomorphic to a warped product space $\mathbb{R} \times_w X'$ with $w_d = \exp$ and $w_{\mathfrak{m}} = \exp^{N-1}$, where (X', d', \mathfrak{m}') is an $\mathsf{RCD}(0, N)$ space.

Functions w_d and w_m affect respectively the distance and the measure we take on the product space $\mathbb{R} \times X'$. In the next section and in Section 1.7 we see precisely their role in defining warped product spaces.

General splitting principle and applications

We state now the main result of this work (see [GMa]) and a couple of applications to spaces with positive spectrum of the Laplacian.

On a smooth manifold M, almost by definitions, the structure of warped product emerges when one can find a function $b: M \to \mathbb{R}$ satisfying

$$\begin{aligned} |D\mathbf{b}| &\equiv 1, \\ \Delta \mathbf{b} &= \psi_{\mathfrak{m}} \circ \mathbf{b}, \\ \text{Hess } \mathbf{b} &= \psi_{\mathsf{d}} \circ \mathbf{b}(\text{Id} - \mathbf{e}_{1} \otimes \mathbf{e}_{1}) \qquad \text{where } \mathbf{e}_{1} := \frac{\nabla \mathbf{b}}{|\nabla \mathbf{b}|}, \end{aligned} \tag{0.0.1}$$

for suitable functions $\psi_{\mathfrak{m}}, \psi_{\mathsf{d}} : \mathbf{M} \to \mathbb{R}$. In this case it is easy to see that $\mathbf{M} \sim \mathbb{R} \times_w \mathbf{N}$ where:

- -) the smooth manifold N is given by $N := b^{-1}(\{0\});$
- -) denoting with Pr(x) the point along the gradient flow trajectory of b defined by $\gamma'_t = \nabla b(\gamma_t), \ \gamma_0 = x$ such that $Pr(x) \in N$, the metricmeasure isomorphism sends $x \in M$ to $(b(x), Pr(x)) \in \mathbb{R} \times N$;
- -) the metric tensor on $\mathbb{R} \times_w \mathbb{N}$ is given by $dt^2 + w_d(t)(dx')^2$, where $(dx')^2$ is the metric on N and

$$w_{\mathsf{d}}(t) := \exp\Big(\int_0^t \psi_{\mathsf{d}}(s) \,\mathrm{d}s\Big);$$

-) the measure on $\mathbb{R} \times_w \mathbb{N}$ is given by $d\mathcal{L}^1(t) \otimes w_{\mathfrak{m}}(t) d\mathrm{vol}_{\mathbb{N}}(x')$, where $\mathrm{vol}_{\mathbb{N}}$ is the volume measure on \mathbb{N} and

$$w_{\mathfrak{m}}(t) := \exp\Big(\int_0^t \psi_{\mathfrak{m}}(s) \,\mathrm{d}s\Big).$$

As shown in the previous section, the strategies used to prove Theorems 0.0.2, 0.0.3, 0.0.4 and 0.0.5 are based on the same idea (with several non-trivial technicalities):

- -) in Theorem 0.0.2 the function b satisfies (0.0.1) with $\psi_{\mathfrak{m}} = \psi_{\mathsf{d}} = 0$;
- -) in Theorem 0.0.3 and Theorem 0.0.4 the map $\sqrt{2b}$ satisfies (0.0.1) with $\psi_{\mathsf{d}} = \frac{1}{t}$ and $\psi_{\mathfrak{m}} = (N-1)\frac{1}{t}$;
- -) in Theorem 0.0.5 the function b satisfies (0.0.1) with $\psi_{\mathfrak{m}} = N 1$ and $\psi_{\mathsf{d}} = 1$.

Our aim is then to prove that if there exists a function b which satisfies (0.0.1) for suitable ψ_d and ψ_m then the space splits, i.e. X is isomorphic as a metric measure space to the warped product $\mathbb{R} \times_w X'$ with warping functions w_d and w_m , and the measure preserving isometry is given by $x \mapsto (b(x), \mathsf{Fl}_{-b(x)}(x))$, where Fl is the (Regular Lagrangian) flow of ∇b . In this sense, our splitting principle can be seen as a general tool that allows to translate the analytic information encoded in (0.0.1) into a geometric information. The theorem can be roughly stated as follows.

Theorem 0.0.6. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ space with $K \in \mathbb{R}$, $N \in [1, +\infty)$ and $\operatorname{supp}(\mathfrak{m}) = X$. Assume that there exists a function $b: X \to \mathbb{R}$ in $H^{2,2}_{loc}(X) \cap D(\Delta_{loc})$ which satisfies (0.0.1) for some $\psi_d, \psi_{\mathfrak{m}} : \mathbb{R} \to \mathbb{R}$ locally Lipschitz.

Then X is isomorphic as a metric measure space to a warped product space $\mathbb{R} \times_w X'$ for suitable warping functions w_d, w_m .

A main difference between our work and the previous similar results we mentioned is that in studying how the gradient flow of b acts on vector fields we have to (more) carefully distinguish between the component parallel to ∇b and the one orthogonal to it, since in the previous studies the specific geometry of the problem provided some simplifications that are not present here, see in particular Proposition 2.4.3.

In principle, one would like to obtain sharp informations about the RCD property satisfied by the quotient space X'. This, however, seems tricky to do in our generality and our results in this direction are sub-optimal: we prove, under quite general assumptions on the warping functions, that if the original space is RCD(K, N), then so is the quotient one. In particular, and in line with [CD+], we do not see an improving in the upper dimension bound. It seems that to obtain this a more careful analysis of the Bochner inequality – akin to that in [K] - on warped spaces is needed, but this is outside the scope of this work.

We conclude with two applications that in the smooth setting are due to P. Li and J. Wang (see [LW01] and [LW02]): in those works they studied complete manifolds with positive spectrum of the Laplacian, reaching two splitting-type theorems depending on the number (and volume) of ends of the space.

Roughly speaking, an end is a minimal unbounded connected component of X \ K for a certain compact K: for instance, \mathbb{R}^N with $N \ge 2$ has only one end, the line \mathbb{R} has two ends, the space of lines on \mathbb{R}^2 passing through the origin with rational angular coefficient has infinite ends.

While the volume of an end obviously depends on the compact K, its finiteness does not: if an end has infinite volume with respect to the compact K then it has infinite volume with respect to every compact $K' \supset K$.

The first theorem, whose smooth version can be found in [LW01], is the following. It states that if the first eigenvalue of the Laplacian is strictly positive (more precisely greater or equal to N-2 with $N \ge 3$) and the space has more than one end with infinite volume then the space splits.

Theorem 0.0.7. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(-(N-1), N)$ space with $N \ge 3$ and $\operatorname{supp}(\mathfrak{m}) = X$, and assume that the first eigenvalue of the Laplacian λ_1 is $\ge N-2$. Then one of the following holds:

- i) X has only one end with infinite volume;
- ii) X is isomorphic as metric measure space to a warped product space $\mathbb{R} \times_w X'$, where X' is a compact $\mathsf{RCD}(-(N-1), N)$ space and the warping functions are

 $w_{\mathsf{d}}(t) := \cosh(t)$ and $w_{\mathfrak{m}}(t) := \cosh^{N-1}(t).$

Moreover, in this case $\lambda_1 = N - 2$.

The idea of the proof of this theorem is similar to its smooth version, but, as usually happens, translating the proof in a more general case produces some technical difficulties. The main differences between the two cases are the following:

- -) to construct a non-constant bounded harmonic function on X (see Section 3.3) we need the Gauss-Green formula in the non-smooth setting (proved in [BPS]), and this requires a cautious work on sets with finite perimeter (see Theorem 3.3.6);
- -) as in the smooth case the rigidity comes from an equality in the Bochner inequality: in proving this Li and Wang use the fact that the Laplacian

is the trace of the Hessian, and this is not true in a generic $\mathsf{RCD}(K, N)$ space (as proved in [Han], this holds if and only if the dimension of the space is equal to N); this problem is solved using a more general Kato-type inequality and an improved version of the Bochner inequality;

- -) another key point is that, given $d|du| \parallel du$, we have that locally $|du| = \varphi \circ u$: in the smooth case this is pretty simple to see, in the non-smooth setting the proof is much more involved;
- -) in proving that the quotient space X' is $\mathsf{RCD}(K', N')$ our argument does not provide the best constants K', N'.

Let M be an N-dimensional Riemannian manifold with Ricci curvature bounded from below by K < 0. Then Cheng's Theorem states that the first eigenvalue of the Laplacian λ_1 satisfies $\lambda_1 \leq \frac{-(N-1)K}{4}$. The same estimate holds in RCD(K, N) spaces, so the second theorem we are going to prove (see [LW02] for its smooth version) studies the case in which λ_1 is the maximum admissible. Assuming that N > 3, from Theorem 0.0.7 it follows that X has only one end with infinite volume; the theorem states that if it has also (at least) an end with finite volume then the space splits.

Theorem 0.0.8. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(-(N-1), N)$ space with N > 3and $\operatorname{supp}(\mathfrak{m}) = X$, and assume that the first eigenvalue of the Laplacian λ_1 is equal to $\frac{(N-1)^2}{4}$. Then one of the following holds:

- i) X has only one end;
- ii) X is isomorphic as metric measure space to a warped product space $\mathbb{R} \times_w X'$, where X' is a compact $\mathsf{RCD}(0, N)$ space and the warping functions are

 $w_{\mathsf{d}}(t) := e^t$ and $w_{\mathfrak{m}}(t) := e^{(N-1)t}$.

Its proof is pretty similar to the proof of Theorem 0.0.7. We note that in this case we can, arguing as in [CD+], improve the bound on the curvature of the quotient space (but still not improving the bound on the dimension).

Plan of the work

In Chapter 1 we recall the definitions of all the instruments we need in order to prove the general splitting principle (Theorem 0.0.6) and their properties.

We start with some basic notations on metric measure spaces in Section 1.1. In Section 1.2 we see how to define Sobolev spaces in metric measure spaces via test plans and the modulus of the distributional differential of Sobolev functions, the so called minimal weak upper gradient. The aim of Section 1.3 is to introduce a first order differential calculus on metric measure spaces; to do that we rely on the concept of normed modules, using it to define the cotangent module (equipped with a differential operator) and its dual, the tangent module (equipped with the gradient). We conclude this section defining the divergence operator as the adjoint of the differential and the Laplacian as the divergence of the gradient. In Section 1.4 we introduce the concept of pullback of 1-forms and of speed of test plans.

In Section 1.5 we give various definitions of RCD condition on metric measure spaces: one based on the differential calculus built in Sections 1.2 and 1.3, and others based on optimal transport (different concepts of CD condition), then we see how all these are related to each other. In Section 1.6 we see how the RCD condition allows us to define a second order calculus on metric measure spaces, i.e. the space $W^{2,2}$, the Hessian and the covariant derivative.

Given that theory of differential calculus, we are ready to recall the two main instruments that we are going to use in the proof of Theorem 0.0.6: warped product spaces and Regular Lagrangian Flows. In Section 1.7 we recall the definition of warped product spaces and see how the Sobolev space of a product space behave with respect to the Sobolev spaces of the quotient spaces. Then we see when the Sobolev-to-Lipschitz property passes to the product. Section 1.8 contains the definition and the main properties of Regular Lagrangian Flows.

In Chapter 2 we prove Theorem 0.0.6 and a result about the RCD condition of the quotient space X'.

In Section 2.1 we see how to construct a Busemann function b (a function which satisfies (0.0.1)) starting with a regular enough function with a "good" Hessian. This will be useful in the applications, since usually an equality in the Bochner inequality gives a function with a "good" Hessian in that way. In Section 2.2 we fix the assumptions and the notations that we are going to use in the course of the proof. In Section 2.3 we study how the measure changes if pushed forward with the flow of ∇b and use it to deduce how the disintegrated measure on the level sets of b behaves. In Section 2.4 we see how the flow of ∇b changes the distances. From this follows a relation between the Sobolev space of X and the one of X'. Using the results of the last two sections, in Section 2.5 we prove that the map $T(x) := (b(x), Fl_{-b(x)}(x))$ is a measure preserving isometry, and we conclude with a (less general and sub-optimal) result about the RCD condition of X'.

Chapter 3 contains two applications of the general splitting principle:

Theorems 0.0.7 and 0.0.8.

We start with two more sections of preliminary facts: in Section 3.1 we define harmonic functions and a more general (but coherent with respect to the one defined before) concept of Laplacian, and use it to state an improved version of the Bochner inequality, while in Section 3.2 we recall briefly the theory of quasi-continuous functions and quasi-continuous vector fields in order to state a Gauss-Green formula in the non-smooth setting.

In Section 3.3 we define ends and see how the volume of an end is related with the existence of non-constant bounded harmonic functions on it. Using this result we construct the two non-constant harmonic functions ("bounded" assuming the hypotheses of Theorem 0.0.7 and "positive" assuming the hypotheses of Theorem 0.0.8) that will lead to the proofs of the two main theorems.

In Section 3.4 we prove Theorem 0.0.7 starting, in Section 3.4.1 proving that the bounded harmonic function u constructed in Section 3.3 satisfies (0.0.1), then, in Section 3.4.2 we prove that its minimal weak upper gradient is a composite function of u, and in Section 3.4.3 we conclude the proof using Theorem 0.0.6.

Section 3.5 contains the proof of Theorem 0.0.8, which is pretty similar to the one of Theorem 0.0.7, but in this case our result about the RCD condition of the quotient space does not apply. However this precise case has been studied in [CD+], so we recall briefly their method in Section 3.5.2.

Chapter 1

Preliminaries

In this chapter we recall the definitions, notations and known results we will use along all the work, starting with differential calculus in metric measure spaces and the definition of RCD space and concluding with some properties of warped product spaces and Regular Lagrangian Flows.

1.1 Notation on metric measure spaces

As we will always work with metric measure spaces, the aim of this section is to fix some basic notation on them.

We start considering a metric space (X, d).

A curve on X is an element of C([0, 1], X), and for a curve $\gamma \in C([0, 1], X)$ we will usually indicate with γ_t the point $\gamma(t)$. The evaluation map e_t on the set of curves is defined as follows.

Definition 1.1.1 (Evaluation map). Let (X, d) be a metric space. For every $t \in [0, 1]$, we define the map $e_t : C([0, 1], X) \to X$ as $e_t(\gamma) := \gamma_t$.

We recall now the definitions of metric speed of a curve and absolutely continuous curve.

Definition 1.1.2 (Metric speed). Let (X, d) be a metric space. For every $\gamma \in C([0, 1], X)$ and for every $t \in [0, 1]$ we define the metric speed of γ at the time t as

$$\begin{aligned} |\dot{\gamma}_t| &:= \begin{cases} \lim_{h \to 0} \frac{\mathsf{d}(\gamma_{t+h}, \gamma_t)}{h} & \text{if such limit exists;} \\ +\infty & \text{otherwise.} \end{cases} \end{aligned}$$

Definition 1.1.3 (Absolutely continuous curve). Let (X, d) be a metric space. We say that a curve γ is asbolutely continuous (AC) if there exists

 $f \in L^1([0,1])$ such that

$$\mathsf{d}(\gamma_t, \gamma_s) \le \int_t^s f(r) \,\mathrm{d}r \tag{1.1.1}$$

for every $t, s \in [0, 1]$ with t < s.

It is possible to see that if γ is absolutely continuous then its metric speed is finite for a.e. $t \in [0, 1]$ and it is an admissible function for (1.1.1). Moreover, for every other f satisfying (1.1.1) it holds $|\dot{\gamma}_t| \leq f(t)$ for a.e. $t \in [0, 1]$.

Other key instruments are the kinetic energy of a curve and the concept of geodesic curve.

Definition 1.1.4 (Kinetic energy). We define the operator kinetic energy $KE : C([0, 1], X) \rightarrow [0, +\infty]$ as

$$\operatorname{KE}(\gamma) := \begin{cases} \frac{1}{2} \int_0^1 |\dot{\gamma}_t|^2 \, \mathrm{d}t & \text{if } \gamma \text{ is absolutely continuous;} \\ +\infty & \text{otherwise.} \end{cases}$$

Definition 1.1.5 (Geodesic curve). A geodesic is a curve γ which satisfies

$$\frac{1}{2}\mathsf{d}(\gamma_0,\gamma_1)^2 = \mathrm{KE}(\gamma),$$

or, equivalently,

$$\mathsf{d}(\gamma_t, \gamma_s) = |s - t| \mathsf{d}(\gamma_0, \gamma_1)$$

for every $t, s \in [0, 1]$.

We indicate with Geo(X) the set of geodesics on X.

Let (X, d_X) and (Y, d_Y) be metric spaces, we will indicate with LIP(X, Y)the set of Lipschitz continuous functions (we omit Y if the codomain is \mathbb{R}), and for $f \in \text{LIP}(X, Y)$ we will denote with Lip(f) its Lipschitz constant and with lip(f)(x) its local Lipschitz constant at x, given by the following definition.

Definition 1.1.6 (Local Lipschitz constant). Let (X, d_X) and (Y, d_Y) be metric spaces and let $f \in LIP(X, Y)$. The local Lipschitz constant of f is the function $lip(f) : X \to [0, +\infty)$ defined as

$$\operatorname{lip}(f)(x) := \begin{cases} 0 & \text{if } x \text{ is an isolated point,} \\ \limsup_{y \to x} \frac{\mathsf{d}_{\mathrm{Y}}(f(x), f(y))}{\mathsf{d}_{\mathrm{X}}(x, y)} & \text{otherwise.} \end{cases}$$

Moreover we define the asymptotic Lipschitz constant $\lim_{a}(f) : \mathbf{X} \to [0, +\infty)$ as

$$\operatorname{lip}_{a}(f)(x) := \begin{cases} 0 & \text{if } x \text{ is an isolated point,} \\ \operatorname{lim}_{y,z \to x} \frac{\mathsf{d}_{\mathrm{Y}}(f(z), f(y))}{\mathsf{d}_{\mathrm{X}}(z, y)} & \text{otherwise.} \end{cases}$$

We will indicate with $LIP_{loc}(X, Y)$ the set of locally Lipschitz functions and with $LIP_{bs}(X, Y)$ the set of Lipschitz functions with bounded support.

Let now (X, \mathfrak{m}) be a measure space.

We indicate with $L^0(\mathfrak{m})$ the space of measurable functions on X modulo \mathfrak{m} -a.e. equality.

Definition 1.1.7 (Pushforward measure). Let (X, \mathcal{A}_X) and (Y, \mathcal{A}_Y) be two measurable spaces and let $\varphi : X \to Y$ be a measurable function. Given any measure μ on (X, \mathcal{A}_X) , we define the pushforword measure $\varphi_*\mu$ on (Y, \mathcal{A}_Y) as

$$(\varphi_*\mu)(B) := \mu\left(\varphi^{-1}(B)\right)$$

for every $B \in \mathcal{A}_{Y}$.

Remark 1.1.8. For every $f \in L^1(\varphi_*\mu)$ it holds the change of variable formula

$$\int f \,\mathrm{d}(\varphi_* \mu) = \int f \circ \varphi \,\mathrm{d}\mu. \tag{1.1.2}$$

As we said, the "right" objects to generalize manifolds with Ricci curvature bounded from below are metric measure spaces, which are complete and measurable metric spaces equipped with a non-negative Borel measure finite on balls.

Definition 1.1.9 (Metric measure space). A triple (X, d, \mathfrak{m}) is a metric measure space if

- -) (X, d) is a complete and separable metric space;
- -) **m** is a non-negative Borel measure on X which is finite on balls.

1.2 Sobolev space on metric measure spaces

From the end of the last century several notions of Sobolev space on metric measure spaces has been introduced.

The first approach is due to P. Hajłasz in [Haj], who defined the Sobolev space $W^{1,p}(X)$ in the following way: $f \in W^{1,p}(X)$ if $f \in L^p(\mathfrak{m})$ and there exists $g \in L^p(\mathfrak{m})$ such that for \mathfrak{m} -a.e. x, y it holds

$$|f(x) - f(y)| \le \mathsf{d}(x, y)(g(x) + g(y)).$$

The "module of the differential of f" is defined as the unique g with minimal L^p -norm that satisfies the previous inequality. This definition is consistent with the classical definition of Sobolev spaces in \mathbb{R}^n , but that notion of differential does not satisfy a locality property, i.e. it may depend not only on the local behaviour of the function.

A different approach is given by J. Cheeger in [C], who defined Sobolev functions using the concept of upper gradients introduced in [HeKo]: a Borel function g is an upper gradient of f provided that for every absolutely continuous curve $\gamma : [0, 1] \to X$ the curve $f \circ \gamma$ is absolutely continuous and satisfies $|(f \circ \gamma)'_t| \leq g(\gamma_t)|\dot{\gamma}_t|$.

Cheeger's strategy is then to define an energy functional $\mathsf{E}_{\mathsf{Ch}}:L^2\to[0,+\infty]$ as

$$\mathsf{E}_{\mathsf{Ch}}(f) := \inf \liminf_{n \to +\infty} \frac{1}{2} \int g_n^2 \, \mathrm{d}\mathfrak{m}_n$$

where the infimum is taken among all the sequences $(g_n)_{n \in \mathbb{N}}$ such that there exists a sequence $f_n \to f$ in $L^2(\mathfrak{m})$ for which g_n is an upper gradient of f_n for every $n \in \mathbb{N}$. The Sobolev space $W^{1,2}$ is defined as the space of L^2 functions with finite E_{Ch} energy.

Inspired by Cheeger's approach, in [AGS14a] is proposed a definition via an "approximation with Lipschitz functions" argument: noticing that for a Lipschitz function f the maps $\lim(f)$ and $\lim_{a}(f)$ are upper gradients, one can define the energy functional $\mathsf{E}_*: L^2(\mathfrak{m}) \to [0, +\infty]$ as

$$\mathsf{E}_*(f) := \inf \liminf_{n \to +\infty} \frac{1}{2} \int \operatorname{lip}^2(f) \, \mathrm{d}\mathfrak{m},$$

where the infimum is taken among all the sequences $(f_n)_{n\in\mathbb{N}} \subset \operatorname{LIP}(X)$ such that $f_n \xrightarrow{L^2(\mathfrak{m})} f$, then the Sobolev space $W^{1,2}(X)$ as the space of $L^2(\mathfrak{m})$ functions with finite energy E_* (one can argue similarly with the asymptotic Lipschitz constant lip_a instead of lip in the definition of E_*). In [Sha] N. Shanmugalingam gave a notion of Sobolev space based on how functions behave on (a good class of) curves. This approach relies on the concept of 2-modulus of a family of curves.

A similar approach, based on the concept of test plan instead of 2-moduli, is given by L. Ambrosio, N. Gigli and G. Savaré in [AGS14a].

The approaches given by Cheeger, Shanmugalingam, Ambrosio, Gigli and Savaré are consistent with the classical definition of Sobolev space in \mathbb{R}^n , and their definition of "modulus of the differential" are "local" (Proposition 1.2.11). Moreover, a consequence of a deep approximation result contained in [AGS13] (which we recall at the end of this section, Theorem 1.2.16) is the equivalence between these approaches.

All our work relies on the definition of Sobolev space via test plan, so the aim of this section is to recall the basic definitions and first properties of Sobolev functions in this sense.

We start recalling the definition of test plans, which are probability measures on the space of absolutely continuous curves that, in a certain sense, does not overlap too much.

Definition 1.2.1 (Test plan). Let (X, d, \mathfrak{m}) be a metric measure space. A probability measure $\pi \in \mathscr{P}(C([0, 1], X))$ is said to be a test plan on X provided the following two properties are satisfied:

- i) there exists a constant C > 0 such that $(e_t)_* \pi \leq C \mathfrak{m}$ for every $t \in [0, 1]$;
- ii) it holds that $\int_0^1 \int |\dot{\gamma}_t|^2 \mathrm{d}\boldsymbol{\pi}(\gamma) \,\mathrm{d}t < +\infty$.

We are now ready to give the definition of weak upper gradient of a function f via test plans, and with it the definition of Sobolev class $S^2(X)$, which in the smooth setting is the class of functions with L^2 weak derivative.

Definition 1.2.2 (Weak upper gradient). Let (X, d, \mathfrak{m}) be a metric measure space and let $f : X \to \mathbb{R}$ be a Borel function. Then we say that a non-negative function $G \in L^2(\mathfrak{m})$ is a weak upper gradient of f if for every test plan π on X it holds

$$\int |f(\gamma_1) - f(\gamma_0)| \,\mathrm{d}\boldsymbol{\pi}(\gamma) \leq \int_0^1 \int G(\gamma_t) |\dot{\gamma}_t| \,\mathrm{d}\boldsymbol{\pi}(\gamma) \mathrm{d}t$$

Definition 1.2.3 (Sobolev class). The Sobolev class $S^2(X)$ is defined as the space of all Borel functions $f: X \to \mathbb{R}$ that admit a weak upper gradient.

An easy consequence of this definition and Fatou's Lemma is the lower semicontinuity of weak upper gradients. **Proposition 1.2.4.** Let $(f_n)_n \subset S^2(X)$ and $f : X \to \mathbb{R}$ Borel such that $f_n(x) \to f(x)$ for \mathfrak{m} -a.e. $x \in X$. For every $n \in \mathbb{N}$ let G_n be a weak upper gradient of f_n and assume that $G_n \rightharpoonup G$ weakly in $L^2(\mathfrak{m})$ for some $G \in L^2(\mathfrak{m})$. Then $f \in S^2(X)$ and G is a weak upper gradient of f.

A consequence of Proposition 1.2.4 is that the set of weak upper gradients of a function $f \in S^2(\mathbf{X})$ is closed and convex in $L^2(\mathfrak{m})$, then there exists a unique weak upper gradient with minimal $L^2(\mathfrak{m})$ -norm.

Definition 1.2.5 (Minimal weak upper gradient). Let $f \in S^2(X)$. Then the unique weak upper gradient with minimal $L^2(\mathfrak{m})$ -norm is called minimal weak upper gradient of f and it is denoted by $|Df| \in L^2(\mathfrak{m})$.

The minimal weak upper gradient of an S^2 map f is the map that behaves like the modulus of the distributional differential of f. Note that the structures we have at this moment are not enough to give a definition of differential of f, in order to do so we need the theory of normed modules that we will introduce in the next section, but the modulus of the differential of functions is enough to define the Sobolev space $W^{1,2}(X)$.

Definition 1.2.6 (Sobolev space). Let (X, d, \mathfrak{m}) be a metric measure space. The Sobolev space $W^{1,2}(X)$ is the set of $L^2(\mathfrak{m})$ functions that admit a weak upper gradient, i.e. $W^{1,2}(X) := L^2(\mathfrak{m}) \cap S^2(X)$.

For $f \in W^{1,2}(X)$ we define its $W^{1,2}(X)$ -norm as

$$||f||_{W^{1,2}(\mathbf{X})} := \sqrt{||f||_{L^2(\mathfrak{m})}^2 + ||Df||_{L^2(\mathfrak{m})}^2}.$$

Remark 1.2.7. For every $f \in LIP_{bs}(X)$ it holds $f \in S^2(X)$ and

$$|Df| \le \operatorname{lip}(f) \le \operatorname{lip}_a(f) \le \operatorname{Lip}(f)$$
 m-a.e. in X. (1.2.1)

Remark 1.2.8. The Sobolev space $W^{1,2}(X)$ with the norm $\|\cdot\|_{W^{1,2}(X)}$ is a Banach space, but in general it is not a Hilbert space. For instance if X is a smooth Finsler manifold then $W^{1,2}(X)$ is Hilbert if and only if X is a Riemannian manifold.

The previous Remark motivates the definition of infinitesimally Hilbertian metric measure space.

Definition 1.2.9 (Infinitesimal Hilbertianity). The metric measure space (X, d, \mathfrak{m}) is said to be infinitesimally Hilbertian if $(W^{1,2}(X), \|\cdot\|_{W^{1,2}(X)})$ is a Hilbert space.

Remark 1.2.10. In all this work we heavily rely on the infinitesimal Hilbertianity of the space since, as we said in the Introduction, it is the hypothesis that allows to exclude Finsler manifolds and allows us to define a second order differential calculus on metric measure spaces.

The minimal weak upper gradient obeys to some natural calculus rules such as a locality property, the chain rule and the Leibniz rule.

Proposition 1.2.11 (Locality for the minimal weak upper gradient). Let $f, g \in S^2(\mathbf{X})$. Then |Df| = |Dg| holds \mathfrak{m} -a.e. on $\{f = g\}$.

Proposition 1.2.12 (Chain rule for the minimal weak upper gradient). Let $f \in S^2(\mathbf{X})$. If a Borel set $N \subset \mathbb{R}$ is \mathcal{L}^1 -negligible then |Df| = 0 holds \mathfrak{m} -a.e. on $f^{-1}(N)$.

Moreover, if $\varphi \in LIP(\mathbb{R})$, then $\varphi \circ f \in S^2(X)$ and

$$|D(\varphi \circ f)| = |\varphi'| \circ f|Df| \tag{1.2.2}$$

holds m-a.e..

Proposition 1.2.13 (Leibniz rule for the minimal weak upper gradient). Let $f, g \in S^2(X) \cap L^{\infty}(\mathfrak{m})$. Then $fg \in S^2(X) \cap L^{\infty}(\mathfrak{m})$ and the inequality

$$|D(fg)| \le |f||Dg| + |g||Df|$$

holds m-a.e. on X.

Thanks to Proposition 1.2.11, we can define the local Sobolev class $S^2_{loc}(\mathbf{X})$ and, with that, the local Sobolev space $W_{loc}^{1,2}(\mathbf{X})$.

Definition 1.2.14 (Local Sobolev class). Let (X, d, \mathfrak{m}) be a metric measure space and let $f : X \to \mathbb{R}$ be a Borel function. We say that $f \in S^2_{loc}(X)$ provided that for every bounded Borel set $B \subset X$ there exists a function $f_B \in S^2(\mathbf{X})$ such that $f_B = f$ holds \mathfrak{m} -a.e. in B.

For every $f \in S^2_{loc}(X)$ we define its minimal weak upper gradient as the function (unique up to \mathfrak{m} -negligible sets) |Df| such that for every bounded Borel set B and every function $f_B \in S^2(\mathbf{X})$ which coincides with f on B it holds $|Df| = |Df_B|$.

Definition 1.2.15 (Local Sobolev space). Let (X, d, \mathfrak{m}) be a metric measure space. The local Sobolev space $W_{loc}^{1,2}(X)$ is defined as $S_{loc}^2(X) \cap L_{loc}^2(\mathfrak{m})$. Moreover, given $U \subset X$ open, we define $W_0^{1,2}(U)$ as the closure in $W^{1,2}(X)$

of $LIP_{bs}(U)$.

We conclude this section with the following approximation theorem: its proof can be found in [AGS13] (see also [AGS14a]).

Theorem 1.2.16. Let (X, d, \mathfrak{m}) be a metric measure space. Then Lipschitz functions on X are dense in energy in $W^{1,2}(X)$, namely for every $f \in W^{1,2}(X)$ there exists a sequence $(f_n)_n \subset \text{LIP}(X) \cap L^2(\mathfrak{m})$ such that

$$f_n \xrightarrow{L^2} f, \quad and \quad \lim_{a \to \infty} |Df|.$$
 (1.2.3)

An easy consequence of (1.2.1) and (1.2.3) (and the minimality of the minimal weak upper gradient) is that, given f and $(f_n)_n$ as in Theorem 1.2.16, also the sequences $(\lim (f_n))_n$ and $(|Df_n|)_n$ converge in $L^2(\mathfrak{m})$ to |Df|.

1.3 Normed modules and first order differential calculus

In the previous section we recalled the definition of Sobolev space on a metric measure space defining what the "modulus of the differential" of a function is. In order to develop a tensor calculus on metric measure spaces we recall now a definition of "tangent/cotangent bundle" in the non-smooth setting given by N. Gigli in [G18b].

More precisely, it generalizes the concept of "space of L^2 sections of a normed vector bundle", and the idea, inspired by [W], is to consider moduli over the commutative ring $L^{\infty}(\mathbf{X})$ with a pointwise norm operator.

Definition 1.3.1 (L^2 -normed L^∞ -module). Let $(X, \mathsf{d}, \mathfrak{m})$ be a metric measure space. An $L^2(\mathfrak{m})$ -normed $L^\infty(\mathfrak{m})$ -module is a quadruple ($\mathscr{M}, \|\cdot\|_{\mathscr{M}}, \cdot, |\cdot|$) with the following properties:

- i) $(\mathcal{M}, \|\cdot\|_{\mathcal{M}})$ is a Banach space;
- ii) the multiplication by $L^{\infty}(\mathfrak{m})$ functions $\cdot : L^{\infty}(\mathfrak{m}) \times \mathcal{M} \to \mathcal{M}$ is a bilinear map satisfying
 - $\begin{aligned} f \cdot (g \cdot v) &= (fg) \cdot v & \text{for every } f, g \in L^{\infty}(\mathfrak{m}) \text{ and } v \in \mathscr{M}, \\ \hat{1} \cdot v &= v & \text{for every } v \in \mathscr{M}, \end{aligned}$

where $\hat{1}$ is the function identically equal to 1 on X;

iii) the pointwise norm $|\cdot|: \mathcal{M} \to L^2(\mathfrak{m})$ satisfies

$$\begin{split} |v| &\geq 0 & holds \ \mathfrak{m}\text{-}a.e. \ for \ every \ v \in \mathscr{M}, \\ |f \cdot v| &= |f| |v| & holds \ \mathfrak{m}\text{-}a.e. \ for \ every \ f \in L^{\infty}(\mathfrak{m}) \ and \ v \in \mathscr{M}, \\ \|v\|_{\mathscr{M}} &= \||v|\|_{L^{2}(\mathfrak{m})} & for \ every \ v \in \mathscr{M}. \end{split}$$

A variant of this definition which does not rely on integrability hypotheses is the following.

Definition 1.3.2 (L^0 -normed L^0 -module). An $L^0(\mathfrak{m})$ -normed $L^0(\mathfrak{m})$ -module is a quadruple ($\mathscr{M}^0, \tau, \cdot, |\cdot|$) with the following properties:

- i) (\mathcal{M}^0, τ) is a topological vector space;
- ii) the multiplication by L^0 functions $\cdot : L^0(\mathfrak{m}) \times \mathscr{M}^0 \to \mathscr{M}^0$ is a bilinear map satisfying

$$f \cdot (g \cdot v) = (fg) \cdot v, \text{ and } \hat{1} \cdot v = v$$

for every $f, g \in L^0(\mathfrak{m})$ and $v \in \mathscr{M}^0$, where $\hat{1}$ is the functions identically equal to 1;

iii) the pointwise norm $|\cdot|: \mathscr{M}^0 \to L^0(\mathfrak{m})$ satisfies

$$\begin{aligned} |v| \ge 0 & \mathfrak{m}\text{-}a.e. \text{ for every } v \in \mathscr{M}^0, \\ |f \cdot v| = |f||v| & \mathfrak{m}\text{-}a.e. \text{ for every } f \in L^0(\mathfrak{m}) \text{ and } v \in \mathscr{M}^0; \end{aligned}$$

iv) the topology τ is induced by the map $\mathsf{d}_{\mathscr{M}^0} : \mathscr{M}^0 \times \mathscr{M}^0 \to [0, +\infty)$ defined by

$$\mathsf{d}_{\mathscr{M}^0}(v,w) := \int |v-w| \wedge 1 \, \mathrm{d}\mathfrak{m}'$$

for some $\mathfrak{m}' \in \mathscr{P}(X)$ with $\mathfrak{m} \ll \mathfrak{m}' \ll \mathfrak{m}$.

The concepts of L^2 -normed L^{∞} -module and L^0 -normed L^0 -module are strictly related, indeed they can be seen one as the restriction/completion of the other thanks to the following propositions.

Proposition 1.3.3 (L²-restriction). Let \mathcal{M}^0 be an L^0 -normed L^0 -module and let \mathcal{M} be defined as

$$\mathscr{M} := \left\{ v \in \mathscr{M}^0 : |v| \in L^2(\mathfrak{m}) \right\}.$$

Then \mathscr{M} with the norm $\|\cdot\|_{\mathscr{M}} := \||\cdot|\|_{L^2(\mathfrak{m})}$ and the product and pointwise norm induced by the ones of \mathscr{M}^0 is an L^2 -normed L^{∞} -module.

Proposition 1.3.4 (L^0 -completion). Let \mathscr{M} be an L^2 -normed L^{∞} -module. Then there exists a unique couple (\mathscr{M}^0, ι) where \mathscr{M}^0 is an L^0 -normed L^0 module and the map $\iota : \mathscr{M} \to \mathscr{M}^0$ is a linear operator with dense image that preserves the pointwise norm.

Uniqueness is intended up to unique isomorphism.

In a very natural way we define Hilbert modules and dual normed modules.

Definition 1.3.5 (Hilbert module). Let \mathscr{M} be an L^2 -normed L^{∞} -module. We say that \mathscr{M} is a Hilbert module if $(\mathscr{M}, \|\cdot\|_{\mathscr{M}},)$ is a Hilbert space, or, equivalently, if the pointwise norm satisfies the parallelogram rule, i.e.

$$|v + w|^{2} + |v - w|^{2} = 2(|v|^{2} + |w|^{2})$$

holds \mathfrak{m} -a.e. for every $v, w \in \mathcal{M}$.

Definition 1.3.6 (Dual normed module). Let \mathscr{M} be an L^2 -normed L^{∞} module. Its dual normed module \mathscr{M}^* is defined as the set of linear and continuous operators $L : \mathscr{M} \to L^1(\mathfrak{m})$ such that L(fv) = fL(v) for every $v \in \mathscr{M}$ and $f \in L^{\infty}(\mathfrak{m})$ endowed with the operator norm

$$||L||_* := \sup_{\|v\|_{\mathscr{M}} \le 1} ||L(v)||_{L^1(\mathfrak{m})}$$

and the pointwise norm

$$|L|_* := \operatorname{ess-sup}_{v \in \mathscr{M}, |v| \le 1 \ \mathfrak{m}\text{-}a.e.} L(v).$$

It can be proved that the dual normed module just defined is an L^2 -normed L^{∞} -module.

Tangent and cotangent modules, that we will define in the next section, will be the dual one of the other. Moreover their Hilbertianity will be related to the infinitesimal Hilbertianity of the space.

1.3.1 Cotangent and tangent modules

Thanks to the next existence and uniqueness theorem we define the cotangent module and the differential operator on a generic metric measure space.

Theorem 1.3.7 (Cotangent Module). Let (X, d, \mathfrak{m}) be a metric measure space. Then there exists a unique couple $(L^2(T^*X), d)$, where $L^2(T^*X)$ is an L^2 -normed L^{∞} -module and $d: S^2(X) \to L^2(T^*X)$ is a linear operator, such that the following conditions hold:

- i) |df| = |Df| holds \mathfrak{m} -a.e. for every $f \in S^2(\mathbf{X})$;
- ii) $L^2(T^*X)$ is generated by $\{df : f \in S^2(X)\}$.

We shall refer to $L^2(T^*X)$ as cotangent module and to d as differential.

We will indicate with $L^0(T^*X)$ the L^0 -completion of $L^2(T^*X)$.

The differential just defined inherits from the weak upper gradient similar calculus rules.

Proposition 1.3.8 (Locality for the differential). Let $f, g \in S^2(X)$. Then df = dg holds \mathfrak{m} -a.e. on $\{f = g\}$.

Proposition 1.3.9 (Chain rule for the differential). Let $f \in S^2(X)$. If a Borel set $N \subset \mathbb{R}$ is \mathcal{L}^1 -negligible then df = 0 holds \mathfrak{m} -a.e. on $f^{-1}(N)$. Moreover, if $\varphi \in \operatorname{LIP}(\mathbb{R})$, then $\varphi \circ f \in S^2(X)$ and $d(\varphi \circ f) = \varphi' \circ f df$.

Proposition 1.3.10 (Leibniz rule for the differential). Let $f, g \in S^2(X) \cap L^{\infty}(\mathfrak{m})$. Then $fg \in S^2(X) \cap L^{\infty}(\mathfrak{m})$ and d(fg) = fdg + gdf.

We define now the tangent module, which will be the space of "vector fields", as the dual of the cotangent module (as in Definition 1.3.6).

Definition 1.3.11 (Tangent module). Let (X, d, \mathfrak{m}) be a metric measure space. The tangent module $L^2(TX)$ of X is defined as the dual normed module of $L^2(T^*X)$. Its elements are called vector fields.

Similarly to the definition of $L^0(T^*X)$, we indicate with $L^0(TX)$ the L^0 -completion of $L^2(TX)$.

In order to define the gradient of a Sobolev function we need to assume that the space is inifinitesimally Hilbertian and combine the two following results: the first is a Riesz representation theorem for normed modules, the second is the anticipated relation between Hilbert modules and infinitesimal Hilbertianity.

Theorem 1.3.12 (Riesz). Let \mathscr{M} be a Hilbert L^2 -normed L^{∞} -module. Then the map $\mathscr{M} \to \mathscr{M}^*$ which sends v to $\langle \cdot, v \rangle$ is an isomorphism of modules.

Proposition 1.3.13 (Infinitesimal Hilbertianity). Let (X, d, \mathfrak{m}) be a metric measure space. Then X is infinitesimally Hilbertian if and only if $L^2(TX)$ and $L^2(T^*X)$ are Hilbert modules.

The gradient can then be defined, in infinitesimally Hilbertian spaces, as the dual of the differential.

Definition 1.3.14 (Gradient). Let (X, d, \mathfrak{m}) be an infinitesimally Hilbertian metric measure space. Then we denote by $\nabla f \in L^2(TX)$ the element corresponding to $df \in L^2(T^*X)$ via the Riesz isomorphism. We call ∇f the gradient of f. **Remark 1.3.15.** Infinitesimal Hilbertianity is not really necessary to define the gradient of a function, indeed given $f \in S^2(X)$ the gradient of f can be defined as the set of vector fields v such that $df(v) = |df|^2 = |v|^2$ holds **m**-a.e.. Gradients defined like that obey to good calculus rules, but in general they are not a singleton, and even if they are they may not depend linearly on f. It can be proved that the space is infinitesimally Hilbertian if and only if for every $f \in S^2(X)$ the gradient of f is a singleton (whose element is denoted by ∇f) and for every $f, g \in W^{1,2}(X)$ the formula $\nabla(f+g) = \nabla f + \nabla g$ holds **m**-a.e. in X. In this case this gradient is equivalent to the one defined in Definition 1.3.14.

From the calculus rules for the differential immediately follow the ones for the gradient.

Proposition 1.3.16 (Chain rule for the gradient). Let X be an infinitesimally Hilbertian metric measure space and let $f \in S^2_{loc}(X)$ and $\varphi \in \text{LIP}_{loc}(\mathbb{R})$. Then $\nabla(\varphi \circ f) = \varphi' \circ f \nabla f$.

Proposition 1.3.17 (Leibniz rule for the gradient). Let X be an infinitesimally Hilbertian metric measure space and let $f, g \in S^2_{loc}(X) \cap L^{\infty}_{loc}(\mathfrak{m})$. Then $\nabla(fg) = f\nabla g + g\nabla f$.

1.3.2 Divergence and Laplace operators

We define the divergence operator as the adjoint of the differential.

Definition 1.3.18 (Divergence). Let (X, d, \mathfrak{m}) be a metric measure space, $U \subset X$ open and $v \in L^0(TX)$. We say that $v \in D(\operatorname{div}_{loc}, U)$ provided there is $f \in L^2_{loc}(U)$ such that

$$\int \mathrm{d}g(v)\,\mathrm{d}\mathfrak{m} = -\int fg\,\mathrm{d}\mathfrak{m}$$

for every $g \in W_0^{1,2}(U)$.

In this case the function f, that is easily seen to be uniquely determined, is denoted by $\operatorname{div}(v)$. In the case U = X we simply write $v \in D(\operatorname{div}_{loc})$.

The divergence operator just defined satisfies a locality property and a Leibniz rule.

Proposition 1.3.19 (Locality for the divergence). Let $v, w \in D(\operatorname{div}_{loc}, U)$ and assume that v = w holds \mathfrak{m} -a.e. on an open set $\Omega \subset U$. Then $\operatorname{div}(v) = \operatorname{div}(w)$ holds \mathfrak{m} -a.e. on Ω . **Proposition 1.3.20** (Leibniz rule for the divergence). Let $v \in D(\operatorname{div}_{loc}, U)$ and $f \in \operatorname{LIP}(U) \cap L^{\infty}(U)$. Then $fv \in D(\operatorname{div}_{loc}, U)$ and

$$\operatorname{div}(fv) = \operatorname{d}f(v) + f\operatorname{div}(v)$$

holds \mathfrak{m} -a.e. in U.

The Laplacian operator can be defined as the divergence of the gradient. We note that, in order to define the Laplace operator, infinitesimal Hilbertianity is a necessary condition.

Definition 1.3.21 (Laplacian). Let (X, d, \mathfrak{m}) be infinitesimally Hilbertian, $U \subset X$ open and $f \in W^{1,2}_{loc}(X)$. We say that $f \in D(\Delta_{loc}, U)$ provided $\nabla f \in D(\operatorname{div}_{loc}, U)$, i.e. if there is $h \in L^2_{loc}(X)$ such that

$$\int \langle \nabla f, \nabla \varphi \rangle \, \mathrm{d}\mathfrak{m} = -\int \varphi h \, \mathrm{d}\mathfrak{m}$$

for every $\varphi \in W_0^{1,2}(U)$.

In this case the function h is denoted by Δf . In the case U = X we simply write $f \in D(\Delta_{loc})$.

The Laplacian operator obviously inherits the following calculus rules from the ones of gradient and divergence.

Proposition 1.3.22 (Chain rule for the Laplacian). Let (X, d, \mathfrak{m}) be infinitesimally Hilbertian and $U \subset X$ open. For every $f \in D(\Delta_{loc}, U) \cap LIP(U)$ and $\varphi \in C^2(\mathbb{R})$ with $\varphi'' \in L^{\infty}(\mathbb{R})$ it holds $\varphi \circ f \in D(\Delta_{loc}, U)$ and

$$\Delta(\varphi \circ f) = \varphi' \circ f\Delta f + \varphi'' \circ f |\nabla f|^2.$$
(1.3.1)

Proposition 1.3.23 (Leibniz rule for the Laplacian). Let (X, d, \mathfrak{m}) be infinitesimally Hilbertian and $U \subset X$ open. Let $f, g \in D(\Delta_{loc}, U) \cap LIP(U) \cap L^{\infty}(U)$. Then $fg \in D(\Delta_{loc}, U)$ and

$$\Delta(fg) = f\Delta g + g\Delta f + 2\langle \nabla f, \nabla g \rangle. \tag{1.3.2}$$

Remark 1.3.24. In general the set $D(\Delta_{loc}) \cap LIP(X)$ could contain only constant functions. However this does not happen in the RCD setting thanks to the Bakry-Émery inequality and the regularizing properties of the heat flow. For the details see for instance [AGS14b].

1.4 Pullback of 1-forms and speed of a test plan

The aim of this section is to give a definition of speed of a test plan and a chain rule for it.

All the proofs of this section (except where indicated otherwise) can be found in [G18b] and [G18a].

We start defining maps of bounded compression, i.e. maps for which the push-forward measure (Definition 1.1.7) does not concentrate the mass too much, and pullback modules with respect to maps of bounded compression.

Definition 1.4.1 (Maps of bounded compression). Let (X, \mathfrak{m}_X) and (Y, \mathfrak{m}_Y) be measured spaces. We say that $\varphi : Y \to X$ is a map of bounded compression if there exists C > 0 such that $\varphi_* \mathfrak{m}_Y \leq C \mathfrak{m}_X$.

The least such constant C is called compression constant and is denoted by $\operatorname{Comp}(\varphi)$.

The pullback module is defined via the following existence and uniqueness theorem.

Theorem 1.4.2 (Pullback module). Let $(X, \mathsf{d}_X, \mathfrak{m}_X)$ and $(Y, \mathsf{d}_Y, \mathfrak{m}_Y)$ be metric measure spaces. Let \mathscr{M} be an $L^2(\mathfrak{m}_X)$ -normed $L^{\infty}(\mathfrak{m}_X)$ -module and let $\varphi : Y \to X$ be a map of bounded compression. Then there exists a unique couple $(\varphi^*\mathscr{M}, [\varphi^* \cdot])$ where $\varphi^*\mathscr{M}$ is an $L^2(\mathfrak{m}_Y)$ -normed $L^{\infty}(\mathfrak{m}_Y)$ -module and $[\varphi^* \cdot] : \mathscr{M} \to \varphi^*\mathscr{M}$ is a linear and continuous operator such that

- i) $|[\varphi^*v]| = |v| \circ \varphi$ holds \mathfrak{m}_Y -a.e. for every $v \in \mathscr{M}$;
- ii) the set $\{[\varphi^* v] : v \in \mathcal{M}\}\$ generates $\varphi^* \mathcal{M}$.

We note that for every test plan π the evaluation map e_t has bounded compression (by item (i) of Definition 1.2.1), then Theorem 1.4.2 guarantees the existence and uniqueness of the pullback module $L^2(TX)$. We define the speed of a test plan as follows.

Theorem 1.4.3 (Speed of a test plan). Let (X, d, \mathfrak{m}) be an infinitesimally Hilbertian metric measure space and let π be a test plan on X. Then for a.e. $t \in [0, 1]$ there exists a unique $\pi'_t \in e^*_t L^2(TX)$ such that for every $f \in W^{1,2}(X)$ it holds

$$L^{1}(\boldsymbol{\pi}) - \lim_{h \to 0} \frac{f \circ e_{t+h} - f \circ e_{t}}{h} = [e_{t}^{*} \mathrm{d}f](\boldsymbol{\pi}_{t}').$$

Moreover $|\boldsymbol{\pi}_t'|(\gamma) = |\dot{\gamma}_t|$ for $(\boldsymbol{\pi} \times \mathcal{L}^1)$ -a.e. (γ, t) .

We shall refer to π'_t as speed of π at time t.

In order to write a sort of chain rule for the speed of test plans we need to restrict to Lipschitz maps of bounded compression and say what their differential is.

Definition 1.4.4 (Maps of bounded deformation). Given two metric measure spaces (X, d_X, \mathfrak{m}_X) and (Y, d_Y, \mathfrak{m}_Y) , we say that $\varphi : Y \to X$ is a map of bounded deformation if it is both Lipschitz continuous and of bounded compression.

Theorem 1.4.5 (Pullback of 1-forms). Let $(X, \mathsf{d}_X, \mathfrak{m}_X)$ and $(Y, \mathsf{d}_Y, \mathfrak{m}_Y)$ be metric measure spaces and $\varphi : Y \to X$ of bounded deformation. Then there exists a unique linear and continuous operator $\varphi^* : L^2(T^*X) \to L^2(T^*Y)$ such that

i) $\varphi^* df = d(f \circ \varphi)$ for every $f \in S^2(\mathbf{X})$;

ii) $\varphi^*(g\omega) = g \circ \varphi \ \varphi^* \omega \text{ for every } g \in L^{\infty}(\mathfrak{m}_X) \text{ and } \omega \in L^2(T^*X).$

Moreover

$$|\varphi^*\omega| \le \operatorname{Lip}(\varphi)|\omega| \circ \varphi$$

holds \mathfrak{m}_{Y} -a.e. for every $\omega \in L^{2}(T^{*}X)$.

Theorem 1.4.6 (Differential of maps of bounded deformation). Let X and Y be metric measure spaces with X infinitesimally Hilbertian, and $\varphi : Y \rightarrow$ X of bounded deformation. Then there exists a unique $L^{\infty}(\mathfrak{m}_Y)$ -linear and continuous map $d\varphi : L^2(TY) \rightarrow \varphi^* L^2(TX)$ such that

$$[\varphi^*\omega](\mathrm{d}\varphi(v)) = \varphi^*\omega(v)$$

for every $v \in L^2(TY)$ and $\omega \in L^2(T^*X)$.

Moreover, for every $v \in L^2(TY)$ it holds \mathfrak{m}_Y -a.e. that

$$|\mathrm{d}\varphi(v)| \le \mathrm{Lip}(\varphi)|v|.$$

In case φ is invertible with inverse of bounded compression we can express the differential of φ as a function from $L^2(TY)$ to $L^2(TX)$ denoting with $d\varphi(v)$ the map $\omega \mapsto (\varphi^*\omega(v)) \circ \varphi^{-1}$. The precise statement is the following.

Proposition 1.4.7. Let $(X, \mathsf{d}_X, \mathfrak{m}_X)$ and $(Y, \mathsf{d}_Y, \mathfrak{m}_Y)$ be two metric measure spaces and let $\varphi : Y \to X$ be an invertible map of bounded deformation and assume that its inverse has bounded compression. Then there exists a unique linear continuous operator $d\varphi : L^2(TY) \to L^2(TX)$ such that

$$\omega(\mathrm{d}\varphi(v)) = (\varphi^*\omega(v)) \circ \varphi^{-1} \qquad holds \ \mathfrak{m}_{\mathrm{X}}\text{-}a.e.$$

for every $v \in L^2(TY)$ and $\omega \in L^2(T^*X)$.

Moreover

 $|\mathrm{d}\varphi(v)| \leq \mathrm{Lip}(\varphi)|v| \circ \varphi^{-1}$ holds $\mathfrak{m}_{\mathrm{X}}$ -a.e.

for every $v \in L^2(TY)$.

We conclude this section recalling the chain rule for speeds of test plans (for its proof see [DPG, Proposition 3.28]).

With a slight abuse of notation we still denote with $F : C([0, 1], Y) \to C([0, 1], X)$ the map such that $(F(\gamma))_t = F(\gamma_t)$ for every $t \in [0, 1]$ and every $\gamma \in C([0, 1], Y)$. Given a test plan π on Y, assuming that F is of bounded deformation, it is easy to see that $F_*\pi$ is a test plan on X.

Let $F : Y \to X$ be of bounded deformation, invertible and with inverse of deformation and let π be a test plan on Y. For every $t \in [0,1]$ the differential $dF : L^2(TY) \to L^2(TX)$ defined in Proposition 1.4.7 naturally induces a map $e_t^*L^2(TY) \to e_t^*L^2(TX)$, i.e. the unique linear and continuos map, still denoted by dF, such that

$$dF(e_t^*v) = e_t^*(dF(v)) \qquad \text{for every } v \in L^2(TY), dF(gV) = g \circ FdF(V) \qquad \text{for every } V \in e_t^*L^2(TY), \ g \in L^\infty(\pi).$$

Proposition 1.4.8 (Chain rule for speeds). Let $F : X \to Y$ be a map of bounded deformation, invertible and with inverse with bounded deformation. Then, for every test plan π on X we have

$$(F_*\pi)'_t = (dF)(\pi'_t), \quad for \ a.e. \ t \in [0,1].$$

1.5 RCD spaces

In this section we show the two main intrinsic approaches to define the Riemannian Curvature-Dimension condition:

- -) via optimal transport and convexity of the entropy;
- -) via differential calculus and Bochner inequality.

For the second approach, due to L. Ambrosio, N. Gigli and G. Savaré (see [AGS15]) we already recalled all the needed notions in the previous sections.

Definition 1.5.1 (RCD space via Bochner inequality). Let $N \in [1, +\infty)$ and $K \in \mathbb{R}$. We say that the metric measure space (X, d, \mathfrak{m}) is an RCD(K, N) space if the following properties are satisfied:

- -) the space (X, d, \mathfrak{m}) is infinitesimally Hilbertian;
- -) there exists a constant C > 0 such that $\mathfrak{m}(B_r(x)) \leq e^{Cr^2}$ for every r > 0 and for every $x \in X$;
- -) the Sobolev-to-Lipschitz property holds, i.e. every $f \in W^{1,2}(X)$ with |Df| < C admits a C-Lipschitz representative;
- -) for every $f \in D(\Delta)$ such that $\Delta f \in W^{1,2}(X)$ and every $q \in D(\Delta) \cap$ $L^{\infty}(\mathfrak{m})$ such that $g \geq 0$ and $\Delta g \in L^{\infty}(\mathfrak{m})$ the weak Bochner inequality holds, i.e.

$$\frac{1}{2} \int |Df|^2 \Delta g \,\mathrm{d}\mathfrak{m} \ge \int \left[\frac{(\Delta f)^2}{N} + \langle \nabla f, \nabla \Delta f \rangle + K |Df|^2\right] g \,\mathrm{d}\mathfrak{m}.$$
(1.5.1)

The same definition can be extended to the case $N = +\infty$ simply deleting the term $\frac{(\Delta f)^2}{N}$ in the right-hand-side of (1.5.1). We talk about the first approach in the following section.

1.5.1Wasserstein distance and CD conditions

In this section we recall some basic tools of optimal transport, use them to give three definitions of curvature-dimension condition $(CD/CD^*/CD^e)$ and see how they are related. They were introduced by K.-T. Sturm, J. Lott and C. Villani (CD) in [S06a], [S06b], [LV], by K. Bacher and K.-T. Sturm (CD^*) in [BaSt] and by M. Erbar, K. Kuwada and K.-T. Sturm (CD^e) in [EKS]. The "R" to obtain RCD spaces were added by L. Ambrosio, N. Gigli and G. Savaré in [AGS14b] in the case $N = +\infty$ and by N. Gigli in [G15] for the finite dimensional case, and it adds the requirement that the space is infinitesimally Hilbertian.

We start defining the space of measures with finite second moment and the Wasserstein distance on it.

Definition 1.5.2 (Measure with finite second moment). Let (X, d) be a complete and separable metric space. We indicate with $\mathscr{P}_2(X)$ the set of probability measures $\mu \in \mathscr{P}(\mathbf{X})$ such that

$$\int_{\mathcal{X}} \mathsf{d}^2(x,y) \, \mathrm{d}\mu(y) < +\infty$$

for some $x \in X$ (it is easy to see that if it holds for a point $x \in X$ then it holds for every point). A measure $\mu \in \mathscr{P}_2(X)$ is said to have finite second moment.

Definition 1.5.3 (Wasserstein distance). Let $\mu, \nu \in \mathscr{P}_2(X)$. Denoting with π^1 and π^2 the projections on the first and the second coordinate respectively, $\pi^1(x, y) := x$ and $\pi^2(x, y) := y$, we define

$$W_2^2(\mu,\nu) := \inf \int \mathsf{d}^2(x,y) \,\mathrm{d}\boldsymbol{\pi}(x,y), \tag{1.5.2}$$

where the infimum is taken among all $\boldsymbol{\pi} \in \mathscr{P}(\mathbf{X}^2)$ such that $\pi^1_* \boldsymbol{\pi} = \mu$ and $\pi^2_* \boldsymbol{\pi} = \nu$.

It turns out that the infimum in (1.5.2) is always attained, i.e. it is actually a minimum.

Proposition 1.5.4 (Wasserstein space). The map $W_2 : \mathscr{P}_2(X) \times \mathscr{P}_2(X) \to \mathbb{R}$ is a distance and $(\mathscr{P}_2(X), W_2)$, called Wasserstein space, is complete and separable. Moreover, if (X, d) is geodesic then the Wasserstein space is geodesic too.

The Ricci curvature bound from below by K comes from a convexity property (depending on the bound K) of the Boltzmann-Shannon entropy along Wasserstein geodesics.

Definition 1.5.5 (Boltzmann-Shannon relative entropy). Let (X, d, \mathfrak{m}) be a metric measure space such that there exist $c_1, c_2 > 0$ and $x \in X$ for which

$$\mathfrak{m}(B_r(x)) \le c_1 e^{c_2 r} \qquad for \ every \ r > 0. \tag{1.5.3}$$

We define the Boltzmann-Shannon entropy for every $\mu \in \mathscr{P}_2(X)$ as

$$\operatorname{Ent}_{\mathfrak{m}}(\mu) := \begin{cases} \int_{\mathcal{X}} \rho \log(\rho) \, \mathrm{d}\mathfrak{m} & \text{if } \mu = \rho \mathfrak{m}; \\ +\infty & otherwise. \end{cases}$$
(1.5.4)

Remark 1.5.6. Condition (1.5.3) is satisfied in every CD space, we introduce this as further assumption here in order to give a good definition of entropy for spaces with infinite mass too.

Definition 1.5.7 (K-convexity). Let (X, d) be a metric space. We say that a functional $E: X \to \mathbb{R} \cup \{+\infty\}$ is K-convex if for every $x, y \in E^{-1}(\mathbb{R})$ there exists a geodesic γ connecting them such that

$$E(\gamma_t) \le (1-t)E(x) + tE(y) - \frac{1}{2}Kt(1-t)d^2(x,y)$$

for every $t \in [0, 1]$.

In the smooth setting, as proved in [SvR] (see also [OV] and [CEMS] for preliminary results), *K*-convexity of the relative entropy is equivalent to the lower bound on the Ricci curvature.

Theorem 1.5.8. Let M be a smooth, connected and complete Riemannian manifold and let $K \in \mathbb{R}$. Then the following are equivalent:

- i) for every $x \in M$ and $v \in T_x M$ it holds $\operatorname{Ric}_x(v, v) \geq K |v|^2$;
- ii) the relative entropy Ent_{vol} is K-convex on $(\mathscr{P}_2(X), W_2)$.

Property (ii) can be formulated in any metric measure space, and this motivates the following definition.

Definition 1.5.9 ($CD(K, \infty)$ condition). Let (X, d, \mathfrak{m}) be a metric measure space and let $K \in \mathbb{R}$. We say that X is a $CD(K, \infty)$ space if the Boltzmann-Shannon relative entropy defined in (1.5.4) is K-convex on ($\mathscr{P}_2(X), W_2$).

The definition of CD(K, N) spaces with finite dimension bound N requires more work. We recall three different definitions of it and see the relations between them.

To begin we define the volume distortion coefficients τ and the coefficients σ .

For every $\kappa \in \mathbb{R}$ and $\vartheta \geq 0$ we denote by \mathfrak{s}_{κ} the solution of the Cauchy problem

$$f'' + \kappa f = 0,$$
 $f(0) = 0$ and $f'(0) = 1,$

and can be written explicitly as

$$\mathfrak{s}_{\kappa}(\vartheta) := \begin{cases} \frac{1}{\sqrt{\kappa}} \sin(\sqrt{\kappa}\vartheta) & \text{if } \kappa > 0, \\ \vartheta & \text{if } \kappa = 0, \\ \frac{1}{\sqrt{-\kappa}} \sinh(\sqrt{-\kappa}\vartheta) & \text{if } \kappa < 0. \end{cases}$$

Definition 1.5.10 (Volume distortion coefficients). For $t \in [0, 1]$, $K \in \mathbb{R}$ and $N \in (0, +\infty)$ we define

$$\sigma_{K,N}^{(t)}(\vartheta) := \begin{cases} \frac{\mathfrak{s}_{K/N}(t\vartheta)}{\mathfrak{s}_{K/N}(\vartheta)} & \text{if } \frac{K}{N}\vartheta^2 \neq 0 \text{ and } \frac{K}{N}\vartheta^2 < \pi^2, \\ t & \text{if } \frac{K}{N}\vartheta^2 = 0, \\ +\infty & \text{if } \frac{K}{N}\vartheta^2 \geq \pi^2 \end{cases}$$
(1.5.5)

and for $t \in [0, 1]$, $K \in \mathbb{R}$ and $N \in (1, +\infty)$

$$\tau_{K,N}^{(t)}(\vartheta) := t^{\frac{1}{N}} \left(\sigma_{K,N-1}^{(t)}(\vartheta) \right)^{\frac{N-1}{N}}.$$
(1.5.6)

The $\mathsf{CD}(K, N)$ condition is a sort of convexity for the Rényi entropy functional defined by

$$S_N(\mu|\mathfrak{m}) := -\int_{\mathcal{X}} \rho^{-\frac{1}{N}} d\mu,$$

where ρ is the density of the absolutely continuous part of μ with respect to \mathfrak{m} .

For the definition of CD(K, N) space we will focus on optimal geodesic plans, which are geodesics connecting μ and ν with respect to the Wasserstein distance.

Definition 1.5.11 (Optimal geodesic plan). Let $\mu, \nu \in \mathscr{P}_2(X)$ be such that $W_2(\mu, \nu) < +\infty$. We say that $\pi \in \mathscr{P}(\text{Geo}(X))$ is an optimal geodesic plan between μ and ν if $(e_0)_*\pi = \mu$, $(e_1)_*\pi = \nu$ and

$$\int \mathsf{d}^2(\gamma_0,\gamma_1) \,\mathrm{d}\boldsymbol{\pi} = \int \int_0^1 |\dot{\gamma}_s|^2 \,\mathrm{d}s \,\mathrm{d}\boldsymbol{\pi}(\gamma) = W_2^2(\mu,\nu).$$

We indicate with $OptGeo(\mu, \nu)$ the set of optimal geodesic plans connecting μ and ν .

Definition 1.5.12 (Curvature-Dimension condition). Let $K \in \mathbb{R}$ and $N \in [1, \infty)$. We say that a metric measure space $(X, \mathsf{d}, \mathfrak{m})$ is $\mathsf{CD}(K, N)$ if for every $\mu_0, \mu_1 \ll \mathfrak{m}$ with bounded support there exists π optimal geodesic plan between μ_0 and μ_1 such that, indicating with $\rho_t \mathfrak{m}$ the absolutely continuous part with respect to \mathfrak{m} of $(e_t)_*\pi$, it holds

$$-\int \rho_t^{1-\frac{1}{N'}} d\mathfrak{m} \leq -\int \tau_{K,N'}^{(1-t)} (\mathsf{d}(\gamma_0,\gamma_1)) \rho_0^{-\frac{1}{N'}}(\gamma_0) + \tau_{K,N'}^{(t)} (\mathsf{d}(\gamma_0,\gamma_1)) \rho_1^{-\frac{1}{N'}}(\gamma_1) d\pi(\gamma)$$
(1.5.7)

for every $t \in [0, 1]$ and $N' \ge N$.

The following consistency result can be found in [S06b].

Theorem 1.5.13. Let M be a complete connected Riemannian manifold with Riemannian distance d and Riemannian volume \mathfrak{m} , and let $K \in \mathbb{R}$, $N \in [1, +\infty)$. Then the following are equivalent:

- i) the metric measure space (M, d, \mathfrak{m}) satisfies the CD(K, N) condition;
- ii) for every $x \in M$ and $v \in T_x M$ it holds $\operatorname{Ric}_x(v, v) \geq K|v|^2$ and M has dimension at most N.
In order to study a local-to-global property for the CD(K, N) condition, K. Bacher and K.-T. Sturm in [BaSt] introduced a weaker notion of curvaturedimension condition, the so called reduced curvature-dimension condition, indicated with $CD^*(K, N)$.

Definition 1.5.14 (Reduced Curvature-Dimension condition). Let $K \in \mathbb{R}$ and $N \in (0, \infty)$. We say that a metric measure space (X, d, \mathfrak{m}) is $CD^*(K, N)$ if for every $\mu_0, \mu_1 \ll \mathfrak{m}$ with bounded support there exists π optimal geodesic plan between μ_0 and μ_1 such that, indicating with $\rho_t \mathfrak{m}$ the absolutely continuous part with respect to \mathfrak{m} of $(e_t)_*\pi$, it holds

$$-\int \rho_t^{1-\frac{1}{N'}} d\mathfrak{m} \leq -\int \sigma_{K,N'}^{(1-t)} (\mathsf{d}(\gamma_0,\gamma_1)) \rho_0^{-\frac{1}{N'}}(\gamma_0) + \sigma_{K,N'}^{(t)} (\mathsf{d}(\gamma_0,\gamma_1)) \rho_1^{-\frac{1}{N'}}(\gamma_1) d\boldsymbol{\pi}(\gamma)$$
(1.5.8)

for every $t \in [0,1]$ and $N' \ge N$.

The only difference between the definition of CD spaces and CD^{*} spaces is that in (1.5.7) we used the coefficients $\tau_{K,N}$ as weights and in (1.5.8) we changed them with the coefficients $\sigma_{K,N}$.

From Definitions 1.5.12, 1.5.14 and 1.5.10 it follows easily that CD(0, N) condition and $CD^*(0, N)$ condition are equivalent, indeed

$$\tau_{0,N}^{(t)}(\vartheta) = \sigma_{0,N}^{(t)}(\vartheta) = t.$$

From an easy computation we conclude that, for every $K, \vartheta \in \mathbb{R}$ and $N \in (1, +\infty)$ it holds

$$\sigma_{K,N}^{(t)}(\vartheta) \le \tau_{K,N}^{(t)}(\vartheta), \qquad (1.5.9)$$

and if K > 0, defining $K^* := K \frac{N-1}{N}$, it holds

$$\tau_{K^*,N}^{(t)}(\vartheta) \le \sigma_{K,N}^{(t)}(\vartheta). \tag{1.5.10}$$

By Definitions 1.5.12, 1.5.14 and inequality (1.5.9) we conclude easily that

$$\mathsf{CD}(K,N) \Rightarrow \mathsf{CD}^*(K,N),$$

and for K > 0, using (1.5.10), that

$$\mathsf{CD}^*(K,N) \Rightarrow \mathsf{CD}(K^*,N).$$

We recall the definition of essentially non-branching space given in [RS].

Definition 1.5.15 (Essentially non-branching space). A metric measure space (X, d, m) is said essentially non-branching if for every $\mu, \nu \in \mathscr{P}_2(X)$ absolutely continuous with respect to the reference measure \mathfrak{m} any element of OptGeo (μ, ν) is concentrated on a set of non-branching geodesics, i.e. given $\pi \in \text{OptGeo}(\mu, \nu)$ there exists $\Gamma \subset C([0, 1], X)$ Borel such that $\pi(\Gamma) = 1$ and if $\gamma, \eta \in \Gamma$ coincide on the interval [0, t] for some $t \in (0, 1]$ then $\gamma \equiv \eta$.

In [CaMi] F. Cavalletti and E. Milman proved that the CD(K, N) condition is equivalent to the $CD^*(K, N)$ condition for normalized essentially non-branching spaces. A proof for spaces with infinite mass is still missing.

The third notion of curvature-dimension condition we recall here is due to M. Erbar, K. Kuwada and K.-T. Sturm (see [EKS]).

We indicate with $\mathscr{P}_2^*(\mathbf{X}, \mathbf{d}, \mathbf{m})$ the set of probability measures on X with finite entropy, and with $U_N : \mathscr{P}_2(\mathbf{X}) \to [0, +\infty]$ the map

$$U_N(\mu) := \exp\left(-\frac{1}{N}\operatorname{Ent}_{\mathfrak{m}}(\mu)\right).$$

We define $\mathsf{CD}^e(K, N)$ spaces as the spaces for which the entropy satisfies a (K, N)-convexity condition in the following way.

Definition 1.5.16 (Entropic Curvature Dimension condition). Given $K \in \mathbb{R}$ and $N \in (0, +\infty)$ we say that a metric measure space $(X, \mathsf{d}, \mathfrak{m})$ satisfies the entropic curvature-dimension condition $\mathsf{CD}^e(K, N)$ if and only if for each pair $\mu_0, \mu_1 \in \mathcal{P}_2^*(X, \mathsf{d}, \mathfrak{m})$ there exists a constant speed geodesic $(\mu_t)_{t \in [0,1]} \subset \mathcal{P}_2^*(X, \mathsf{d}, \mathfrak{m})$ connecting μ_0 to μ_1 such that for all $t \in [0, 1]$ it holds

$$U_N(\mu_t) \ge \sigma_{K/N}^{(1-t)}(W_2(\mu_0,\mu_1))U_N(\mu_0) + \sigma_{K/N}^{(t)}(W_2(\mu_0,\mu_1))U_N(\mu_1).$$

In [EKS] it is proven that for essentially non-branching metric measure spaces the $CD^*(K, N)$ condition and the $CD^e(K, N)$ condition are equivalent, moreover, a major result from the same paper, is the equivalence between the CD^e condition and the weak formulation of the Bochner inequality (1.5.1).

Theorem 1.5.17. Let (X, d, \mathfrak{m}) be a metric measure space. The following are equivalent:

- i) the space (X, d, m) is RCD(K, N), i.e. it satisfies the conditions of Definition 1.5.1;
- ii) the space (X, d, m) is infinitesimally Hilbertian and satisfies the entropic curvature-dimension condition $CD^e(K, N)$.

In [RS] T. Rajala and K.-T. Sturm proved that every infinitesimally Hilbertian $CD(K, \infty)$ space is essentially non-branching. From this we conclude that, at least for normalized spaces, if we define $RCD/RCD^*/RCD^e$ spaces as $"CD/CD^*/CD^e$ +infinitesimal Hilbertianity" the three definitions are equivalent. Given that, slightly abusing the notation, we will not distinguish between RCD, RCD^* and RCD^e conditions.

Remark 1.5.18. In [D] Q. Deng proved that $\mathsf{RCD}(K, N)$ spaces are nonbranching, i.e. if $\gamma, \eta \in \operatorname{Geo}(X)$ coincide on the interval [0, t] for some $t \in (0, 1]$ then $\gamma \equiv \eta$.

As anticipated in the Introduction, an important result holding true in every $\mathsf{CD}(K, N)$ spaces is the Bishop-Gromov inequality. Its proof can be found in [S06b].

Theorem 1.5.19 (Bishop-Gromov Inequality). Let $(X, \mathsf{d}, \mathfrak{m})$ be a $\mathsf{CD}(K, N)$ metric measure space with $K \in \mathbb{R}$ and $N \in (1, +\infty)$. Fix $p \in X$ and let $\mu := \mathsf{d}(\cdot, p)_*\mathfrak{m}$. Then $\mu = s\mathcal{L}^1$ for some function $s : [0, \operatorname{diam}(X)] \to \mathbb{R}^+$ such that

$$r \mapsto s(r) \left(\mathfrak{s}_{\frac{K}{N-1}}(r)\right)^{1-N}$$
 is not increasing on $[0, \operatorname{diam}(X)].$ (1.5.11)

Also, the following integrated version of the monotonicity holds:

$$R \mapsto \mathfrak{m}(B_R(p)) \left(\int_0^R \left(\mathfrak{s}_{\frac{K}{N-1}}(r) \right)^{N-1} \mathrm{d}r \right)^{-1} \text{ is not increasing on } [0, \operatorname{diam}(\mathbf{X})].$$
(1.5.12)

The Bishop-Gromov inequality for volumes (1.5.12) estimates the ratio of the volume of balls, then an easy consequence is that every $\mathsf{CD}(K, N)$ space is uniformly locally doubling.

Definition 1.5.20 (locally doubling space). We say that a metric measure space (X, d, \mathfrak{m}) is locally doubling if for every $x \in X$ there are an open set $U \ni x$ and constants C, R > 0 such that

$$\mathfrak{m}(B_{2r}(y)) \le C\mathfrak{m}(B_r(y))$$

for every $y \in U$ and $r \in (0, R)$.

A space is said uniformly locally doubling if constants C and R do not depend on x.

1.5.2 Convergence of (pointed) metric measure spaces and stability

An important property of the class of CD(K, N) spaces is its stability. In literature several different convergences of metric measure spaces (and pointed metric measure spaces) have been studied in this direction, for instance:

- -) in [S06a] and [S06b] Sturm defined a distance between normalized metric measure spaces and proved that curvature-dimension bounds are stable with respect to the convergence induced by that distance;
- -) in [LV] Lott and Villani worked with proper pointed metric measure spaces and pointed measured Gromov Hausdorff (pmGH) convergence for CD(K, N) spaces (note that for $N < +\infty$ a CD(K, N) space is always proper, but this is not true for $N = +\infty$) (see also [V]);
- -) in [GMS] Gigli, Mondino and Savaré introduced the so called pointed measured Gromov (pmG) convergence to prove the stability in case the space has infinite mass (and is not necessarily proper), so to cover also the case $CD(K, \infty)$. Then they proved that for uniformely doubling spaces pmG and pmGH convergences coincide.

We recall the needed definitions and the compactness result for pmGH convergence of CD(K, N) spaces (see [GMS] for the proof of the stability theorem and for more details about the convergence of metric measure spaces).

Definition 1.5.21 (Pointed metric measure space). A pointed metric measure space (X, d, m, x) is a quadruple where (X, d, m) is a metric measure space (as in Definition 1.1.9) and x is a fixed point of supp(m).

Definition 1.5.22 (Pointed measured Gromov Hausdorff convergence). Let $((X_n, \mathsf{d}_n, \mathfrak{m}_n, x_n))_{n \in \mathbb{N}}$ be a sequence of pointed metric measure spaces. We say that $((X_n, \mathsf{d}_n, \mathfrak{m}_n, x_n))_{n \in \mathbb{N}}$ converges in the pointed measured Gromov Hausdorff sense to $(X_{\infty}, \mathsf{d}_{\infty}, \mathfrak{m}_{\infty}, x_{\infty})$ provided that there exist two sequences $R_n \uparrow +\infty$ and $\varepsilon_n \downarrow 0$ and Borel maps $f_n : X_n \to X_\infty$ such that the following hold:

- -) $f_n(x_n) = x_\infty;$
- -) $\sup_{x,y\in B_{R_n}(x_n)} |\mathsf{d}_n(x,y) \mathsf{d}_\infty(f_n(x),f_n(y))| \le \varepsilon_n;$
- -) the ε_n -neighborhood of $f_n(B_{R_n}(x_n))$ contains $B_{R_n-\varepsilon_n}(x_\infty)$;
- -) for every $\varphi \in C_{bs}(X_{\infty})$ it holds

$$\lim_{n\to\infty}\int\varphi\circ f_n\,\mathrm{d}\mathfrak{m}_n=\int\varphi\,\mathrm{d}\mathfrak{m}_\infty.$$

We denote this convergence with $(X_n, \mathsf{d}_n, \mathfrak{m}_n, x_n) \xrightarrow{pmGH} (X_\infty, \mathsf{d}_\infty, \mathfrak{m}_\infty, x_\infty).$

Theorem 1.5.23. Let $((X_n, \mathsf{d}_n, \mathfrak{m}_n, x_n))_{n \in \mathbb{N}}$ be a sequence of pointed metric measure spaces such that for every $n \in \mathbb{N}$ the space $(X_n, \mathsf{d}_n, \mathfrak{m}_n)$ is $\mathsf{CD}(K_n, N_n)$ (resp. $\mathsf{RCD}(K_n, N_n)$) for certain $K_n \in \mathbb{R}$ and $N_n \in [1, +\infty)$. Moreover assume that $K_n \to K \in \mathbb{R}$, $N_n \to N \in [1, +\infty)$ and

$$0 < \liminf_{n \to +\infty} \mathfrak{m}_n(B_1(x_n)) \le \limsup_{n \to +\infty} \mathfrak{m}_n(B_1(x_n)) < +\infty.$$

Then $((X_n, \mathsf{d}_n, \mathfrak{m}_n, x_n))_{n \in \mathbb{N}}$ has a subsequence that converges to a $\mathsf{CD}(K, N)$ (resp. $\mathsf{RCD}(K, N)$) space $(X_\infty, \mathsf{d}_\infty, \mathfrak{m}_\infty, x_\infty)$.

1.6 Second order differential calculus

The goal of this section is to give the definition of Hessian and covariant derivative in RCD spaces and see their calculus rules.

We start giving the definition of tensor product of Hilbert modules.

If \mathscr{H}_1^0 and \mathscr{H}_2^0 are the $L^0(\mathfrak{m})$ -completions of two Hilbert modules \mathscr{H}_1 and \mathscr{H}_2 , we indicate with $\mathscr{H}_1^0 \otimes_{Alg} \mathscr{H}_2^0$ the space of formal finite sums of objects of the form $v \otimes w$ with $v \in \mathscr{H}_1^0$ and $w \in \mathscr{H}_2^0$, with $(v, w) \mapsto v \otimes w$ being bilinear. On $\mathscr{H}_1^0 \otimes_{Alg} \mathscr{H}_2^0$ we define a "scalar product" in the following way: for every $v_1, v_2 \in \mathscr{H}_1^0$ and $w_1, w_2 \in \mathscr{H}_2^0$ we define

$$\langle v_1 \otimes w_1, v_2 \otimes w_2 \rangle := \langle v_1, v_2 \rangle \langle w_1, w_2 \rangle \in L^0(\mathfrak{m}),$$

then we extend it by linearity.

The scalar product just defined is symmetric and the following hold:

 $\begin{array}{ll} \langle A, A \rangle \geq 0 & \mathfrak{m}\text{-a.e. for every } A \in \mathscr{H}_1^0 \otimes_{Alg} \mathscr{H}_2^0; \\ \langle A, A \rangle = 0 & \mathfrak{m}\text{-a.e. on } E \text{ if and only if } \chi_E A = 0; \\ f \langle A, B \rangle = \langle fA, B \rangle & \mathfrak{m}\text{-a.e. for every } A, B \in \mathscr{H}_1^0 \otimes_{Alg} \mathscr{H}_2^0 \text{ and } f \in L^0(\mathfrak{m}). \end{array}$

Definition 1.6.1 (Hilbert-Schmidt norm). On $\mathscr{H}_1^0 \otimes_{Alg} \mathscr{H}_2^0$ we define the Hilbert-Schmidt norm as

$$A|_{\mathsf{HS}} := \sqrt{\langle A, A \rangle}.$$

Definition 1.6.2. We define the space $\mathscr{H}_1^0 \otimes \mathscr{H}_2^0$ as the completion of $\mathscr{H}_1^0 \otimes_{Alg} \mathscr{H}_2^0$ with respect to the distance

$$\mathsf{d}_{\otimes}(A,B) := \sum_{i \in \mathbb{N}} \frac{1}{2^{i} \mathfrak{m}(E_{i})} \int (|A - B|_{\mathsf{HS}} \wedge 1) \, \mathrm{d}\mathfrak{m}_{A}$$

where $(E_i)_{i \in \mathbb{N}}$ is a Borel partition of X in sets of positive measure.

Definition 1.6.3 (Tensor product of Hilbert modules). The tensor product $\mathscr{H}_1 \otimes \mathscr{H}_2$ is defined as the subset of $\mathscr{H}_1^0 \otimes \mathscr{H}_2^0$ with $L^2(\mathfrak{m})$ Hilbert-Schmidt norm.

We observe that the space $\mathscr{H}_1 \otimes \mathscr{H}_2$ endowed with the norm

$$\|A\| := \sqrt{\int |A|^2_{\mathsf{HS}} \,\mathrm{d}\mathfrak{m}}$$

is a Hilbert space. It easily follows that $\mathscr{H}_1 \otimes \mathscr{H}_2$ is a Hilbert module. We introduce now the following notation.

Definition 1.6.4. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, \infty)$ space. Then we define

$$L^{2}((T^{*})^{\otimes 2}\mathbf{X}) := L^{2}(T^{*}\mathbf{X}) \otimes L^{2}(T^{*}\mathbf{X})$$

Given $A \in L^2((T^*)^{\otimes 2}X)$ we indicate with A(v, w) the function $A(v \otimes w) \in L^0(\mathfrak{m})$ for every $v, w \in L^2(TX)$.

Similarly, we define $L^2(T^{\otimes 2}X) := L^2(TX) \otimes L^2(TX)$.

We observe that $L^2((T^*)^{\otimes 2}X)$ can be seen as the dual of $L^2(T^{\otimes 2}X)$ via the duality map

$$(\omega \otimes \eta)(v \otimes w) := \omega(v)\eta(w)$$

for $v, w \in L^2(TX)$ and $\omega, \eta \in L^2(T^*X)$.

In order to define the space $W_{loc}^{2,2}(\mathbf{X})$ and the Hessian of a function we recall the definition of test functions, which are the most regular functions we can consider in the non-smooth setting, and they will be used instead of the C_c^{∞} functions.

Definition 1.6.5 (Local test functions). Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, \infty)$ space with $K \in \mathbb{R}$. We define the set of local test functions as

$$\operatorname{Test}_{loc}(\mathbf{X}) := \left\{ f \in \operatorname{LIP}_{loc}(\mathbf{X}) \cap L^{\infty}_{loc}(\mathbf{X}) \cap D(\Delta_{loc}) : \Delta f \in W^{1,2}_{loc}(\mathbf{X}) \right\}.$$

From the weak Bochner inequality (1.5.1) follows that for every function $f \in \text{Test}_{loc}(X)$ it holds $|Df| \in W_{loc}^{1,2}(X)$. A consequence of this fact is that the set of local test functions is an algebra and it is dense in $W_{loc}^{1,2}(X)$.

The following result provides the existence of good cut-off functions (for the construction see [MN], for similar results [AMS] or [GP20]).

Proposition 1.6.6 (Good cut-off functions). Let $K \in \mathbb{R}$, $N \in (1, +\infty)$ and let X be an $\mathsf{RCD}(K, N)$ space. Then for every $0 < r < R < +\infty$, every compact $K \subset X$ and every open $U \subset X$ such that $\operatorname{diam}(U) \leq R$ and $\mathsf{d}(P, U^c) > r$ there exists a function $\eta \in \operatorname{Test}_{loc}(X)$ such that

- i) $0 \le \eta \le 1$ on X, $\operatorname{supp}(\eta) \subset U$ and $\eta \equiv 1$ in P,
- ii) $r|D\eta| + r^2 |\Delta\eta| \leq C$, where C depends only on R, N, K.

We give now the definition of $W^{2,2}_{loc}(\mathbf{X})$ and Hessian of a function $f \in W^{2,2}_{loc}(\mathbf{X})$.

Definition 1.6.7 $(W_{loc}^{2,2}(X) \text{ space and Hessian})$. Let $K \in \mathbb{R}$ and let $(X, \mathsf{d}, \mathfrak{m})$ be an $\mathsf{RCD}(K, \infty)$ space. For $f \in W_{loc}^{1,2}(X)$ we say that $f \in W_{loc}^{2,2}(X)$ if there exists $A \in L((T^*)^{\otimes 2}X)$ such that

$$2\int \varphi A(\nabla \psi_1, \nabla \psi_2) \, \mathrm{d}\mathfrak{m} = -\int \langle \nabla f, \nabla \psi_1 \rangle \operatorname{div}(\varphi \nabla \psi_2) + \langle \nabla f, \nabla \psi_2 \rangle \operatorname{div}(\varphi \nabla \psi_1) \\ + \varphi \langle \nabla f, \nabla \langle \nabla \psi_1, \nabla \psi_2 \rangle \rangle \, \mathrm{d}\mathfrak{m}$$

for every $\psi_1, \psi_2 \in \text{Test}_{loc}(X)$ and $\varphi \in \text{LIP}_{bs}(X)$. Such tensor A is uniquely determined and it will be denoted by Hess(f).

Remark 1.6.8. This definition is justified by the following formula on Riemannian manifolds: let M be a Riemannian manifold, then for every functions $f, \psi_1, \psi_2 \in C^{\infty}(\mathbf{M})$ it holds

$$2 \operatorname{Hess}(f)(\psi_1, \psi_2) = \langle \nabla \langle \nabla f, \nabla \psi_1 \rangle, \nabla \psi_2 \rangle + \langle \nabla \langle \nabla f, \nabla \psi_2 \rangle, \nabla \psi_1 \rangle - \langle \nabla \langle \nabla \psi_1, \nabla \psi_2 \rangle, \nabla f \rangle.$$

The existence of many $W_{loc}^{2,2}$ functions is given by the following inclusion result proved in [G18b].

Proposition 1.6.9. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, \infty)$ space with $K \in \mathbb{R}$. Then $\operatorname{Test}_{loc}(X) \subset D(\Delta_{loc}) \subset W^{2,2}_{loc}(X)$.

The set of local test functions is not dense in $W^{2,2}_{loc}(\mathbf{X})$, then we define the space $H^{2,2}_{loc}(\mathbf{X})$ as the closure of $\text{Test}_{loc}(\mathbf{X})$ in $W^{2,2}_{loc}(\mathbf{X})$.

Definition 1.6.10 (The space $H^{2,2}_{loc}(X)$). Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, \infty)$ space with $K \in \mathbb{R}$. We define the space $H^{2,2}_{loc}(X) \subset W^{2,2}_{loc}(X)$ as the collection of all functions $f \in W^{2,2}_{loc}(X)$ such that the following holds: for any $U \subset X$ open bounded there is a sequence $(f_n) \subset \operatorname{Test}_{loc}(X)$ such that

$$||f_n - f||_{L^2(U)} + |||df_n - df|||_{L^2(U)} + ||| \operatorname{Hess} f_n - \operatorname{Hess} f|_{\mathsf{HS}}||_{L^2(U)} \to 0$$

as $n \to \infty$, where here and in what follows we denote by $|\cdot|_{\text{HS}}$ the pointwise norm in $L^2_{loc}(T^{\otimes 2}X)$.

We see now some calculus rules for the Hessian.

Proposition 1.6.11 (Leibniz rule for the Hessian). Let $f, g \in W^{2,2}_{loc}(X) \cap LiP_{loc}(X) \cap L^{\infty}_{loc}(\mathfrak{m})$. Then $fg \in W^{2,2}_{loc}(X)$ and

$$\operatorname{Hess}(fg) = f \operatorname{Hess}(g) + g \operatorname{Hess}(f) + df \otimes dg + dg \otimes df$$

holds m-a.e. in X.

Proposition 1.6.12 (Chain rule for the Hessian). Let $f, g \in W^{2,2}_{loc}(\mathbf{X}) \cap \operatorname{LiP}_{loc}(\mathbf{X})$ and let $\varphi \in C^2(\mathbb{R})$ such that φ' and φ'' are bounded and $\varphi(0) = 0$. Then $\varphi \circ f \in W^{2,2}_{loc}(\mathbf{X})$ and the following formula holds \mathfrak{m} -a.e.:

$$\operatorname{Hess}(\varphi \circ f) = \varphi' \circ f \operatorname{Hess}(f) + \varphi'' \circ f df \otimes df.$$
(1.6.1)

Proposition 1.6.13 (Product rule for gradients). Let $f, g \in H^{2,2}_{loc}(\mathbf{X}) \cap \operatorname{LIP}_{loc}(\mathbf{X})$. Then $\langle \nabla f, \nabla g \rangle \in W^{1,2}_{loc}(\mathbf{X})$ and

$$d\langle \nabla f, \nabla g \rangle = \text{Hess}(f)(\nabla g, \cdot) + \text{Hess}(g)(\nabla f, \cdot)$$
(1.6.2)

holds m-a.e. in X.

As we saw for the Hessian, thanks to an identity on Riemannian manifolds we can define the covariant derivative on RCD spaces. On M Riemannian manifold, for every v vector field and $\varphi, \psi \in C^{\infty}(\mathbf{M})$ it holds

$$\langle \nabla_{\nabla\varphi} v, \nabla\psi \rangle = \langle \nabla \langle v, \nabla\psi \rangle, \nabla\varphi \rangle - \operatorname{Hess}(\psi)(v, \nabla\varphi).$$

We indicate with : the scalar product between two elements of $L^2(T^{\otimes 2}X)$.

Definition 1.6.14 (Covariant derivative). Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, \infty)$ space for a certain $K \in \mathbb{R}$. The space $W_C^{1,2}(TX)$ is the set of vector fields $v \in L^2(TX)$ for which there exists $T \in L^2(T^{\otimes 2}X)$ such that for every $\varphi, \psi, \xi \in$ $\mathrm{Test}(X)$ it holds

$$\int \xi T : (\nabla \varphi \otimes \nabla \psi) \, \mathrm{d}\mathfrak{m} = -\int \langle v, \nabla \psi \rangle \, \mathrm{div}(\xi \nabla \varphi) + \xi \operatorname{Hess}(\psi)(v, \nabla \varphi) \, \mathrm{d}\mathfrak{m}.$$

Such T, which is uniquely determined, is called covariant derivative of v and denoted by ∇v .

Remark 1.6.15. The space $W_C^{1,2}(TX)$ with the norm

$$\|v\|_{W^{1,2}_C(T\mathbf{X})} := \left(\|v\|^2_{L^2(T\mathbf{X})} + \|\nabla v\|^2_{L^2(T^{\otimes 2}\mathbf{X})}\right)^{\frac{1}{2}}$$

is a separable Hilbert space. Moreover, denoting with \sharp the Riesz isomorphism between $L^2((T^*)^{\otimes 2}X)$ and $L^2(T^{\otimes 2}X)$, for every $f \in H^{2,2}(X) \cap LIP(X)$ it holds $\nabla f \in W^{1,2}(TX)$ and $\nabla(\nabla f) = (\text{Hess}(f))^{\sharp}$.

The covariant derivative satisfies the following Leibniz rule.

Proposition 1.6.16 (Leibniz rule for the covariant derivative). Let X be an $\mathsf{RCD}(K,\infty)$ space. Let $v \in W^{1,2}_C(TX) \cap L^{\infty}(TX)$ and $f \in W^{1,2}(X) \cap L^{\infty}(\mathfrak{m})$. Then $fv \in W^{1,2}_C(TX)$ and

$$\nabla(fv) = \nabla f \otimes v + f \nabla v.$$

We conclude this section recalling the definition of test vector field.

Definition 1.6.17 (Test vector field). Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, \infty)$ space with $K \in \mathbb{R}$. We define the set of test vector fields as

$$\operatorname{TestV}(\mathbf{X}) := \left\{ \sum_{i=1}^{n} g_i \nabla f_i : n \in \mathbb{N} \text{ and } f_i, g_i \in \operatorname{Test}(\mathbf{X}) \text{ for } i = 1, ..., n \right\}.$$

Similarly to test functions and $W^{2,2}(X)$, it holds TestV(X) is contained $W_C^{1,2}(TX)$, but it is not dense in it. Then we indicate with $H_C^{1,2}(TX)$ the closure of TestV(X) in $W_C^{1,2}(TX)$.

1.7 Warped product spaces

As we saw in the Introduction, all rigidity theorems state that, under certain hypotheses, the space is isomorphic to a warped product space $\mathbb{R} \times_w X'$. In this section we define warped products of metric measure spaces and see how the Sobolev spaces behave on them. All these results can be found in [GH].

Definition 1.7.1 (Warped kinetic energy of curves). Let (X, d_X) and (Y, d_Y) be length spaces and let $w_\mathsf{d} : Y \to [0, +\infty)$ be a continuous function. For every curve $\gamma := (\gamma^Y, \gamma^X)$ with $\gamma^Y \in AC(Y)$ and $\gamma^X \in AC(X)$ we define its w_d -kinetic energy as

$$\mathrm{KE}_w[\gamma] := \int_0^1 |\dot{\gamma}_t^{\mathrm{Y}}|^2 + w_\mathsf{d}^2(\gamma_t^{\mathrm{Y}}) |\dot{\gamma}_t^{\mathrm{X}}|^2 \,\mathrm{d}t.$$

In a natural way we define the warped distance and the warped product of metric spaces.

Definition 1.7.2 (Warped product of metric spaces). Taken two length spaces (X, d_X) and (Y, d_Y) , let $w_d : Y \to [0, +\infty)$ be a continuous function. For every points $p, q \in X \times Y$ we define the pseudo-metric $d_w(p,q)$ as

$$\mathsf{d}_w(p,q)^2 := \inf \left\{ \mathrm{KE}_w[\gamma] \right\},\,$$

where the infimum is taken among the curve $\gamma = (\gamma^{Y}, \gamma^{X})$ such that $\gamma^{Y} \in AC(Y), \gamma^{X} \in AC(X), \gamma_{0} = p$ and $\gamma_{1} = q$.

We denote with $(Y \times_w X, d_w)$ the completion of the quotient of $Y \times X$ with respect to the equivalence relation induced by the pseudo-metric d_w .

We indicate with $\pi: Y \times X \to Y \times_w X$ the quotient map induced by d_w .

Definition 1.7.3 (Warped product of metric measure spaces). Assume that $(X, \mathsf{d}_X, \mathfrak{m}_X)$ and $(Y, \mathsf{d}_Y, \mathfrak{m}_Y)$ are two complete, separable and length metric spaces equipped with non-negative Radon measures, and let $w_{\mathfrak{m}}, w_{\mathsf{d}} : Y \to [0, \infty)$ be continuous functions. Moreover, assume that at least one of the following holds:

-) $\mathfrak{m}_{\mathbf{X}}$ is a finite measure;

-) w_d is always positive.

The warped product space $(Y \times_w X, \mathsf{d}_w, \mathfrak{m}_w)$ is defined as the warped product of metric spaces $(Y \times_w X, \mathsf{d}_w)$ equipped with the Radon measure \mathfrak{m}_w defined as

$$\mathfrak{m}_w := \pi_* \left((w_{\mathfrak{m}} \mathfrak{m}_{\mathrm{Y}}) \times \mathfrak{m}_{\mathrm{X}} \right)$$

Remark 1.7.4. The alternative conditions in Definition 1.7.3 (\mathfrak{m}_X finite or w_d never 0) are needed to ensure that the measure \mathfrak{m}_w is Radon.

In all the work we indicate with (y, x) the elements of $Y \times_w X$, omitting the quotient map.

We introduce some notations and assumptions that we are going to use along this section.

Assumption 1.7.5. Let (X, d, \mathfrak{m}) be a metric measure space and $I \subset \mathbb{R}$ be a closed and possibly unbounded interval. Given $w_d, w_{\mathfrak{m}} : I \to [0, +\infty)$ continuous functions, assume that at least one of the alternative conditions in Definition 1.7.3 is satisfied. Then we indicate with $(X_w, d_w, \mathfrak{m}_w)$ the warped product space $I \times_w X$ with warping functions w_d and $w_{\mathfrak{m}}$.

The first result we recall from [GH] is the characterization of the Sobolev space on warped product spaces. It relates the Sobolev space of X_w with the ones of the quotient spaces and tells us who the minimal weak upper gradient of a Sobolev function in X_w is.

Theorem 1.7.6. Given Assumption 1.7.5, $f \in W^{1,2}(X_w)$ if and only if the following conditions hold:

i) for \mathfrak{m} -a.e. $x \in X$ the map $f^{(x)} := f(\cdot, x)$ is in $W^{1,2}(\mathbb{R}, \mathsf{d}_{\mathrm{Eucl}}, w_{\mathfrak{m}}\mathcal{L}^1)$;

- ii) for \mathcal{L}^1 -a.e. $r \in \mathbb{R}$ the map $f^{(r)} := f(r, \cdot)$ is in $W^{1,2}(\mathbf{X}, \mathsf{d}, \mathfrak{m})$;
- *iii)* the identity

$$|Df|_{\mathbf{X}_{w}}^{2}(r,x) = |Df^{(x)}|_{\mathbb{R}}^{2}(r) + \frac{1}{w_{\mathsf{d}}^{2}(r)}|Df^{(r)}|_{\mathbf{X}}^{2}(x)$$
(1.7.1)

holds for \mathfrak{m}_w -a.e. (r, x).

The proof of points (i) and (ii) is an easy consequence of Theorem 1.2.16 and Fubini's Theorem, while the proof of point (iii) follows from a nontrivial comparison argument with the Cartesian product case $(w_m = w_d = 1)$.

The second result we recall from [GH] studies the Sobolev-to-Lipschitz property for warped product spaces.

In general it is not true that if X has the Sobolev-to-Lipschitz property then the warped product space $I \times_w X$ has it too. For instance it is trivial to observe that if $w_{\mathfrak{m}}$ is 0 on some interval $J \subset I$ which disconnects Ithen \mathfrak{m}_w has disconnected support, and this violates the Sobolev-to-Lipschitz property. Even assuming $w_{\mathfrak{m}}$ strictly positive it is not clear if the Sobolevto-Lipschitz property passes to the warped product space or not. In order to prove the Sobolev-to-Lipschitz property for the warped product space then Gigli and Han introduced a variant of the length property which takes into account the measure.

Definition 1.7.7 (Measured-length space). We say that a metric measure space (X, d, m) is a measured-length space if there exists a Borel set $A \subset$ X whose complement is \mathfrak{m} -negligible with the following property: for every $x_0, x_1 \in A$ there exists $\varepsilon > 0$ such that for every $\varepsilon_0, \varepsilon_1 \in (0, \varepsilon]$ there is a test plan $\pi^{\varepsilon_0, \varepsilon_1}$ such that

-) the map $(0,\varepsilon]^2 \ni (\varepsilon_0,\varepsilon_1) \mapsto \pi^{\varepsilon_0,\varepsilon_1}$ is weakly Borel in the sense that for any $\varphi \in C_b(C([0,1],\mathbf{X}))$ the map

$$(0,\varepsilon]^2 \ni (\varepsilon_0,\varepsilon_1) \mapsto \int \varphi \,\mathrm{d}\boldsymbol{\pi}^{\varepsilon_0,\varepsilon_1}$$

is Borel;

-) for every $\varepsilon_0, \varepsilon_1 \in (0, \varepsilon]$ we have

$$(e_{0})_{*}\boldsymbol{\pi}^{\varepsilon_{0},\varepsilon_{1}} = \frac{1}{\mathfrak{m}(B_{\varepsilon_{0}}(x_{0}))}\mathfrak{m}|_{B_{\varepsilon_{0}}(x_{0})}, \quad (e_{1})_{*}\boldsymbol{\pi}^{\varepsilon_{0},\varepsilon_{1}} = \frac{1}{\mathfrak{m}(B_{\varepsilon_{1}}(x_{1}))}\mathfrak{m}|_{B_{\varepsilon_{1}}(x_{1})};$$

-) we have

$$\limsup_{\varepsilon_0,\varepsilon_1\downarrow 0} \int \int_0^1 |\dot{\gamma}_t|^2 \,\mathrm{d}t \,\mathrm{d}\boldsymbol{\pi}^{\varepsilon_0,\varepsilon_1}(\gamma) \leq \mathsf{d}^2(x_0,x_1).$$

The measured-length property and the a.e. locally doubling property (a variant of Definition 1.5.20 requiring that the property described in Definition 1.5.20 holds for every $x \in B$ with $B \subset X$ with **m**-negligible complement instead that for every $x \in X$) both passes to the warped product space. Notice also that in case w_d and w_m strictly positive (as we have in Chapters 2 and 3) the locally doubling property pass to the warped product space.

Proposition 1.7.8. Consider assumptions and notations as in Assumption 1.7.5, moreover assume that $w_{\mathfrak{m}}$ is strictly positive in the interior of I and that X is a measured-length space. Then the warped product space $I \times_w X$ is a measured-length space.

Proposition 1.7.9. Consider assumptions and notations as in Assumption 1.7.5, and assume that X is a.e. locally doubling. Then the warped product space $I \times_w X$ is a.e. locally doubling.

We conclude this section seeing how the last two definitions are related to the Sobolev-to-Lipschitz property.

Proposition 1.7.10. Let (X, d, \mathfrak{m}) be an a.e. locally doubling measuredlength space. Then it has the Sobolev-to-Lipschitz property.

In particular, the following theorem holds.

Theorem 1.7.11. Consider assumptions and notations as in Assumption 1.7.5, moreover assume that X is an a.e. locally doubling measured-length space and that w_m is strictly positive in the interior of I. Then the warped product space $I \times_w X$ is an a.e. locally doubling measured-length space. In particular, it has the Sobolev-to-Lipschitz property.

1.8 Regular Lagrangian Flows

As we stated in the Introduction, the strategy to prove rigidity theorems is based on taking a "good" function and studying the flow of its gradient. In Section 1.3 we defined the gradient of a function and the space of vector fields thanks to the theory of normed modules. In this section we give the notion of Regular Lagrangian Flow and some useful properties such as existence, uniqueness and regularity. This theory is due to R. J. DiPerna and P.-L. Lions (see [DPL]) in the Euclidean space (see also [A]) and generalized to the metric setting by L. Ambrosio and D. Trevisan in [AT14] (see also [AT15]). We refer to [GR18] for a reformulation of the results in [AT14] using differential calculus in RCD spaces.

As in the smooth setting, there are two approaches to this problem: Lagrangian and Eulerian.

The former studies the trajectories of particles: given v_t time dependent vector field, for $x \in X$ we want to find a curve $\gamma : [0, T] \to X$ that in some sense satisfies

$$\begin{cases} \gamma_t' = v_t(\gamma_t), \\ \gamma_0 = x. \end{cases}$$

The Eulerian approach studies how the mass changes in time in a fixed point: as before, let v_t be a time dependent vector field, then given a starting measure $\bar{\mu}$, the push-forward measure with respect to the flow $\mu_s := (\mathsf{Fl}_s^{(v_t)})_*\bar{\mu}$ must satisfy, in a weak sense, the continuity equation

$$\begin{cases} \frac{d}{dt}\mu_t + \operatorname{div}(v_t\mu_t) = 0, \\ \mu_0 = \bar{\mu}. \end{cases}$$

In this section we recall the definitions of Regular Lagrangian Flow and weak solution of the continuity equation in the non-smooth setting, we see how the two approaches are related and we state an existence and uniqueness theorem for the flow of a regular enough vector field. We conclude recalling a regularity theorem (from [BS20b]), which states that if a vector field has bounded (symmetric) covariant derivative then its Regular Lagrangian Flow admits a Lipschitz representative, and some properties of flows of vector fields that do not depend on time.

In this section we assume that (X, d, \mathfrak{m}) is an $\mathsf{RCD}(K, \infty)$ space for some $K \in \mathbb{R}$.

Definition 1.8.1 (Solution of the continuity equation). Let $\mu : [0,T] \rightarrow \mathscr{P}(X)$ and $v : [0,T] \rightarrow L^0(TX)$ be Borel maps. We say that they solve the continuity equation

$$\frac{\mathrm{d}}{\mathrm{d}t}\mu_t + \mathrm{div}(v_t\mu_t) = 0 \tag{1.8.1}$$

if the following conditions hold:

- i) there exists C > 0 such that $\mu_t \leq C\mathfrak{m}$ for every $t \in [0, T]$;
- *ii) it holds*

$$\int_0^T \int |v_t|^2 \,\mathrm{d}\mu_t \,\mathrm{d}t < +\infty;$$

iii) for every $f \in W^{1,2}(X)$ the map $t \mapsto \int f d\mu_t$ is absolutely continuous and for a.e. $t \in [0, T]$ it holds

$$\frac{\mathrm{d}}{\mathrm{d}t} \int f \,\mathrm{d}\mu_t = \int \mathrm{d}f(v_t) \,\mathrm{d}\mu_t.$$

Existence and uniqueness of solutions of the continuity equation in the sense of Definition 1.8.1 can be established under suitable regularity assumption on the vector field v_t . Among other things, a control on the covariant derivative ∇v_t is required.

Theorem 1.8.2 (Uniqueness of solutions of the continuity equation). Let $(v_t): [0,T] \to L^2(TX)$ be Borel and such that $v_t \in D(\operatorname{div})$ for every $t \in [0,T]$. Moreover, assume that $|||v_t|||_{L^2(\mathfrak{m})} \in L^1(0,T)$, $||\operatorname{div}(v_t)||_{L^2(\mathfrak{m})} \in L^1(0,T)$, $||\operatorname{div}(v_t)^-||_{L^\infty} \in L^\infty(0,T)$, and $||\nabla v_t||_{L^2(T^{\otimes 2}X)} \in L^1(0,T)$. Moreover, let $\bar{\mu} \in \mathscr{P}(X)$ be such that $\bar{\mu} \leq C\mathfrak{m}$ for a constant C > 0.

Then there exists a unique $\mu : [0,T] \to \mathscr{P}(X)$ such that the couple (μ, v) is a solution of the continuity equation (1.8.1) with $\mu_0 = \overline{\mu}$.

Passing now to the Lagrangian approach, we recall the definition of Regular Lagrangian Flow.

Definition 1.8.3 (Regular Lagrangian Flow). Let $(v_t) : [0,T] \to L^2(TX)$ be Borel. We say that a map $FI : [0,T] \times X \to X$ is a Regular Lagrangian Flow associated to (v_t) if the following are satisfied:

i) there exists C > 0 such that

$$(\mathsf{Fl}_t)_*\mathfrak{m} \le C\mathfrak{m} \qquad for \ every \ t \in [0, T];$$
 (1.8.2)

- ii) for \mathfrak{m} -a.e. $x \in X$ the function $[0,T] \ni t \to \mathsf{Fl}_t(x)$ is continuous and satisfies $\mathsf{Fl}_0(x) = x$;
- iii) for every $f \in \text{Test}(X)$ it holds that for \mathfrak{m} -a.e. $x \in X$ the function $(0,T) \ni t \to f \circ \mathsf{Fl}_t(x)$ is absolutely continuous and

$$\frac{\mathrm{d}}{\mathrm{d}t}f \circ \mathsf{Fl}_t(x) = \langle \nabla f, v_t \rangle \circ \mathsf{Fl}_t(x)$$
(1.8.3)

for a.e. $t \in (0, T)$.

Regualar Lagrangian Flows can be characterized via the following proposition. **Proposition 1.8.4.** Let $(v_t) : [0,T] \to L^2(TX)$ be Borel and such that $|v_t| \in L^{\infty}([0,T], L^{\infty}(X))$. Moreover, let $\mathsf{FI} : [0,T] \times X \to X$ be a Borel map satisfying (i) and (ii) of Definition 1.8.3. Then the following are equivalent:

- i) FI is a Regular Lagrangian Flow associated to (v_t) ;
- ii) for every $f \in W^{1,2}(X)$ the map $[0,T] \ni t \mapsto f \circ \mathsf{Fl}_t \in L^2(X)$ is Lipschitz and

$$\lim_{h \to 0} \frac{f \circ \mathsf{Fl}_{t+h} - f \circ \mathsf{Fl}_t}{h} = \mathrm{d}f(v_t) \circ \mathsf{Fl}_t \qquad for \ a.e. \ t \in [0, T], \quad (1.8.4)$$

the limit being intended in $L^2(\mathfrak{m})$;

iii) for every $f \in W^{1,2}_{loc}(\mathbf{X})$ the map $[0,T] \ni t \mapsto f \circ \mathsf{Fl}_t \in L^2_{loc}(\mathbf{X})$ is Lipschitz and for a.e. $t \in [0,T]$ formula (1.8.4) holds with the limit being intended in $L^2_{loc}(\mathfrak{m})$.

As mentioned before, existence and uniqueness of Regular Lagrangian Flows are related to existence and uniqueness of solutions of the continuity equation.

Theorem 1.8.5 (Existence and uniqueness of Regular Lagrangian Flows). Let (v_t) be as in Theorem 1.8.2. Then there exists a unique Regular Lagrangian Flow Fl_t associated to v_t .

Uniqueness is intended as: if both FI and $\tilde{\mathsf{FI}}$ are two such flows, then for \mathfrak{m} -a.e. x we have $\mathsf{FI}_t(x) = \tilde{\mathsf{FI}}_t(x)$ for any $t \in [0, T]$.

Moreover, the following hold:

-) for \mathfrak{m} -a.e. $x \in X$ the curve $t \mapsto \mathsf{Fl}_t(x)$ is absolutely continuous and

$$\mathsf{ms}(\mathsf{FL}(x), t) = |v_t| \circ F_t(x) \qquad a.e. \ t \in [0, 1]; \tag{1.8.5}$$

-) indicating with μ_t the unique solution of the continuity equation given by Theorem 1.8.2, μ_t satisfies

$$\mu_t = (\mathsf{Fl}_t^{(v)})_* \bar{\mu}.$$

We conclude recalling a regularity theorem about Regular Lagrangian Flows. Its proof is due to E. Bruè and D. Semola and can be found in [BS20b].

They introduced the notion of symmetric covariant derivative defined as follows.

Definition 1.8.6 (Symmetric covariant derivative). The space $W_{C,s}^{1,2}(TX)$ is defined as the space of all vector fields $v \in L^2(TX)$ for which there exists a tensor $S \in L^2(T^{\otimes 2}X)$ such that for every $\varphi, \psi_1, \psi_2 \in \text{Test}(X)$ it holds

$$2\int \varphi S(\nabla \psi_1, \nabla \psi_2) \,\mathrm{d}\mathfrak{m} = \int -\langle v, \nabla \psi_1 \rangle \operatorname{div}(\varphi \nabla \psi_2) - \langle v, \nabla \psi_2 \rangle \operatorname{div}(\varphi \nabla \psi_1) \\ + \operatorname{div}(\varphi v) \langle \nabla \psi_1, \nabla \psi_2 \rangle \rangle \,\mathrm{d}\mathfrak{m}.$$

The tensor S is called symmetric covariant derivative of v and we denote it with $\nabla_{sym} v$. We endow the space $W_{C,s}^{1,2}(TX)$ with the norm

$$\|v\|_{W^{1,2}_{C,s}(T\mathbf{X})}^2 := \|v\|_{L^2(T\mathbf{X})}^2 + \|\nabla_{sym}v\|_{L^2(T^{\otimes 2}\mathbf{X})}^2.$$

As noted in [BS20b], $W_C^{1,2}(TX) \subset W_{C,s}^{1,2}(TX)$ and the symmetric covariant derivative of a vector field is the symmetric part of its covariant derivative whenever the latter exists.

Let (v_t) be a time dependent vector field such that for every $t \in [0, T]$ it admits a symmetric covariant derivative. We define

$$L(v) := \sup_{t \in [0,T]} \|\nabla_{sym} v_t\|_{L^{\infty}}.$$

Theorem 1.8.7. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ space. Assume that (v_t) satisfies the hypotheses of Theorem 1.8.5 and assume that $L(v) < +\infty$. Then the Regular Lagrangian Flow FI of (v_t) admits a Lipschitz representative and for every $x, y \in X$ it holds

$$\mathsf{d}(\mathsf{Fl}_t(x),\mathsf{Fl}_t(y)) \le e^{Lt}\mathsf{d}(x,y).$$

Remark 1.8.8. In [BS20b] the authors assumed the space to be compact in order to deduce Lipschitz regularity of the flow. This was needed as they were using the main result in [GT18] that, at that time, provided a necessary second-order differentiation formula on finite-dimensional and compact RCD spaces. A subsequent improvement of this paper [GT21] established the same second-order differentiation formula in the non-compact setting, thus allowing Bruè-Semola's result to be extended to the non-compact setting.

We consider now an autonomous vector field (v_t) , i.e. such that there exists $v \in L^2(TX)$ which satisfies $v_t \equiv v$ for every $t \in [0, T]$.

Let v be as in Theorem 1.8.5. From Theorem 1.8.5 follows easily that Fl can be extended uniquely to a map $[0, +\infty) \times X \to X$ such that for every $t, s \in [0, +\infty)$ it holds the group property

$$\mathsf{Fl}_t \circ \mathsf{Fl}_s = \mathsf{Fl}_{t+s} \qquad \mathfrak{m}\text{-a.e.}, \tag{1.8.6}$$

where, as in Theorem 1.8.5, uniqueness is to be intended up to a negligible set of trajectories.

Assuming that $\operatorname{div}(v) \in L^{\infty}(\mathfrak{m})$ (notice that before we only assumed that its negative part was in $L^{\infty}(\mathfrak{m})$) and denoting with Fl^{-v} the Regular Lagrangian Flow of -v, it holds (see for instance [GR18, Lemma 3.18]) that $\mathsf{Fl}_t^{-v} \circ \mathsf{Fl}_t = \operatorname{Id}$ for every $t \in [0, +\infty)$, then, defining $\mathsf{Fl}_{-t} := \mathsf{Fl}_t^{-v}$ for every $t \in [0, +\infty)$ we can extend the flow Fl on $\mathbb{R} \times X$ and in this case the group property (1.8.6) holds for every $t, s \in \mathbb{R}$.

Moreover the following holds:

in (*ii*) and (*iii*) of Proposition 1.8.4 we can replace "Lipschitz" with " C^{1} " and formula 1.8.4 holds for every t. (1.8.7)

Chapter 2

A general splitting principle on RCD spaces

In this chapter we present the main result of [GMa]: the general splitting principle Theorem 0.0.6.

2.1 Producing a coordinate function

In the practice of studying rigidity properties of spaces with lower Ricci bounds, one often finds out a function u with somehow controlled gradient and Laplacian and for which equality holds in the Bochner inequality, and this in turn gives information about the structure of the Hessian of u. The typical informations of u are the identities

$$\begin{aligned} |\mathrm{d}u| &= \varphi \circ u, \\ \Delta u &= \xi \circ u, \end{aligned}$$
$$\mathrm{Hess}(u) &= \zeta \circ u \, |\mathrm{d}u| \, \mathrm{Id} + \tilde{\zeta} \circ u \, |\mathrm{d}u| \, \mathrm{e}_1 \otimes \mathrm{e}_1, \end{aligned}$$

where $e_1 = \frac{\nabla u}{|\nabla u|}$ on $|\nabla u| > 0$, for suitable functions $\varphi, \zeta, \tilde{\zeta}, \xi : \mathbb{R} \to \mathbb{R}$.

As we shall see, when this occurs the space splits as warped product $\mathbb{R} \times_w X'$ and the ' \mathbb{R} -coordinate' of the isomorphism is the post-composition of u with a suitable function $\eta : \mathbb{R} \to \mathbb{R}$.

To have a better understanding of the warped product we will ultimately obtain, it is convenient to identify right now who is such suitable postcomposition. This is the scope of the following lemma, whose proof only relies in handling chain rules. **Lemma 2.1.1** (Producing a coordinate function). Let $K \in \mathbb{R}$, $N \in [1, +\infty)$ and let (X, d, \mathfrak{m}) be a $\mathsf{RCD}(K, N)$ space. Assume that $u \in H^{2,2}_{loc}(X)$ satisfies

$$|\mathbf{d}u| = \varphi \circ u \qquad \mathbf{\mathfrak{m}}\text{-}a.\,e. \tag{2.1.1}$$

for some $\varphi : u(\mathbf{X}) \to (0, \infty)$ in $C_{loc}^{1,1}$. Put $\mathbf{e}_1 := \frac{\nabla u}{|\nabla u|}$ (this is \mathfrak{m} -a.e. well defined as $|\nabla u| > 0$ \mathfrak{m} -a.e. as a consequence of our assumption on φ) and assume that for some locally Lipschitz functions $\zeta, \tilde{\zeta} : u(\mathbf{X}) \to \mathbb{R}$ we have

$$\operatorname{Hess} u = \zeta \circ u |\mathrm{d}u| \mathrm{Id} + \tilde{\zeta} \circ u |\mathrm{d}u| \mathrm{e}_1 \otimes \mathrm{e}_1.$$
(2.1.2)

Then

$$\varphi' = \zeta + \tilde{\zeta}, \tag{2.1.3}$$

any function $\eta: u(\mathbf{X}) \to (0, \infty)$ in $C_{loc}^{1,1}$ such that $\eta' = \frac{1}{\varphi}$ is invertible and the function $\mathbf{b} := \eta \circ u$ is in $H_{loc}^{2,2}(\mathbf{X})$ with

$$|db| = 1,$$
 (2.1.4a)

$$\operatorname{Hess} \mathbf{b} = \zeta \circ \eta^{-1} \circ \mathbf{b} \big(\operatorname{Id} - \mathbf{e}_1 \otimes \mathbf{e}_1 \big). \tag{2.1.4b}$$

If moreover we have $u \in D(\Delta_{loc})$ with $\Delta u = \xi \circ u$ for some $\xi : u(X) \to \mathbb{R}$ Borel locally bounded, then $b \in D(\Delta_{loc})$ with

$$\Delta \mathbf{b} = \left(\frac{\xi}{\varphi} - \varphi'\right) \circ \eta^{-1} \circ \mathbf{b}.$$
 (2.1.5)

Proof. Since $u \in H^{2,2}_{loc}(X)$ is with $|du| \in L^{\infty}_{loc}(X)$ (by (2.1.1)) we have that $|du| \in W^{1,2}_{loc}(X)$ (recall (1.6.2)), thus we can write

$$\varphi' \circ u \, \mathrm{d}u = \mathrm{d}|\mathrm{d}u| = \mathrm{Hess}\, u(\frac{\mathrm{d}u}{|\mathrm{d}u|}) = (\zeta \circ u + \tilde{\zeta} \circ u) \, \mathrm{d}u$$

and since |du| > 0 m-a.e., property (2.1.3) follows.

The fact that η is invertible follows directly from $\eta' = \frac{1}{\varphi} > 0$ and the fact that $b \in H^{2,2}_{loc}(X)$ from the chain rule noticing that:

- -) if $u \in \text{Test}_{loc}(X)$ and $\varphi \in C^{\infty}_{loc}(u(X))$, then $\varphi \circ u \in \text{Test}_{loc}(X)$ (by direct computation);
- -) formula (1.6.1) shows that if u is also locally Lipschitz, then the Hessian of $\varphi \circ u$ is locally in L^2 .

Then (2.1.4a) follows from $|db| = \eta' \circ u |du| = (\eta' \varphi) \circ u$ and the choice of η . For (2.1.4b) we use the chain rule for the Hessian (1.6.1) to compute

Hess
$$\mathbf{b} = \eta' \circ u$$
 Hess $u + \eta'' \circ u \mathrm{d}u \otimes \mathrm{d}u$
(by (2.1.1),(2.1.2)) $= \zeta \circ u \mathrm{Id} + \tilde{\zeta} \circ u \mathrm{e}_1 \otimes \mathrm{e}_1 - \varphi' \circ u \mathrm{e}_1 \otimes \mathrm{e}_1$
(by (2.1.3)) $= \zeta \circ u (\mathrm{Id} - \mathrm{e}_1 \otimes \mathrm{e}_1),$

which is (2.1.4b). The last claim is also a direct consequence of the assumptions and of the chain rule for the Laplacian (1.3.1).

2.2 Set up and statement of the splitting result

From now on we shall assume the following:

- a) (X, d, \mathfrak{m}) is an $\mathsf{RCD}(K, N)$ space, $K \in \mathbb{R}$, $N < \infty$, with $\operatorname{supp}(\mathfrak{m}) = X$.
- b) $b: X \to \mathbb{R}$ is a function in $H^{2,2}_{loc} \cap D(\Delta_{loc})$ such that

$$|db| = 1, \qquad \mathfrak{m}\text{-a.e.} \tag{2.2.1a}$$

$$\Delta \mathbf{b} = \psi_{\mathfrak{m}} \circ \mathbf{b}, \tag{2.2.1b}$$

Hess
$$\mathbf{b} = \psi_{\mathsf{d}} \circ \mathbf{b} \left(\mathrm{Id} - \mathbf{e}_1 \otimes \mathbf{e}_1 \right)$$
 where $\mathbf{e}_1 := \nabla \mathbf{b} = \frac{\nabla \mathbf{b}}{|\nabla \mathbf{b}|}$, (2.2.1c)

for some $\psi_{\mathfrak{m}}, \psi_{\mathsf{d}} : \mathbb{R} \to \mathbb{R}$ locally Lipschitz. By (2.2.1a) and the Sobolevto-Lipschitz property, b has a 1-Lipschitz representative and we shall always identify it with such representative.

These assumptions have the following rather direct consequences:

- -) The Regular Lagrangian Flow Fl : ℝ × X → X of ∇b is well defined. Indeed, for η ∈ LIP_{bs}(X) the vector field η∇b admits a Regular Lagrangian Flow thanks to the properties (2.2.1). Then taking into account the finite speed of propagation of this flow (from (2.2.1a) and (1.8.5)) it is easy to see that taking η equal 1 on a ball of radius R and then letting R ↑ ∞ and using the fact that div(∇b) = Δb is bounded on the level sets of b we can find the desired flow (Fl_t).
- -) $\mathsf{FI}: \mathbb{R} \times X \to X$ has a continuous representative, still denoted by FI , and its Lipschitz constant on $[-T, T] \times X$ is bounded for every T > 0. Such Lipschitz regularity follows from the fact that the covariant derivative of ∇b is bounded on $b^{-1}([-T, T])$, because of (2.2.1c), and Theorem 1.8.7.
- -) The formula

$$\mathbf{b}(\mathsf{Fl}_t(x)) = \mathbf{b}(x) + t \tag{2.2.2}$$

holds for every $t \in \mathbb{R}$, $x \in X$. Indeed, we know from (1.8.3) that for ma.e. x the function $t \mapsto b(\mathsf{Fl}_t(x))$ is in $W^{1,1}_{loc}(\mathbb{R})$ (in fact locally absolutely continuous, as b is Lipschitz) with derivative equal to $|\mathrm{db}|^2(\mathsf{Fl}_t(x))$. By integration and using (1.8.2) we see from (2.2.1a) that for given $t \in \mathbb{R}$ the formula (2.2.2) holds for m-a.e. x. The claim follows by the continuity of Fl and b. -) For any two $x, y \in X$ with b(x) = b(y) there is a Lipschitz curve joining them lying entirely on the same level set. Indeed, if γ is any Lipschitz path connecting x any y, say a geodesic, then $t \mapsto \mathsf{Fl}_{b(x)-b(\gamma_t)}(\gamma_t)$ is still Lipschitz (by the above Lipschitz regularity) joins x and y and lies on the same level set of x, y (by (2.2.2)).

We can now define the metric measure space (X', d', \mathfrak{m}') by putting $X' := b^{-1}(0)$ and then defining

$$\mathsf{d}'(x,y)^2 := \inf \int_0^1 |\dot{\gamma}_t|^2 \,\mathrm{d}t, \qquad (2.2.3)$$

the inf being taken among all continuous curves $\gamma:[0,1]\to {\rm X}'\subset {\rm X}$ joining x and y and

$$\mathfrak{m}' := c^{-1} \Pr_*(\mathfrak{m}_{|_{\mathbf{b}^{-1}([0,1])}}), \quad \text{for} \quad c := \int_0^1 w_\mathfrak{m} \, \mathrm{d}\mathcal{L}^1, \quad (2.2.4)$$

where $\Pr: X \to X'$ is defined as $\Pr(x) := \mathsf{Fl}_{-\mathsf{b}(x)}(x)$, and $w_{\mathfrak{m}}$ is a positive function that we will define soon (see (2.2.9a) and (2.2.10), for now it is enough that c is a positive constant). It is clear that d' is a (finite) distance on X' and, since clearly from the above \Pr is locally Lipschitz, that it is complete and induces the same topology on X' as the one induced by d. Then obviously \mathfrak{m}' is a Borel measure and from (2.2.6) below it easily follows that it is finite on bounded sets (and thus Radon).

We also define

$$\begin{array}{rcccc} \mathsf{T} : & \mathsf{X} & \to & \mathbb{R} \times \mathsf{X}', \\ & x & \mapsto & (\mathsf{b}(x), \Pr(x)) \end{array} \tag{2.2.5}$$

and notice that

T is locally Lipschitz, invertible, with locally Lipschitz inverse, (2.2.6)

where on $\mathbb{R} \times X'$ we are, for the moment, putting the distance

$$\mathsf{d}_{+}((t,x),(s,y)) := \mathsf{d}'(x,y) + |t-s|.$$

Indeed, the local Lipschitz regularity of T is obvious. For the other inequality notice that

$$\mathsf{d}(x,y) \le \mathsf{d}(\mathsf{Fl}_{\mathsf{b}(x)}(\Pr(x)), \mathsf{Fl}_{\mathsf{b}(x)}(\Pr(y))) + \mathsf{d}(\mathsf{Fl}_{\mathsf{b}(x)-\mathsf{b}(y)}(y), y)$$
(2.2.7)

for every $x, y \in X$, and use that $t \mapsto \mathsf{Fl}_t(y)$ is 1-Lipschitz (by (2.2.1a) and (1.8.5)), that $\mathsf{Fl}_z : X' \to X$ is Lipschitz uniformly on $z \in [-T, T]$ for any

T > 0 (by Theorem 1.8.7 applied to the vector field $\eta \circ b\nabla b$ with η smooth cut-off function with bounded support) and that $d \leq d'$ on X' (obviously from the definition). Observe also that

the inverse of T is the map
$$(t, x') \mapsto \mathsf{Fl}_t(x')$$
 (2.2.8)

as it is easily seen from the definitions recalling also (2.2.2).

With this said, we now want to equip the set $\mathbb{R} \times X'$ with a warped product structure. To this aim, let us introduce the locally absolutely continuous functions $w_{\mathfrak{m}}, w_{\mathsf{d}} : \mathbb{R} \to (0, +\infty)$ defined by

$$(\log(w_{\mathfrak{m}}))' = \psi_{\mathfrak{m}}, \qquad (2.2.9a)$$

$$(\log(w_{\mathsf{d}}))' = \psi_{\mathsf{d}},\tag{2.2.9b}$$

normalized in such a way that

$$w_{\mathfrak{m}}(0) = w_{\mathsf{d}}(0) = 1.$$
 (2.2.10)

Then on $\mathbb{R} \times X'$ we define the (Radon) warped product measure \mathfrak{m}_w by the formula

$$\int f(t,x) \,\mathrm{d}\mathfrak{m}_w := \int \left(\int f(t,x) \,\mathrm{d}\mathfrak{m}'(x) \right) w_{\mathfrak{m}}(t) \,\mathrm{d}\mathcal{L}^1(t) \tag{2.2.11}$$

and the warped product distance d_w as

$$\mathsf{d}_w\big((t,x),(s,y)\big)^2 = \inf \int_0^1 |\dot{\eta}_r|^2 + w_\mathsf{d}^2(\eta_r)|\dot{\gamma}_r|^2 \,\mathrm{d}r, \qquad (2.2.12)$$

the inf being taken among all absolutely continuous curves $\gamma : [0, 1] \to X'$ and $\eta : [0, 1] \to \mathbb{R}$ joining x to y and t to s respectively. Since w_d is continuous and strictly positive, it is clear that d_w is locally equivalent to (i.e. up to multiplicative constants controls and is controlled by) the distance d_+ used above. In particular, it induces the product topology.

We can now state the main result of this chapter (which is, stated slightly differently, Theorem 0.0.6).

Theorem 2.2.1. Under the assumptions (a, b) stated above, the following holds.

The map T defined in (2.2.5) is a measure preserving isometry from X to $\mathbb{R} \times_w X'$, the latter being the space $\mathbb{R} \times X'$ equipped with the distance d_w and the measure \mathfrak{m}_w .

The proof of this result will come as a result of the analysis in the following sections, the conclusion being in Section 2.5.

2.3 Behaviour of the measure under the flow

Let us define the locally Lipschitz map $\mathbb{R}^2 \ni (t, z) \mapsto \Psi_{\mathfrak{m}, t}(z) \in \mathbb{R}$ as

$$\Psi_{\mathfrak{m},t}(z) := \frac{w_{\mathfrak{m}}(z-t)}{w_{\mathfrak{m}}(z)}.$$
(2.3.1)

Then from (2.2.9a) we get

$$\partial_t \Psi_{\mathfrak{m},t}(z) = -\psi_{\mathfrak{m}}(z-t)\Psi_{\mathfrak{m},t}(z), \quad \text{and} \quad \Psi_{\mathfrak{m},0} \equiv 1 \quad (2.3.2)$$

and

$$\partial_t \Psi_{\mathfrak{m}} + \partial_z \Psi_{\mathfrak{m}} + \psi_{\mathfrak{m}} \Psi_{\mathfrak{m}} = 0.$$
(2.3.3)

Lemma 2.3.1. With the same assumptions and notation of Section 2.2 and for $\Psi_{\mathfrak{m}}$ defined as in (2.3.1) we have

$$(\mathsf{Fl}_t)_*\mathfrak{m} = \Psi_{\mathfrak{m},t} \circ \mathfrak{b}\,\mathfrak{m} \qquad \forall t \in \mathbb{R}.$$

$$(2.3.4)$$

Proof. Recalling the simple implication " $T_*\mu = \nu$ implies $T_*(\rho\mu) = \rho \circ T^{-1}\nu$ ", and by the finite speed of propagation of the flow, to conclude it is sufficient to prove that for any Lipschitz probability density ρ with bounded support we have $(\mathsf{Fl}_t)_*(\rho\mathfrak{m}) = \rho_t\mathfrak{m}$, where $\rho_t = (\rho \circ \mathsf{Fl}_{-t})(\Psi_{\mathfrak{m},t} \circ b)$.

By the uniqueness result for the continuity equation (that is central for the theory of Regular Lagrangian Flows, see Section 1.8), this latter claim will follow if we prove that (ρ_t) solves

$$\partial_t \rho_t + \operatorname{div}(\rho_t \nabla \mathbf{b}) = 0$$
 a.e. t.

Notice that $(t, x) \mapsto \rho_t(x)$ is Lipschitz, thus the computations that we are going to perform are justified. Now observe that letting $h \to 0$ in $\frac{\rho \circ \mathsf{Fl}_{-t} \circ \mathsf{Fl}_{-h} - \rho \circ \mathsf{Fl}_{-t}}{h}$ we see that $\partial_t(\rho \circ \mathsf{Fl}_{-t}) = -\langle \nabla(\rho \circ \mathsf{Fl}_{-t}), \nabla b \rangle$, thus

$$\partial_t \rho_t = - \langle \nabla (\rho \circ \mathsf{Fl}_{-t}), \nabla \mathbf{b} \rangle \ \Psi_{\mathfrak{m},t} \circ \mathbf{b} + (\rho \circ \mathsf{Fl}_{-t}) (\partial_t \Psi_{\mathfrak{m},t}) \circ \mathbf{b}$$

On the other hand

$$\begin{split} \operatorname{div}(\rho_t \nabla \mathbf{b}) &= \langle \nabla \rho_t, \nabla \mathbf{b} \rangle + \rho_t \Delta \mathbf{b} \\ &= \langle \nabla(\rho \circ \mathsf{FI}_{-t}), \nabla \mathbf{b} \rangle \, \Psi_{\mathfrak{m},t} \circ \mathbf{b} + (\rho \circ \mathsf{FI}_{-t})(\partial_z \Psi_{\mathfrak{m},t}) \circ \mathbf{b} \, |\mathrm{db}|^2 + \rho_t \Delta \mathbf{b}, \end{split}$$

thus recalling (2.2.1) and adding up we conclude that

$$\partial_t \rho_t + \operatorname{div}(\rho_t \nabla \mathbf{b}) = (\rho \circ \mathsf{FI}_{-t}) \big(\partial_t \Psi_{\mathfrak{m},t} + \partial_z \Psi_{\mathfrak{m},t} + \psi_{\mathfrak{m}} \Psi_{\mathfrak{m},t} \big) \circ \mathbf{b} \stackrel{(2.3.3)}{=} 0,$$

as desired.

For $B \subset X'$ Borel we define $\hat{B} \subset X$ as

$$\hat{B} := \mathsf{Fl}_{\cdot}^{-1}(B) = \bigcup_{t \in \mathbb{R}} \mathsf{Fl}_{t}^{-1}(B) = \{\mathsf{Fl}_{t}(B) : t \in \mathbb{R}\}.$$
(2.3.5)

Clearly \hat{B} is Borel. Then we have the following.

Lemma 2.3.2. With the same assumptions and notation of Section 2.2 (recall in particular the definition (2.2.9a) and the normalization (2.2.10)) the following holds.

Let $B \subset X'$ be Borel and put $\mu := b_*(\mathfrak{m}|_{\hat{B}})$. Then

$$(\operatorname{tr}_t)_*\mu = \Psi_{\mathfrak{m},t}\,\mu, \qquad \text{for every } t \in \mathbb{R},$$

$$(2.3.6)$$

where $\operatorname{tr}_t : \mathbb{R} \to \mathbb{R}$ is the translation map sending z to z + t and moreover

$$\mu = \mathfrak{m}'(B) \, w_{\mathfrak{m}} \, \mathcal{L}^1. \tag{2.3.7}$$

Proof. If $\mathfrak{m}'(B) = \infty$ the conclusion follows from the definition of \mathfrak{m}' and (2.3.4), thus we assume $\mathfrak{m}'(B) < \infty$. By (2.3.4) and the invariance of \hat{B} under the flow it follows immediately that $(\mathsf{Fl}_t)_*(\mathfrak{m}_{|_{\hat{B}}}) = \Psi_{\mathfrak{m},t} \circ \mathfrak{b} \mathfrak{m}_{|_{\hat{B}}}$. Then (2.3.6) follows from

$$(\operatorname{tr}_{t})_{*}\mu = (\operatorname{tr}_{t})_{*} \operatorname{b}_{*}(\mathfrak{m}_{|_{\hat{B}}}) \stackrel{(2.2.2)}{=} \operatorname{b}_{*}(\operatorname{Fl}_{t})_{*}(\mathfrak{m}_{|_{\hat{B}}})$$
$$= \operatorname{b}_{*}(\Psi_{\mathfrak{m},t} \circ \operatorname{b} \mathfrak{m}_{|_{\hat{B}}}) \stackrel{*}{=} \Psi_{\mathfrak{m},t} \operatorname{b}_{*}(\mathfrak{m}_{|_{\hat{B}}}) = \Psi_{\mathfrak{m},t}\mu,$$

where the starred equality is justified by

$$\int \varphi \, \mathrm{db}_*(\Psi_{\mathfrak{m},t} \circ \mathrm{b}\,\mathfrak{m}|_{\hat{B}}) = \int (\varphi \Psi_{\mathfrak{m},t}) \circ \mathrm{b}\,\mathrm{d}\mathfrak{m}|_{\hat{B}} = \int \varphi \Psi_{\mathfrak{m},t} \,\mathrm{db}_*(\mathfrak{m}|_{\hat{B}}).$$

Averaging (2.3.6) in t we see that the left hand side becomes absolutely continuous with respect to \mathcal{L}^1 , thus showing that $\mu \ll \mathcal{L}^1$, say $\mu = \rho \mathcal{L}^1$. Then (2.3.6) becomes: for any $t \in \mathbb{R}$ we have $\rho(z - t) = \Psi_{\mathfrak{m},t}(z)\rho(z)$ for \mathcal{L}^1 -a.e. z. From this identity and $\partial_t \Psi_{\mathfrak{m},t}|_{t=0} = -\psi_{\mathfrak{m}}$ (that comes from (2.3.2)) it easily follows that the distributional derivative ρ' of ρ satisfies $\rho' = \psi_{\mathfrak{m}}\rho$, thus ensuring that ρ is a multiple of $w_{\mathfrak{m}}$, say $\rho = cw_{\mathfrak{m}}$.

To find the value of c notice that

$$\mathbf{b}_*(\mathfrak{m}_{|_{\hat{B}}})([0,1]) = \mathfrak{m}(\hat{B} \cap \mathbf{b}^{-1}[0,1])) = \mathfrak{m}'(B) \int_0^1 w_{\mathfrak{m}} \, \mathrm{d}\mathcal{L}^1$$

by definition of \mathfrak{m}' , and also that

$$\mathbf{b}_*(\mathfrak{m}_{|_{\hat{B}}})([0,1]) = \mu([0,1]) = \int_0^1 \rho \, \mathrm{d}\mathcal{L}^1 = c \int_0^1 w_\mathfrak{m} \, \mathrm{d}\mathcal{L}^1.$$

Let now $B \subset X'$ be Borel as before and with $\mathfrak{m}'(B) < \infty$. Define B as in (2.3.5) and notice that since $\mu := b_*(\mathfrak{m}_{|_{\hat{B}}})$ is σ -finite (in fact Radon), we can disintegrate $\mathfrak{m}_{|_{\hat{B}}}$ along the map b, as the standard proof (see e.g. [B, Section 10.6]) naturally carries over. Thus we obtain a weakly measurable family $z \mapsto \nu_z$ of probability measures on X such that ν_z is concentrated on $b^{-1}(z)$ for μ -a.e. z (equivalently by Lemma 2.3.2 above: for \mathcal{L}^1 -a.e. z) and so that

$$\int \varphi \,\mathrm{d}\mathfrak{m}_{|_{\hat{B}}} = \iint \varphi \,\mathrm{d}\nu_z \,\mathrm{d}\mu(z) \stackrel{(2.3.7)}{=} \mathfrak{m}'(B) \int \Big(\int \varphi \,\mathrm{d}\nu_z\Big) w_{\mathfrak{m}}(z) \,\mathrm{d}\mathcal{L}^1(z)$$
(2.3.8)

holds for every $\varphi : \mathbf{X} \to \mathbb{R}^+$ Borel. Then we have:

Lemma 2.3.3. With the same assumptions and notation of Section 2.2 the following holds.

Let $B \subset X'$ be Borel with $\mathfrak{m}'(B) < \infty$ and \hat{B} as in (2.3.5). Then the disintegration $\{\nu_z\}_{z\in\mathbb{R}}$ of $\mathfrak{m}_{|_{\hat{B}}}$ with respect to b satisfies: for any $t \in \mathbb{R}$ it holds

$$(\mathsf{Fl}_t)_*\nu_z = \nu_{z+t} \qquad \mu\text{-}a.e. \ z.$$
 (2.3.9)

Proof. Fix $t \in \mathbb{R}$. We shall prove that the family $\{(\mathsf{Fl}_t)_*\nu_{z-t}\}$ is an admissible disintegration of $\mathfrak{m}_{|_{\hat{B}}}$ with respect to b: this is (equivalent to) the claim. For μ -a.e. z we know that ν_z is concentrated on $b^{-1}(z)$, thus ν_{z-t} is concentrated on $b^{-1}(z-t)$ and therefore, by (2.2.2), $(\mathsf{Fl}_t)_*\nu_{z-t}$ is concentrated on $b^{-1}(z)$. To conclude, with a density argument based also on (2.2.6) it is therefore sufficient to prove that for any Borel functions $g: X' \to \mathbb{R}^+$ and $h: \mathbb{R} \to \mathbb{R}^+$ we have

$$\int h \circ \mathbf{b} \, g \circ \Pr \, \mathrm{d}\mathfrak{m}_{|_{\hat{B}}} = \int \int h \circ \mathbf{b} \, g \circ \Pr \, \mathrm{d}(\mathsf{Fl}_t)_* \nu_{z-t} \, \mathrm{d}\mu(z).$$
(2.3.10)

We start claiming that it holds

$$\int h(z)\Psi_{\mathfrak{m},t}(z) \left(\int g \circ \operatorname{Pr} d\nu_{z-t}\right) d\mu(z) = \int h(z)\Psi_{\mathfrak{m},t}(z) \left(\int g \circ \operatorname{Pr} d\nu_z\right) d\mu(z).$$
(2.3.11)

Since $Pr \circ \mathsf{Fl}_t = Pr$, we have on one hand

$$\int_{\hat{B}} h \circ b g \circ \Pr d(\mathsf{Fl}_t)_* \mathfrak{m} = \int_{\hat{B}} (\Psi_{\mathfrak{m},t}h) \circ b g \circ \Pr d\mathfrak{m}$$
$$= \int \Psi_{\mathfrak{m},t}(z)h(z) \Big(\int g \circ \Pr d\nu_z\Big) d\mu(z)$$

and on the other

$$\begin{split} \int_{\hat{B}} h \circ b \, g \circ \Pr d(\mathsf{Fl}_t)_* \mathfrak{m} &= \int_{\hat{B}} h \circ b \circ \mathsf{Fl}_t \, g \circ \Pr d\mathfrak{m} \\ (by \ (2.2.2)) &= \int h(z+t) \Big(\int g \circ \Pr d\nu_z \Big) \, d\mu(z) \\ &= \int h(z) \Big(\int g \circ \Pr d\nu_{z-t} \Big) \, d(\operatorname{tr}_t)_* \mu(z) \\ (by \ (2.3.6)) &= \int \Psi_{\mathfrak{m},t}(z) h(z) \Big(\int g \circ \Pr d\nu_{z-t} \Big) \, d\mu(z), \end{split}$$

thus proving our claim (2.3.11). Now notice that replacing h with $h\Psi_{\mathfrak{m},t}$ (recall that $\Psi_{\mathfrak{m},t} > 0$ everywhere) and using again that $\Pr \circ \mathsf{Fl}_t = \Pr$, from (2.3.11) we get

$$\int h(z) \left(\int g \circ \Pr d((\mathsf{Fl}_t)_* \nu_{z-t}) \right) d\mu(z) = \int h(z) \left(\int g \circ \Pr d\nu_z \right) d\mu(z).$$

To conclude we use again the fact that $(\mathsf{Fl}_t)_*\nu_{z-t}$ and ν_z are both concentrated on $b^{-1}(z)$ to deduce from the above that

$$\iint h \circ \mathbf{b} \, g \circ \Pr \, \mathrm{d}((\mathsf{Fl}_t)_* \nu_{z-t}) \, \mathrm{d}\mu(z) = \iint h \circ \mathbf{b} \, g \circ \Pr \, \mathrm{d}\nu_z \, \mathrm{d}\mu(z)$$
$$= \int h \circ \mathbf{b} \, g \circ \Pr \, \mathrm{d}\mathfrak{m}_{|_{\hat{B}}},$$

that is the desired (2.3.10).

Proposition 2.3.4. With the same assumptions and notation of Section 2.2 the following holds.

The map $T: X \to X' \times_w \mathbb{R}$ (recall (2.2.5)) is measure preserving. In other words and recalling (2.2.8), for any $\varphi: X \to \mathbb{R}^+$ Borel we have

$$\int \varphi \,\mathrm{d}\mathfrak{m} = \int \Big(\int \varphi \,\mathrm{d}(\mathsf{Fl}_t)_*\mathfrak{m}'\Big) w_{\mathfrak{m}}(t) \,\mathrm{d}t$$

Proof. Letting $B \subset X'$ be arbitrary Borel with $\mathfrak{m}'(B) < \infty$ and letting φ be 0 outside \hat{B} , we see that to conclude it suffices to prove that we can choose $\nu_z := \mathfrak{m}'(B)^{-1}(\mathsf{Fl}_z)_*(\mathfrak{m}'|_B)$ in the disintegration formula (2.3.8).

To see this, observe that from Lemma 2.3.3 above and Fubini's theorem we see that for a.e. z we have $(\mathsf{Fl}_t)_*\nu_z = \nu_{z+t}$ for a.e. t. Fix \bar{z} for which this holds and for which $\nu_{\bar{z}}$ is concentrated on $b^{-1}(\bar{z})$ and then define $\bar{\nu}_z := (\mathsf{Fl}_{z-\bar{z}})_*\nu_{\bar{z}}$ for any z. Then $\bar{\nu}_z = \nu_z$ for a.e. z, and thus the $\bar{\nu}_z$'s are admissible in formula (2.3.8).

To conclude it is therefore enough to show that $\bar{\nu}_0 = \mathfrak{m}'(B)^{-1}\mathfrak{m}'|_B$. To see this, let $C \subset B$ be Borel, define \hat{C} as in (2.3.5) and recall the definition (2.2.4) of \mathfrak{m}' to get

$$\mathfrak{m}'(C) = \left(\int_0^1 w_{\mathfrak{m}} \,\mathrm{d}\mathcal{L}^1\right)^{-1} \mathfrak{m}(\hat{C} \cap \mathrm{b}^{-1}([0,1]))$$
$$\stackrel{(2.3.8)}{=} \left(\int_0^1 w_{\mathfrak{m}} \,\mathrm{d}\mathcal{L}^1\right)^{-1} \mathfrak{m}'(B) \int_0^1 \bar{\nu}_t(\hat{C}) w_{\mathfrak{m}}(t) \,\mathrm{d}\mathcal{L}^1(t) = \mathfrak{m}'(B) \bar{\nu}_0(C),$$

having also used the definition of $\bar{\nu}_t$ in the last equality. The conclusion follows by the arbitrariness of $C \subset B$.

2.4 Behaviour of the distance under the flow

We shall work under the same assumptions and notation as in Section 2.2.

We start decomposing the tangent module $L^0(TX)$ into the submodules V^{\parallel}, V^{\perp} of vector fields that are pointwise parallel/orthogonal to ∇b . Thus we put

$$V^{\parallel} := \{ v \in L^0(T\mathbf{X}) : v = f \nabla \mathbf{b} \text{ for some } f : \mathbf{X} \to \mathbb{R} \},\$$
$$V^{\perp} := \{ v \in L^0(T\mathbf{X}) : \langle v, \nabla \mathbf{b} \rangle \equiv 0 \}.$$

Given an arbitrary vector field $v \in L^0(TX)$ it components v^{\parallel}, v^{\perp} in V^{\parallel}, V^{\perp} respectively are defined as (recall that $|db| \equiv 1$):

$$v^{\parallel}:=\langle v,
abla \mathrm{b}
angle
abla \mathrm{b} \qquad ext{and} \qquad v^{\perp}:=v-v^{\parallel}.$$

To better understand the structure of V^{\parallel}, V^{\perp} , it is convenient to introduce the following two classes of functions:

$$\mathcal{G} := \{ g \circ \Pr : g \in L^{\infty} \cap W^{1,2}(\mathbf{X}') \text{ with bounded support} \}, \\ \mathcal{H} := \{ h \circ \mathbf{b} : h \in L^{\infty} \cap W^{1,2}(\mathbb{R}) \text{ with bounded support} \}.$$

Notice that both \mathcal{G} and \mathcal{H} are algebra of functions. We shall typically use letters g, h for functions on X', \mathbb{R} respectively and \hat{g}, \hat{h} for $g \circ \Pr, h \circ b$ respectively. It is clear that $\nabla \hat{g} \in V^{\perp}$ and $\nabla \hat{h} \in V^{\parallel}$ and thus that

 $V^{\perp} \supset$ sub-module of the tangent module generated by $\nabla \hat{g}$ with $\hat{g} \in \mathcal{G}$,

 $V^{\parallel} \supset$ sub-module of the tangent module generated by $\nabla \hat{h}$ with $\hat{h} \in \mathcal{H}$. (2.4.1) We shall prove in a moment that these inclusion are actually identities. To see this, it is convenient to introduce the algebra \mathcal{A} of functions on X as

$$\mathcal{A} := \{ \text{algebra of functions on X of the form } \sum_{i=1}^{n} \hat{g}_i \hat{h}_i, \text{ with } n \in \mathbb{N} \}.$$
(2.4.2)

We then have the following result:

Lemma 2.4.1. With the same assumptions and notation of Section 2.2 and with the definitions just given, the following holds.

The algebra \mathcal{A} is densely contained in $W^{1,2}(\mathbf{X})$. Similarly, the algebra of functions $f: \mathbf{X}' \times \mathbb{R} \to \mathbb{R}$ such that $f \circ \mathsf{T} \in \mathcal{A}$ is densely contained in $W^{1,2}(\mathbf{X}' \times_w \mathbb{R})$.

Proof. The second claim follows directly from [GH, Section 3]. Now recall that, directly from the definition of Sobolev spaces, if $T : (X_1, \mathsf{d}_1, \mathfrak{m}_1) \rightarrow (X_2, \mathsf{d}_2, \mathfrak{m}_2)$ is measure preserving and biLipschitz, then

$$L^{-1} \|f\|_{W^{1,2}(\mathbf{X}_2)} \le \|f \circ T\|_{W^{1,2}(\mathbf{X}_1)} \le L \|f\|_{W^{1,2}(\mathbf{X}_2)},$$

where L is the biLipschitz constant. Thus the conclusion would follow if we knew that T was globally biLipschitz, as the measure preserving property comes from Proposition 2.3.4.

In general, it might be that T is only locally biLipschitz, but this is sufficient to conclude, as with a cut-off argument we see that functions with bounded support are dense in $W^{1,2}$ and we can approximate functions in $W^{1,2}(\mathsf{X} \times_w \mathbb{R})$ with bounded support with functions as in the statement with uniformly bounded support.

From this density result it follows that gradients of functions in \mathcal{A} generate the whole tangent module. Since for $f = \sum_i \hat{g}_i \hat{h}_i$ we have $\nabla f = \sum_i \hat{h}_i \nabla \hat{g}_i + \hat{g}_i \nabla \hat{h}_i$, it follows that the sub-module generated by gradients of functions in \mathcal{G} and \mathcal{H} is the whole $L^0(TX)$. Hence (2.4.1) improves into

 V^{\perp} = sub-module of the tangent module generated by $\nabla \hat{g}$ with $\hat{g} \in \mathcal{G}$,

$$V^{\parallel}$$
 = sub-module of the tangent module generated by $\nabla \hat{h}$ with $\hat{h} \in \mathcal{H}$.
(2.4.3)

In particular, we obtain:

$$v \in L^0(TX)$$
 with $\langle v, \nabla \hat{g} \rangle = 0$ for every $\hat{g} \in \mathcal{G} \Rightarrow v \in V^{\parallel}$

$$(2.4.4)$$

Let us now pick $\varphi \in C^2(\mathbb{R})$ with globally bounded second derivative, put $\tilde{b} := \varphi \circ b$ and notice that the Regular Lagrangian Flow $(\tilde{\mathsf{Fl}}_t)$ of $\nabla \tilde{b}$ is globally well defined (the uniform bound on φ'' grants, among other things, that $|\nabla \tilde{b}|$ grows at most linearly, that in turn shows that $\tilde{\mathsf{Fl}}_t(x)$ does not go to infinity in finite time). We claim that for some 'reparametrization map' rep : $\mathbb{R}^2 \to \mathbb{R}$ locally Lipschitz we have

$$\mathsf{Fl}_t(x) = \mathsf{Fl}_{\operatorname{rep}(t, \mathsf{b}(x))}(x) \quad \text{for every } t \in \mathbb{R}, \ x \in \mathcal{X}.$$
 (2.4.5)

Since $T: X \to X' \times \mathbb{R}$ is locally biLipschitz, this will follow if we show that

$$\operatorname{Pr} \circ \mathsf{Fl}_t = \operatorname{Pr}, \quad \text{on X}, \qquad \text{for every } t \in \mathbb{R}, \qquad (2.4.6a)$$

$$\mathbf{b}(\mathbf{FI}_t(x)) = f_t(\mathbf{b}(x)), \quad \text{for every } t \in \mathbb{R}, \ x \in \mathbf{X}, \quad (2.4.6b)$$

for some locally Lipschitz function $f : \mathbb{R}^2 \to \mathbb{R}$.

As for the flow (Fl_t) of ∇b , we have that for any $t \in \mathbb{R}$ the map $\mathsf{Fl}_t : X \to X$ is Lipschitz, with a control on the Lipschitz constant locally uniformly bounded in t. For $\hat{g} \in \mathcal{G}$ we can compute $\partial_t(g \circ \tilde{\mathsf{Fl}}_t) = \langle \nabla \hat{g}, \nabla \tilde{b} \rangle \circ \tilde{\mathsf{Fl}}_t = \varphi' \circ b \langle \nabla \hat{g}, \nabla b \rangle \circ \tilde{\mathsf{Fl}}_t = 0$, i.e. $\hat{g} \circ \tilde{\mathsf{Fl}}_t = \hat{g}$ m-a.e.. Then the arbitrariness of \hat{g} and a continuity argument give (2.4.6a).

To prove (2.4.6b) we start noticing that with the same continuity arguments used to deduce (2.2.2), we see that for any $x \in X$ the curve $t \mapsto b(\tilde{\mathsf{Fl}}_t(x))$ is C^1 and satisfies

$$\partial_t \mathbf{b}(\tilde{\mathsf{FI}}_t(x)) = \varphi'(\mathbf{b}(\tilde{\mathsf{FI}}_t(x))) \quad \text{for every } t \in \mathbb{R}.$$
 (2.4.7)

In other words, the function $g_t(x) := b(\tilde{\mathsf{Fl}}_t(x))$ solves the Cauchy problem $\partial_t g_t(x) = \varphi'(g_t(x))$ with $g_0(x) = b(x)$, hence its value at time t depends only on t and b(x), as in (2.4.6b). The local Lipschitz regularity of f then follows by standard ODE estimates (recall that φ is C^2).

Also, $\tilde{\mathsf{Fl}}_t : X \to X$ has bounded compression. Since this map is invertible with inverse $\tilde{\mathsf{Fl}}_{-t}$ that is also of bounded compression, it admits a differential $d\tilde{\mathsf{Fl}}_t : L^0(TX) \to L^0(TX)$ (see Theorem 1.4.6) characterized by

$$\langle \mathrm{d}\tilde{\mathsf{F}}\mathsf{l}_t(v), \mathrm{d}f \rangle \circ \tilde{\mathsf{F}}\mathsf{l}_t = \langle v, \mathrm{d}(f \circ \tilde{\mathsf{F}}\mathsf{l}_t) \rangle \qquad \text{for every } v \in L^0(T\mathrm{X}), \ f \in W^{1,2}(\mathrm{X})$$
(2.4.8)

Arguing verbatim as for [DPG, Proposition 3.31] we obtain the following result, whose proof we omit (see also [GV, Appendix A]).

Lemma 2.4.2. With the same assumptions and notation of Section 2.2 and with \tilde{b} , (\tilde{Fl}_t) as just defined, the following holds.

Let $v \in L^2(TX)$ and put $v_s := d\tilde{\mathsf{Fl}}_s(v)$. Then the map $\mathbb{R} \ni s \mapsto \frac{1}{2}|v_s|^2 \circ \tilde{\mathsf{Fl}}_s \in L^1(X)$ is C^1 and its derivative, intended as limit of the difference quotients both strongly in L^1 and pointwise a.e., is given by

$$\partial_s(\frac{1}{2}|v_s|^2 \circ \tilde{\mathsf{FI}}_s) = \operatorname{Hess} \tilde{\mathrm{b}}(v_s, v_s) \circ \tilde{\mathsf{FI}}_s \qquad for \ every \ s \in \mathbb{R}.$$
(2.4.9)

The special structure of the Hessian of \tilde{b} allows for a more explicit computation of the above. Notice that the chain rule (1.6.1) grants that $\tilde{b} \in H^{2,2}_{loc}(X)$ with

$$\operatorname{Hess} \tilde{\mathbf{b}} = (\psi_{\mathsf{d}} \varphi') \circ \mathbf{b} \left(\operatorname{Id} - \mathbf{e}_1 \otimes \mathbf{e}_1 \right) + \varphi'' \circ \mathbf{b} \, \mathbf{e}_1 \otimes \mathbf{e}_1 \tag{2.4.10}$$

Also, let us notice that

 $v \in V^{\perp} \Rightarrow \mathrm{d}\tilde{\mathsf{Fl}}_t(v) \in V^{\perp}$ for every $t \in \mathbb{R}$, (2.4.11a)

$$v \in V^{\parallel} \Rightarrow d\tilde{\mathsf{Fl}}_t(v) \in V^{\parallel}$$
 for every $t \in \mathbb{R}$. (2.4.11b)

Indeed, for the first notice that

$$\langle \mathrm{d}\tilde{\mathsf{FI}}_t(v), \nabla \mathrm{b}\rangle \circ \tilde{\mathsf{FI}}_t \stackrel{(2.4.8)}{=} \langle v, \nabla(b \circ \tilde{\mathsf{FI}}_t) \rangle \stackrel{(2.4.6\mathrm{b})}{=} (\partial_x f_t) \circ \mathrm{b} \langle v, \nabla \mathrm{b} \rangle = 0$$

while for the second we pick $\hat{g} \in \mathcal{G}$ and compute

$$\langle \mathrm{d}\tilde{\mathsf{FI}}_t(v), \nabla \hat{g} \rangle \circ \tilde{\mathsf{FI}}_t \stackrel{(2.4.8)}{=} \langle v, \nabla (\hat{g} \circ \tilde{\mathsf{FI}}_t) \rangle \stackrel{(2.4.6a)}{=} \langle v, \nabla \hat{g} \rangle = 0,$$

thus the arbitrariness of \hat{g} and (2.4.4) give the claim.

We are now ready to state and prove the main result of this section.

Proposition 2.4.3. With the same assumptions and notation of Section 2.2 the following holds.

Let $\boldsymbol{\pi}$ be a test plan on X. Then for $\boldsymbol{\pi}$ -a.e. γ the curve $\Pr(\gamma)$ defined as $t \mapsto \Pr(\gamma_t)$ is absolutely continuous and

$$\mathsf{ms}(\mathrm{Pr}(\gamma), t) \le \frac{1}{w_{\mathsf{d}}(\mathrm{b}(\gamma_t))} \mathsf{ms}(\gamma, t) \qquad for \ a. e. \ t, \tag{2.4.12}$$

where we are writing $ms(\gamma, t)$ for the metric speed of the curve γ at the time t.

Moreover, equality holds in (2.4.12) provided $t \mapsto b(\gamma_t)$ is constant for π -a.e. γ i.e. if π is concentrated on curves lying on level sets of b.

Proof. We follow the line of thought used to prove [DPG, Proposition 3.23], this time paying attention to the action of the flow on the parallel and perpendicular directions.

Let $\varphi(z) := -\frac{1}{2}z^2$, $v \in L^2(T\mathbf{X})$ and let $v^{\parallel}, v^{\perp} \in L^2(T\mathbf{X})$ be its components in V^{\parallel}, V^{\perp} , respectively. Put $v_s := d\tilde{\mathsf{Fl}}_s(v), v_s^{\parallel} := d\tilde{\mathsf{Fl}}_s(v^{\parallel})$ and $v_s^{\perp} := d\tilde{\mathsf{Fl}}_s(v^{\perp})$. Then by (2.4.9) and (2.4.10) it follows that

$$|v_s^{\parallel}| \circ \tilde{\mathsf{FI}}_s = |v^{\parallel}| e^{-s}, \tag{2.4.13a}$$

$$|v_s^{\perp}| \circ \tilde{\mathsf{FI}}_s = |v^{\perp}| \exp\Big(\int_0^s -(\mathbf{b}\,\psi_{\mathsf{d}}\circ\mathbf{b})\circ\tilde{\mathsf{FI}}_r\,\mathrm{d}r\Big). \tag{2.4.13b}$$

Formula (2.4.7) gives $\mathbf{b} \circ \tilde{\mathsf{Fl}}_r = e^{-r}\mathbf{b}$, thus we have

$$\begin{split} \int_0^s -(\mathbf{b}\psi_{\mathsf{d}} \circ \mathbf{b}) \circ \tilde{\mathsf{Fl}}_r \, \mathrm{d}r &= \int_0^s -\mathbf{b}e^{-r}\psi_{\mathsf{d}}(\mathbf{b}e^{-r}) \, \mathrm{d}r = \int_{\mathbf{b}e^{-s}}^{\mathbf{b}} -\psi_{\mathsf{d}}(z) \, \mathrm{d}z \\ &= \log\left(\frac{w_{\mathsf{d}}(\mathbf{b}e^{-s})}{w_{\mathsf{d}}(\mathbf{b})}\right) \end{split}$$

and therefore

$$|v_s| \circ \tilde{\mathsf{FI}}_s = \sqrt{|v_s^{\parallel}|^2 \circ \tilde{\mathsf{FI}}_s + |v_s^{\perp}|^2 \circ \tilde{\mathsf{FI}}_s} \le |v| \left(e^{-s} + \frac{w_{\mathsf{d}}(\mathrm{b}e^{-s})}{w_{\mathsf{d}}(\mathrm{b})}\right)$$
(2.4.14)

Now the inequality (2.4.12) follows along the same lines used in [DPG, Proposition 3.33]:

- -) the test plan admits a derivative π'_t as an element of a suitable pullback of the tangent module satisfying $|\pi'_t|(\gamma) = \mathsf{ms}(\gamma, t)$ for π -a.e. γ and a.e. t (see Theorem 1.4.3);
- -) the classical formula $\partial_t(\tilde{\mathsf{FI}}_s(\gamma_t)) = d\tilde{\mathsf{FI}}_{s,\gamma_t}(\gamma'_t)$ admits a natural analogue in this setting if one works with derivatives of test plans in place of derivative of single curves (see Proposition 1.4.8);
- -) coupling these two informations with (2.4.14) we see that for π -a.e. γ we have

$$\mathsf{ms}(\tilde{\mathsf{FI}}_s(\gamma),t) \leq \mathsf{ms}(\gamma,t)(e^{-s} + \tfrac{w_{\mathsf{d}}(\mathrm{b}e^{-s})}{w_{\mathsf{d}}(\mathrm{b})}) \qquad \text{for a.e. } t;$$

-) the conclusion (2.4.12) follows letting $s \to \infty$ in the above, noticing that $\tilde{\mathsf{Fl}}_s \to \mathrm{Pr}$ pointwise, recalling the normalization choice (2.2.10) and using the lower semicontinuity of the metric speed.

For the equality case we argue along the following lines:

- -) If the test plan π is concentrated on curves lying on level sets of b, then its speed is orthogonal to ∇b (as in the proof of Proposition 3.4.6);
- -) formula (2.4.13b) and the link between derivatives of test plans and metric speed recalled before ensure that for π -a.e. γ we have

$$\mathsf{ms}(\tilde{\mathsf{FI}}_{s}(\gamma), t) = \mathsf{ms}(\gamma, t) \frac{w_{\mathsf{d}}(\mathrm{b}e^{-s})}{w_{\mathsf{d}}(\mathrm{b})} \quad \text{for a.e. } t; \qquad (2.4.15)$$

-) recall that from Theorem 1.8.7 if $L := \||\operatorname{Hess} b|_{\mathsf{HS}}\|_{L^{\infty}} < \infty$, then we have the bi-Lipschitz estimate

$$e^{-L|s|}\mathsf{d}(x,y) \le \mathsf{d}(\mathsf{Fl}_s(x),\mathsf{Fl}_s(y)) \le e^{L|s|}\mathsf{d}(x,y)$$
(2.4.16)

valid for any $x, y \in X$ and $s \in \mathbb{R}$, for the flow of ∇b ; in our setting we don't know if Hess b is bounded, but with a cut-off argument based on the fact that |Hess b|_{HS} is bounded by a function of b, it is not hard to see that (2.4.16) still holds for $s \in [-1, 1]$ and $x, y \in b^{-1}([-1, 1])$ for some constant L;

- -) we can assume that π is concentrated on curves lying on a bounded set. Thus for s sufficiently big (F̃l_s)_{*}π is concentrated on curves lying in b⁻¹([-1, 1]);
- -) we use the identity $\Pr(x) = \mathsf{Fl}_{-e^{-s}\mathsf{b}(x)}(\tilde{\mathsf{Fl}}_s(x))$ and the previous item to deduce that for all s sufficiently big the estimate

$$\mathsf{ms}(\mathrm{Pr}(\gamma), t) \stackrel{(2.4.16)}{\geq} e^{-Le^{-s\mathsf{b}(\gamma_t)}} \mathsf{ms}(\tilde{\mathsf{FI}}_s(\gamma), t)$$
$$\stackrel{(2.4.15)}{=} e^{-Le^{-s\mathsf{b}(\gamma_t)}} \mathsf{ms}(\gamma, t) \frac{w_{\mathsf{d}}(\mathsf{b}e^{-s})}{w_{\mathsf{d}}(\mathsf{b})} \quad \text{for a.e. } t$$

holds for π -a.e. γ . Letting $s \uparrow \infty$ and recalling again the normalization (2.2.10) we get the equality in (2.4.12).

Corollary 2.4.4. With the same assumptions and notation of Section 2.2 the following holds.

Let $g \in L^2_{loc}(X')$ be Borel and put, as before, $\hat{g} := g \circ \Pr$. Then $g \in W^{1,2}_{loc}(X')$ if and only if $\hat{g} \in W^{1,2}_{loc}(X)$ and in this case

$$|\mathrm{d}\hat{g}| = \frac{1}{w_{\mathrm{d}}\circ\mathrm{b}}|\mathrm{d}g| \circ \Pr \qquad \mathfrak{m}\text{-}a.e.. \qquad (2.4.17)$$

Proof. It is clear from Proposition 2.3.4 that $g \in L^2_{loc}(X')$ if and only if $\hat{g} \in L^2_{loc}(X)$. Now assume that $g \in W^{1,2}_{loc}(X')$ and let π be a test plan on X concentrated on curves lying on some bounded set. Then, since Pr is locally Lipschitz and by Proposition 2.3.4, we have that $\Pr_*\pi$ is a test plan on X', where with a little abuse of notation we are denoting by Pr the map sending the curve γ to the curve $t \mapsto \Pr(\gamma_t)$. Since g is Sobolev we have

$$\begin{split} \int |\hat{g}(\gamma_1) - \hat{g}(\gamma_0)| \,\mathrm{d}\boldsymbol{\pi}(\gamma) &= \int |g(\eta_1) - g(\eta_0)| \,\mathrm{d}\mathrm{Pr}_*\boldsymbol{\pi}(\eta) \\ &\leq \iint_0^1 |\mathrm{d}g|(\eta_t)|\dot{\eta}_t| \,\mathrm{d}t \,\mathrm{d}\mathrm{Pr}_*\boldsymbol{\pi}(\eta) \\ (``\eta &= \Pr(\gamma)'' + (2.4.12)) \qquad \leq \iint_0^1 \frac{1}{w_{\mathsf{d}}(\mathsf{b}(\gamma_t))} |\mathrm{d}g|(\Pr(\gamma_t))|\dot{\gamma}_t| \,\mathrm{d}t \,\mathrm{d}\boldsymbol{\pi}(\gamma), \end{split}$$

thus proving, by the arbitrariness of $\boldsymbol{\pi}$, that $\hat{g} \in W_{loc}^{1,2}(\mathbf{X})$ and that inequality \leq holds in (2.4.17).

Now assume that $\hat{g} \in W^{1,2}_{loc}(\mathbf{X})$ and let $\boldsymbol{\pi}'$ be a test plan on X' concentrated on curves lying on some bounded set. Fix T > 0, and consider the push forward $\boldsymbol{\pi}$ of the plan $\boldsymbol{\pi}' \times (\frac{1}{2T} \mathcal{L}^1|_{[-T,T]})$ via the map $(x,t) \mapsto \mathsf{Fl}_t(x)$. Then from Proposition 2.3.4, identity (2.2.2) and the fact that $\mathsf{Fl} : \mathsf{X}' \times \mathbb{R} \to \mathsf{X}$ is locally Lipschitz we see that $\boldsymbol{\pi}$ is a test plan on X concentrated on curves lying on level sets of b, thus since \hat{g} is Sobolev we have

$$\int |g(\eta_1) - g(\eta_0)| \,\mathrm{d}\boldsymbol{\pi}'(\eta) = \int |\hat{g}(\gamma_1) - \hat{g}(\gamma_0)| \,\mathrm{d}\boldsymbol{\pi}(\gamma)$$

$$\leq \iint_0^1 |\mathrm{d}\hat{g}|(\gamma_t)|\dot{\gamma}_t| \,\mathrm{d}t \,\mathrm{d}\boldsymbol{\pi}(\gamma)$$

$$((2.4.12) + \mathrm{def.} \ \boldsymbol{\pi}) \qquad = \iint_0^1 \left(\frac{1}{2T} \int_{-T}^T w_{\mathsf{d}}(s) |\mathrm{d}\hat{g}|(\mathsf{Fl}_s(\eta_t)) \,\mathrm{d}s\right) |\dot{\eta}_t| \,\mathrm{d}t \,\mathrm{d}\boldsymbol{\pi}'(\eta).$$

Since π' was arbitrary, we see that g is locally Sobolev with

$$|\mathrm{d}g|(x') \leq \frac{1}{2T} \int_{-T}^{T} w_{\mathsf{d}}(s) |\mathrm{d}\hat{g}|(\mathsf{Fl}_{s}(x')) \,\mathrm{d}s \stackrel{*}{\leq} |\mathrm{d}g|(x') \qquad \mathfrak{m}'\text{-a.e. } x',$$

where the starred inequality comes from the already proven inequality \leq in (2.4.17). Thus the starred inequality must be an equality, and this forces the equality in (2.4.17) to hold **m**-a.e. on $b^{-1}([-T,T])$. The conclusion follows by the arbitrariness of T.

2.5 Isometry and RCD condition of the quotient space

Lemma 2.5.1. With the same notation and assumptions as in Section 2.2 the following holds.

The warped product space $\mathbb{R} \times_w X'$ has the Sobolev-to-Lipschitz property.

Proof. The arguments used in [DPG, Theorem 3.34] carry over. One first proves that X' is locally doubling (because X is and Pr is locally Lipschitz) and has the measured length property, see Definition 1.7.7 (this follows from the fact that X has such property and the estimate (2.4.12)).

Then Theorem 1.7.11 applies.

The isomorphism between X and $\mathbb{R} \times_w X'$ is reached via duality with Sobolev norms thanks to the following result (see [G13a, Proposition 4.20] for its proof). **Proposition 2.5.2** (Isomorphisms via duality with Sobolev norms). Let $(X_1, \mathsf{d}_1, \mathfrak{m}_1)$ and $(X_2, \mathsf{d}_2, \mathfrak{m}_2)$ be two metric measure spaces with the Sobolevto-Lipschitz property and let $T : X_1 \to X_2$ be a Borel map. Assume that both \mathfrak{m}_1 and \mathfrak{m}_2 give finite mass to bounded sets. Then the following are equivalent:

- i) up to a modification on a m₁-negligible set, T is an isomorphism of metric measure spaces, i.e. T_{*}m₁ = m₂ and d₂(T(x), T(y)) = d₁(x, y) for any x, y ∈ supp(m₁);
- *ii)* the following two are true:
 - ii-a) there exists a Borel \mathfrak{m}_1 -negligible set $\mathcal{N} \subset X_1$ and a Borel map $\mathsf{S}: X_2 \to X_1$ such that $\mathsf{S}(\mathsf{T}(x)) = x$ for every $x \in X_1 \setminus \mathcal{N}$;
 - *ii-b)* the right composition with T produces a bijective isometry of the Sobolev space $W^{1,2}(X_2)$ in $W^{1,2}(X_1)$, *i.e.* $f \in W^{1,2}(X_2)$ if and only if $f \circ \mathsf{T} \in W^{1,2}(X_1)$ and in this case $||f||_{W^{1,2}(X_2)} = ||f \circ \mathsf{T}||_{W^{1,2}(X_1)}$.

Theorem 2.5.3. With the same notation and assumptions as in Section 2.2 the following holds.

The map $T: X \to \mathbb{R} \times_w X'$ is a measure preserving isometry.

Proof. Since we already know that T is measure preserving (by Proposition 2.3.4) and both X and $\mathbb{R} \times_w X'$ have the Sobolev-to-Lipschitz property, according to Proposition 2.5.2 it is sufficient to prove that $f \in W^{1,2}(\mathbb{R} \times_w X')$ if and only if $f \circ \mathsf{T} \in W^{1,2}(\mathsf{X})$ and in this case

$$|\mathrm{d}f|_{\mathbb{R}\times_w \mathrm{X}'} \circ \mathsf{T} = |\mathrm{d}(f \circ \mathsf{T})|_{\mathrm{X}} \qquad \mathfrak{m}\text{-a.e.}.$$
(2.5.1)

By a density argument based on Lemma 2.4.1, it suffices to prove the above for functions f such that $f \circ \mathsf{T} \in \mathcal{A}$.

Corollary 2.4.4 and (1.7.1) ensure that (2.5.1) holds if $f \circ \mathsf{T} \in \mathcal{G}$, while (2.2.1a) give that (2.5.1) holds for $f \circ \mathsf{T} \in \mathcal{H}$ (see also the arguments used in [G13a, Section 6.2]). The conclusion now follows exactly as in [G13a, Section 6.2] (see also [DPG, Section 3.8]) using the fact that both X and $\mathbb{R} \times_w X'$ are infinitesimally Hilbertian (X' is so because of identity (2.4.17), then the property carries to warped products by Theorem 1.7.6) and that functions in \mathcal{G} and \mathcal{H} have orthogonal gradients. We omit the details.

Proposition 2.5.4. With the same notation and assumptions as in Section 2.2 the following holds.

Assume furthermore that for some $\bar{z} \in \mathbb{R}$ we have $\psi_{\mathsf{d}} \leq 0$ on $(-\infty, \bar{z}]$ and $\psi_{\mathsf{d}} \geq 0$ on $[\bar{z}, +\infty)$ (thus in particular $\psi_{\mathsf{d}}(\bar{z}) = 0$). Then $(X', \mathsf{d}', \mathfrak{m}')$ is an $\mathsf{RCD}(\frac{1}{w_{\mathsf{d}}^2(\bar{z})}K, N)$ space.

Proof. As already noticed, by Corollary 2.4.4 it easily follows that X' is infinitesimally Hilbertian, so we need only to prove the CD(K, N) condition.

For $z \in \mathbb{R}$ let $X'_z := b^{-1}(z)$ and equip it with the distance d'_z defined as

$$\mathsf{d}'_z(x,y)^2 := \inf \int_0^1 |\dot{\gamma}_t|^2 \,\mathrm{d}t,$$

the inf being taken among all absolutely continuous curves $\gamma : [0, 1] \to X'_z \subset X$. We also equip X'_z with the measure $\mathfrak{m}'_z := (\mathsf{Fl}_z)_*\mathfrak{m}'$. Then using the equality case in Proposition 2.4.3 and using the fact that X' has the measured length property (briefly mentioned in the proof of Lemma 2.5.1 above) it is not hard to see that $\mathsf{Fl}_z : X' \to X'_z$ satisfies

$$\mathsf{d}'_{z}(\mathsf{Fl}_{z}(x),\mathsf{Fl}_{z}(y)) = w_{\mathsf{d}}(z)\mathsf{d}'(x,y) \qquad \forall x, y \in \mathbf{X}'.$$

Thus if we establish that $X'_{\bar{z}}$ is $\mathsf{RCD}(K, N)$, taking into account how the CD condition scales with the distance (see [S06b, Proposition 1.4]) we conclude. With this said, replacing b with $b + \bar{z}$ we can assume that $\bar{z} = 0$ and then the goal is to prove that X' is $\mathsf{RCD}(K, N)$.

The key geometric property that allows us to conclude, and for which we shall use the assumption on ψ_d , is the following:

Let $\mu_0, \mu_1 \in \mathscr{P}(\mathbf{X})$ be $\ll \mathfrak{m}$ and π so that $W_2^2(\mu_0, \mu_1) = \iint_0^1 |\dot{\gamma}_t|^2 dt d\pi(\gamma)$. Assume that $\operatorname{supp}(\mu_i) \subset b^{-1}([-T, T]), i = 0, 1$. Then π is concentrated on curves lying on $b^{-1}([-T, T])$. (2.5.2)

To see why this holds, let $\varphi : \mathbb{R} \to \mathbb{R}$ be C^2 with bounded second derivative, convex, identically 0 on [-T, T] and strictly positive elsewhere. Then the Regular Lagrangian Flow $(\tilde{\mathsf{Fl}}_t)$ of $\nabla \tilde{b}$ with $\tilde{b} = \varphi \circ b$ is the identity on the strip $b^{-1}([-T, T])$ and converges to the 'projection on the boundary of such strip' outside of it, namely defining $\Pr_T : X \to X$ as:

$$\Pr_{T}(x) := \begin{cases} x, & \text{if } b(x) \in [-T, T], \\ \mathsf{Fl}_{T-b(x)}(x), & \text{if } b(x) > T, \\ \mathsf{Fl}_{-T-b(x)}(x), & \text{if } b(x) < -T, \end{cases}$$

we have $\tilde{\mathsf{Fl}}_s(x) \to \Pr_T(x)$ as $s \uparrow \infty$ for any $x \in X$. Then arguing exactly as for Proposition 2.4.3 we obtain that: for every test plan π we have that for π -a.e. γ it holds

$$\mathsf{ms}(\mathrm{Pr}_{T}(\gamma), t) \le \frac{w_{\mathsf{d}}(\mathrm{b}(\mathrm{Pr}_{T}(\gamma_{t})))}{w_{\mathsf{d}}(\mathrm{b}(\gamma_{t}))} \mathsf{ms}(\gamma, t) \qquad \text{for a.e. } t.$$
(2.5.3)
Now recall that by (2.2.9b) we have $\log(\frac{w_d(z_2)}{w_d(z_1)}) = \int_{z_1}^{z_2} \psi_d d\mathcal{L}^1$ for every $z_1, z_2 \in \mathbb{R}$, $z_1 < z_2$ and use the assumption on ψ_d (with $\bar{z} = 0$) to conclude from (2.5.3) that for π -a.e. γ it holds

$$\mathsf{ms}(\operatorname{Pr}_T(\gamma), t) \le \mathsf{ms}(\gamma, t) \quad \text{for a.e. } t.$$
 (2.5.4)

It directly follows from this that the total kinetic energy of the test plan $\mathsf{KE}(\boldsymbol{\pi}) := \frac{1}{2} \iint_{0}^{1} |\dot{\gamma}_{t}|^{2} dt d\boldsymbol{\pi}(\gamma)$ does not increase under the action of Pr_{T} . In particular, this occurs for $\boldsymbol{\pi}$ as in (2.5.2), thus obtaining $\mathsf{KE}((\mathrm{Pr}_{T})_{*}\boldsymbol{\pi}) \leq \frac{1}{2}W_{2}^{2}(\mu_{0},\mu_{1})$. On the other hand, we know from [GRS] that for $\mu_{0},\mu_{1} \ll \mathfrak{m}$ there is exactly one plan $\boldsymbol{\pi}$ for which $\mathsf{KE}(\boldsymbol{\pi}) \leq \frac{1}{2}W_{2}^{2}(\mu_{0},\mu_{1})$, thus we conclude that $(\mathrm{Pr}_{T})_{*}\boldsymbol{\pi} = \boldsymbol{\pi}$, which is the claim (2.5.2).

A direct consequence of (2.5.2) and the very definition of $\mathsf{CD}(K, N)$ spaces (see Section 1.5.1) is that the space $(X_T, \mathsf{d}_T, \mathfrak{m}_T)$ given by $X_T := b^{-1}([-T, T])$ with d_T being the restriction of the distance and $\mathfrak{m}_T := \frac{1}{\int_{-T}^T w_\mathfrak{m}} \mathfrak{m}|_{b^{-1}(-T,T)}$ is $\mathsf{CD}(K, N)$ as a consequence of $(X, \mathsf{d}, \mathfrak{m})$ being so (recall that a scaling of the measure does not affect Curvature-Dimension bounds). Therefore by the stability of the CD condition (see Theorem 1.5.23) to conclude it suffices to prove that for any fixed $p \in X'$ the spaces $(X_T, \mathsf{d}_T, \mathfrak{m}_T, p)$ converge to $(X', \mathsf{d}', \mathfrak{m}', p)$ as $T \downarrow 0$ in the pointed-measured-Gromov-Hausdorff sense.

To see this, consider the map $\Pr: X_T \to X'$ and notice that $\Pr(p) = p$, that $\Pr_* \mathfrak{m}_T = \mathfrak{m}'$ (by (2.2.4)) and that

$$|\mathsf{d}(\Pr(x),\Pr(y)) - \mathsf{d}(x,y)| \le \mathsf{d}(\Pr(x),x) + \mathsf{d}(\Pr(y),y) \le 2T \qquad \forall x,y \in \mathcal{X}_T.$$

This suffices to prove that $(X_T, \mathsf{d}_T, \mathfrak{m}_T, p)$ pmGH-converge to $(X', \mathsf{d}, \mathfrak{m}', p)$, which therefore is a $\mathsf{CD}(K, N)$ space, so we are left to prove that $\mathsf{d} = \mathsf{d}'$ on X'. To see this, notice that since $(X', \mathsf{d}, \mathfrak{m}')$ is $\mathsf{CD}(K, N)$ and $\operatorname{supp}(\mathfrak{m}') = X'$ (as direct consequence of the assumption $\operatorname{supp}(\mathfrak{m}) = X$), we have that (X', d) is a geodesic space, i.e. given $x, y \in X'$ there is a curve $\gamma : [0, 1] \to X'$ with $\int_0^1 |\dot{\gamma}_t|^2 dt = \mathsf{d}^2(x, y)$. It follows by the very definition of d' that $\mathsf{d}' \leq \mathsf{d}$, and since the other inequality is trivially true, the proof is complete.

Chapter 3

Applications to spaces with positive spectrum

In this chapter, using the general strategy saw in Chapter 2, we prove Theorems 0.0.7 and 0.0.8.

All we have to do is to construct functions with the properties required in Lemma 2.1.1, use them to find the Busemann functions and conclude with Theorem 2.2.1.

In order to find the required functions we will need other preliminary definitions and results that we show in the next two sections.

3.1 Preliminaries: harmonic functions and Bochner inequality

The construction of the Busemann functions we need is based on the strong maximum principle for harmonic functions. In this section we recall the definition of harmonic function and the statement of the maximum principles in the non-smooth setting. Moreover we use the notion of measure valued Laplacian to state an improved version of the Bochner inequality.

We start recalling the definition of measure valued Laplacian, a generalization of the notion of Laplacian that we have seen in Definition 1.3.21 given by Gigli in [G15].

Definition 3.1.1 (Measure valued Laplacian). Let (X, d, \mathfrak{m}) be an infinitesimally Hilbertian space, $U \subset X$ open and $f \in W^{1,2}_{loc}(X)$. We shall say that $f \in D(\Delta_{loc}, U)$ provided there exists a Radon functional μ such that

$$\int \langle \nabla f, \nabla \varphi \rangle \, \mathrm{d}\mathfrak{m} = -\int \varphi \, \mathrm{d}\mu$$

for every $\varphi \in LIP(X)$ with support bounded and contained in U.

In this case the measure μ is denoted by Δf . In case U = X we simply write $f \in D(\Delta_{loc})$.

Recall that a *Radon functional* is a linear functional L from the space $C_{bs}(\mathbf{X})$ of continuous functions on \mathbf{X} with bounded support to \mathbb{R} such that for every $K \subset \mathbf{X}$ there is $C_K \geq 0$ such that

 $|L(f)| \le C_K \sup |f|$ for every $f \in C_{bs}(X)$ with support in K.

Thus Radon functionals should be thought of (and we shall do so) as signed measures that have finite total variation on compact sets, but that in principle might have both positive and negative parts of infinite mass, see also the discussion in [CaMo20], where it was observed that Radon functionals are the correct object to use in place of the Radon measures used in [G15]. Notice also that in the field of non-smooth analysis some authors use the term Radon measure for what we are calling here Radon functionals, see for instance [AB, Remark 2.12].

There are natural compatibility conditions between this notion of Laplacian and the one in Definition 1.3.21, in particular the following proposition holds (see [G15, Chapter 4]).

Proposition 3.1.2. Let (X, d, \mathfrak{m}) be an infinitesimally Hilbertian metric measure space. Assume that $f \in D(\Delta_{loc})$ and $\Delta f = g\mathfrak{m}$ for a certain $g \in L^2_{loc}(X)$. Then $f \in D(\Delta_{loc})$ with $\Delta f = g$.

Moreover, as the Laplacian Δ , the measure valued Laplacian satisfies natural calculus rules.

Proposition 3.1.3 (Leibniz rule for the measure valued Laplacian). Let $f \in D(\Delta_{loc}, U)$ and let $g \in \text{LIP}_{loc}(X) \cap D(\Delta_{loc}, U)$. Then $fg \in D(\Delta_{loc}, U)$ and it holds the following formula:

$$\boldsymbol{\Delta}(fg) = f\boldsymbol{\Delta}g + g\boldsymbol{\Delta}f + \langle \nabla f, \nabla g \rangle \mathfrak{m}_{|_{U}}.$$

Proposition 3.1.4 (Chain rule for the measure valued Laplacian). Let $f \in D(\Delta_{loc}, U) \cap \text{LIP}_{loc}(X)$ and let $\varphi \in C^2(\mathbb{R})$. Then $\varphi \circ f \in D(\Delta_{loc}, U)$ and it holds the following formula:

$$\Delta(\varphi \circ f) = \varphi' \circ f \Delta f + \varphi'' \circ f |\nabla f|^2 \mathfrak{m}_{|_U}.$$
(3.1.1)

These concepts of Laplacian are linked to energy minimizers via Theorems 3.1.5 and 3.1.7 (see [G15], [GMo] and [GR19]).

Theorem 3.1.5. Let (X, d, \mathfrak{m}) be a proper infinitesimally Hilbertian metric measure space, $U \subset X$ open and $f \in L^2_{loc}(U)$. Then the following are equivalent:

- -) $f \in D(\Delta_{loc}, U)$ and $\Delta f = 0$;
- -) $f \in D(\Delta_{loc}, U)$ and $\Delta f = 0;$
- -) for any $g \in W_0^{1,2}(U)$ we have

$$\int_{U} |\mathrm{d}f|^2 \,\mathrm{d}\mathfrak{m} \le \int_{U} |\mathrm{d}(f+g)|^2 \,\mathrm{d}\mathfrak{m}. \tag{3.1.2}$$

Definition 3.1.6 (Harmonic function). Let (X, d, \mathfrak{m}) be a proper infinitesimally Hilbertian space, $U \subset X$ be open and $f \in L^2_{loc}(U)$. We say that fis harmonic if it satisfies any of the three equivalent conditions of Theorem 3.1.5.

In a similar way we can define sub/super-harmonic functions. Indeed, similarly to Theorem 3.1.5, the following holds.

Theorem 3.1.7. Let (X, d, \mathfrak{m}) be a proper infinitesimally Hilbertian metric measure space, $U \subset X$ open and $f \in L^2_{loc}(U)$. Then the following are equivalent:

- -) $f \in D(\Delta_{loc}, U)$ and $\Delta f \ge 0$ (resp. ≤ 0);
- -) (3.1.2) holds for every g negative (resp. positive) $W_0^{1,2}(U)$ function.

Definition 3.1.8 (Sub/Super-harmonic function). Let (X, d, \mathfrak{m}) be a proper infinitesimally Hilbertian, $U \subset X$ be open and $f \in L^2_{loc}(U)$. We say that f is subharmonic (resp. superharmonic) if it satisfies any of the two equivalent conditions of Theorem 3.1.7.

We state now the weak and strong maximum principle in the non-smooth setting. For their proof see [BB] (see also [GR19] for a different proof).

Theorem 3.1.9 (Weak maximum principle). Let $K \in \mathbb{R}$ and let $(X, \mathsf{d}, \mathfrak{m})$ be an $\mathsf{RCD}(K, \infty)$ space. Let U be an open bounded subset of X and let $f \in W^{1,2}_{loc}(U) \cap C(\overline{U})$ be subharmonic. Then

$$\sup_{U} f \le \sup_{\partial U} f.$$

Theorem 3.1.10 (Strong maximum principle). Let $K \in \mathbb{R}$ and $N \in [1, +\infty)$, and let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ space. Let U be an open and connected subset of X and let $f \in W^{1,2}_{loc}(U) \cap C(\overline{U})$ be subharmonic and such that for some $\bar{x} \in U$ it holds

$$f(\bar{x}) = \max_{x \in \bar{U}} f(x).$$

Then f is constant.

In order to study the properties of specific harmonic functions we will rely on the following regularity result for harmonic functions, that extend to the RCD setting a classical gradient estimate due to Cheng and Yau (see [CY]) in the smooth setting. Its proof in the RCD setting can be found in [HKX].

Theorem 3.1.11 (Cheng-Yau type gradient estimate). Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K,N)$ metric measure space with $K \leq 0$ and $N \in [1,\infty)$. Then there exists a constant C depending only on N such that every positive harmonic function u on any geodesic ball $B_{2R} \subset X$ satisfies

$$\frac{|\mathrm{d}u|}{u} \le C \frac{1 + \sqrt{-KR}}{R} \qquad in \ B_R. \tag{3.1.3}$$

We conclude this introductory section recalling (a suitable version of) the Bochner inequality. In order to state it, we need to recall the concept of essential dimension $\dim(X)$ of a finite dimensional RCD space.

Theorem 3.1.12. Let X be an $\mathsf{RCD}(K, N)$ space with $K \in \mathbb{R}$ and $N \in$ $[1, +\infty)$. Then there exists an integer dim $(X) \in [1, N]$ such that the tangent module $L^0(TX)$ has constant dimension equal to dim(X).

The proof of this result is highly non-trivial, and ultimately coming from [BS20a] (but see also [G13a], [MN], [DPR], [DPMR], [KM], [GP22], [GP21]).

We can now state the desired inequality.

Theorem 3.1.13 (Improved Bochner Inequality). Let X be an $\mathsf{RCD}(K, N)$ space with $K \in \mathbb{R}$ and $N \in [1, +\infty)$. Then for any $f \in \text{Test}_{loc}(X)$ we have $|\mathrm{d}f|^2 \in D(\mathbf{\Delta}_{loc})$ and

$$\boldsymbol{\Delta}\left(\frac{|\mathrm{d}f|^2}{2}\right) \ge \left(|\operatorname{Hess}(f)|^2_{\mathsf{HS}} + K|\mathrm{d}f|^2 + \langle \mathrm{d}f, \mathrm{d}\Delta f \rangle + \frac{(\Delta f - \operatorname{tr}\operatorname{Hess}(f))^2}{N - \dim(\mathsf{X})}\right)\mathfrak{m},$$
(3.1.4)

where $\frac{(\Delta f - \operatorname{tr} \operatorname{Hess}(f))^2}{N-n}$ is taken to be 0 in the case dim(X) = N.

This result was proved in [Han] (strongly based on the earlier [GKO], [AGS14b], [EKS], [G18b]).

3.2 Preliminaries: sets of finite perimeter and Gauss-Green formula

In this section we recall the statement of the Gauss-Green formula for RCD spaces proved by E. Bruè, E. Pasqualetto and D. Semola in [BPS]. More precisely, we give the definitions and results in [DGP] and [BPS] that allow us to generalize in the non-smooth setting the formula

$$\int_{E} \operatorname{div}(v) \, \mathrm{d}x = \int_{\partial E} v \cdot \nu \, \mathrm{d}\mu_{\partial E}, \qquad (3.2.1)$$

where E is a regular bounded subset of \mathbb{R}^n , v is a C^1 vector field on a neighborhood of \overline{E} , $\mu_{\partial E}$ is the (n-1)-dimensional measure on ∂E and ν is the outward normal vector field defined on ∂E .

With the structure we presented in the previous sections we are already able to write the left hand side of (3.2.1).

To define the measure on ∂E we recall the notion of perimeter of a set.

Definition 3.2.1 (Perimeter and sets of finite perimeter). Given a Borel set $E \subset X$ and an open set $A \subset X$ we define the perimeter Per(E, A) as

$$\operatorname{Per}(E,A) := \inf \left\{ \liminf_{n \to +\infty} \int_{A} |\mathrm{d}u_{n}| \,\mathrm{d}\mathfrak{m} : u_{n} \in \operatorname{LIP}_{loc}(A), u_{n} \xrightarrow{L_{loc}^{1}(A,\mathfrak{m})} \chi_{E} \right\}.$$

We say that E has finite perimeter if $Per(E) < +\infty$. In this case it can be proved that the set function $A \mapsto Per(E, A)$ is the restriction to open sets of a finite and positive Borel measure $Per(E, \cdot)$ on X defined by

 $\operatorname{Per}(E,B) := \inf \left\{ \operatorname{Per}(E,A) : A \text{ open and } A \supset B \right\}.$

The measure $Per(E, \cdot)$ is usually denoted with $|D\chi_E|$: this notation comes from the fact that a set has finite perimeter if and only if its characteristic function χ_E has bounded variation, and in this case its total variation coincides with the perimeter measure.

An important result about sets of finite perimeter is the coarea formula that is well known to be valid even in the metric setting (see [BPS], [BCM] and references therein, starting from the original [Mir], where the first instance of the coarea formula in the metric setting has been obtained).

Theorem 3.2.2 (Coarea formula). Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ metric measure space with $N < +\infty$ and let $U \subset X$ open. Let $u \in \mathrm{LIP}_{loc}(U)$ be positive and such that $u^{-1}([a, b])$ is compact in U for every $[a, b] \subset (0, 1)$. Then $\{u < t\}$ has finite perimeter for a.e. $t \in (0, 1)$, and for any $f : U \to [-\infty, +\infty]$ Borel and in $L^1_{loc}(U, |\nabla u|\mathfrak{m}|_U)$ it holds that

$$\int_{U} \varphi(u) f |\mathrm{d}u| \,\mathrm{d}\mathfrak{m} = \int_{U} \varphi(t) \int f \,\mathrm{d}\operatorname{Per}(\{u < t\}, \cdot) \,\mathrm{d}t$$

for every $\varphi : [0,1] \to \mathbb{R}$ Borel with $\operatorname{supp}(\varphi) \subset (0,1)$.

The other instrument we need is the outward normal ν_E to a set E with finite perimeter. We note that the theory of L^0 -modules is not enough for this, since vector fields as we introduced them are defined only **m**-a.e..

3.2.1 Capacity and quasi-continuous functions

In this section we introduce the concept of capacity of a set, and thanks to that we define quasi-continuous functions. The results in this section are classical (see for instance [EG]).

Definition 3.2.3 (Capacity). Let X be a metric measure space and fix $E \subset X$. We indicate with \mathscr{F}_E the set of $W^{1,2}(X)$ functions f which satisfy $f|_U \geq 1$ for some U open neighborhood of E.

The capacity of E, $\operatorname{Cap}(E) \in [0, +\infty]$ is defined as

$$\operatorname{Cap}(E) := \begin{cases} \inf_{f \in \mathscr{F}_E} \|f\|_{W^{1,2}(X)}^2 & \text{if } \mathscr{F}_E \neq \emptyset; \\ +\infty & \text{if } \mathscr{F}_E = \emptyset \end{cases}$$

Proposition 3.2.4. The capacity Cap is a submodular (for every $E, F \subset X$ it holds $\operatorname{Cap}(E \cup F) + \operatorname{Cap}(E \cap F) \leq \operatorname{Cap}(E) + \operatorname{Cap}(F)$) outer measure on X. Moreover it is bounded on bounded sets and for every Borel set $E \subset X$ it holds $\mathfrak{m}(E) \leq \operatorname{Cap}(E)$.

Remark 3.2.5. Via Cavalieri's formula we can define the integral with respect to outer measures. In our case, for every function $f: X \to [0, +\infty]$ we define

$$\int f \, \mathrm{dCap} := \int_0^{+\infty} \mu(\{f > t\}) \, \mathrm{d}t. \tag{3.2.2}$$

Since Cap is submodular then the integral defined in (3.2.2) is subadditive, and this is used to prove that the function d_{Cap} below is a distance.

Definition 3.2.6 (The space $L^0(\text{Cap})$). We say that two Borel functions $f, g : X \to \mathbb{R}$ are equal Cap-a.e. if it holds $\text{Cap}(\{f \neq g\}) = 0$, and we

indicate with $L^0(Cap)$ the set of all equivalence classes of Borel functions up to Cap-a.e. equality.

Moreover, let $(A_k)_k$ be an increasing sequence of open subsets of X with finite capacity such that for every bounded $B \subset X$ there exists a $k \in \mathbb{N}$ such that $B \subset A_k$. We define the following distance on $L^0(\text{Cap})$:

$$\mathsf{d}_{\operatorname{Cap}}(f,g) := \sum_{k \in \mathbb{N}} \frac{1}{2^k (\operatorname{Cap}(A_k) \lor 1)} \int_{A_k} |f - g| \land 1 \operatorname{dCap}.$$

Proposition 3.2.7. The metric space $(L^0(Cap), \mathsf{d}_{Cap})$ is complete.

Definition 3.2.8 (Quasi-continuous function). A function $f : X \to \mathbb{R}$ is said to be quasi-continuous if for every $\varepsilon > 0$ there exists a Borel set $E \subset X$ with $\operatorname{Cap}(E) < \varepsilon$ such that the function $f|_{X \setminus E} : X \setminus E \to \mathbb{R}$ is continuous.

Remark 3.2.9. By definitions easily follows that the set of quasi-continuous functions (up to equality Cap-a.e.) is contained in $L^0(Cap)$.

Theorem 3.2.10 (Quasi-continuous representative of Sobolev functions). Let (X, d, \mathfrak{m}) be an infinitesimally Hilbertian metric measure space. Then there exists a unique continuous map QCR : $W^{1,2}(X) \to L^0(\text{Cap})$ such that for every $f \in W^{1,2}(X)$ the function QCR(f) is quasi-continuous and \mathfrak{m} -a.e. it holds QCR(f) = f.

3.2.2 $L^0(Cap)$ -modules and quasi-continuous vector fields

In this section we define the tangent $L^0(\text{Cap})$ -module and see, similarly to quasi-continuous functions, that Sobolev vector fields have a quasi-continuous representative. The proofs of the results in this section can be found in [DGP].

Similarly to what we saw in Definition 1.3.2, we define $L^0(\text{Cap})$ -normed $L^0(\text{Cap})$ -modules.

Definition 3.2.11 ($L^0(Cap)$ -normed $L^0(Cap)$ -module). Let (X, d, \mathfrak{m}) be a metric measure space. We say that a quadruple ($\mathscr{M}, \tau, \cdot, |\cdot|$) is a $L^0(Cap)$ -normed $L^0(Cap)$ -module over (X, d, \mathfrak{m}) if the following hold:

i) (\mathcal{M}, τ) is a topological vector space;

ii) the bilinear map $\cdot : L^0(\operatorname{Cap}) \times \mathscr{M} \to \mathscr{M}$ satisfies

$$f \cdot (g \cdot v) = (fg) \cdot v \qquad for \ every \ f, g \in L^0(\text{Cap}) \ and \ v \in \mathscr{M},$$
$$\hat{1} \cdot v = v \qquad for \ every \ v \in \mathscr{M},$$

where $\hat{1}$ is the function identically equal to 1 on X;

iii) the pointwise norm $|\cdot|: \mathscr{M} \to L^0(\operatorname{Cap})$ satisfies

$$\begin{split} |v| &\geq 0 & \text{for every } v \in \mathscr{M}, \text{ equality if and only if } v = 0, \\ |v+w| &\leq |v|+|w| & \text{for every } v, w \in \mathscr{M}, \\ |f \cdot v| &= |f||v| & \text{for every } f \in L^0(\text{Cap}) \text{ and } v \in \mathscr{M}, \end{split}$$

where all equalities and inequalities are intended in the Cap-a.e. sense;

iv) taking $(A_k)_k$ as in Definition 3.2.6, the distance $\mathsf{d}_{\mathscr{M}}$ on \mathscr{M} defined as

$$\mathsf{d}_{\mathscr{M}}(v,w) := \sum_{k \in \mathbb{N}} \frac{1}{2^k (\operatorname{Cap}(A_k) \lor 1)} \int_{A_k} |v - w| \land 1 \operatorname{dCap}$$

for every $v, w \in \mathcal{M}$ is complete and induces the topology τ .

Definition 3.2.12 (Hilbert $L^0(\text{Cap})$ -module). Let \mathscr{M} be a $L^0(\text{Cap})$ -normed $L^0(\text{Cap})$ -module. We say that it is a Hilbert module if the parallelogram rule

$$|v + w|^{2} + |v - w|^{2} = 2(|v|^{2} + |w|^{2})$$

holds Cap-a.e. in X for every $v, w \in \mathcal{M}$.

With the following theorem we define the tangent $L^0(Cap)$ -module.

Theorem 3.2.13 (Tangent $L^0(\text{Cap})$ -module). Let $K \in \mathbb{R}$ and let $(X, \mathsf{d}, \mathfrak{m})$ be an $\text{RCD}(K, \infty)$ space. Then there exists a unique couple $(L^0_{\text{Cap}}(TX), \overline{\nabla})$ where $L^0_{\text{Cap}}(TX)$ is an $L^0(\text{Cap})$ -module over X and $\overline{\nabla}$: $\text{Test}(X) \to L^0_{\text{Cap}}(TX)$ is a linear operator satisfying the following properties:

- i) for every $f \in \text{Test}(X)$ the equality $|\overline{\nabla}f| = |\mathsf{QCR}(|Df|)|$ holds Cap-a.e. on X;
- *ii)* the space

$$\left\{\sum_{n\in\mathbb{N}}\chi_{E_n}\bar{\nabla}f_n:(f_n)_{n\in\mathbb{N}}\subset\operatorname{Test}(\mathbf{X})\ and\ (E_n)_{n\in\mathbb{N}}\ Borel\ partition\ of\ \mathbf{X}\right\}$$

is dense in $L^0_{Cap}(TX)$.

Uniqueness is intended up to unique isomorphism.

The space $L^0_{\text{Cap}}(TX)$ is called tangent $L^0(\text{Cap})$ -module, its elements are the Cap-vector fields, and the operator $\overline{\nabla}$ is called gradient.

Proposition 3.2.14. The tangent $L^0(\text{Cap})$ -module $L^0_{\text{Cap}}(TX)$ is a Hilbert module.

3.2 Preliminaries: sets of finite perimeter and Gauss-Green formula

We define, similarly to what we did for test vector fields and $H_C^{1,2}$ in Section 1.6, test Cap-vector fields and the set of quasi-continuous vector fields.

Definition 3.2.15 (Test Cap-vector fields). We define the set of test Capvector fields Test $\overline{\mathbf{V}} \subset L^0_{\operatorname{Cap}}(T\mathbf{X})$ as

$$\operatorname{Test}\bar{\mathcal{V}}(\mathcal{X}) := \left\{ \sum_{i=0}^{n} \mathsf{QCR}(g_i) \bar{\nabla} f_i : n \in \mathbb{N}, \ (f_i)_{i=0}^{n}, (g_i)_{i=0}^{n} \subset \operatorname{Test}(\mathcal{X}) \right\}.$$

Definition 3.2.16 (Quasi-continuous Cap-vector field). The set of quasicontinuous vector fields $\mathcal{QC}(TX)$ is defined as the closure of TestV(X) in $L^0_{\text{Cap}}(T\mathbf{X}).$

We conclude this section recalling the last result of [DGP], which prove the existence of a (unique) quasi-continuous representative of $H_C^{1,2}(X)$ vector fields.

We indicate with \overline{Pr} the natural projection from $L^0_{\text{Cap}}(TX)$ to $L^0(TX)$, i.e. the linear continuous operator such that

- i) $\overline{\Pr}(\overline{\nabla}f) = \nabla f$ for every $f \in \text{Test}(X)$;
- *ii*) $\overline{\Pr}(gv) = \Pr(g)\overline{\Pr}(v)$ for every $g \in L^0(\operatorname{Cap})$ and $v \in L^0_{\operatorname{Cap}}(TX)$, where Pr is the natural projection from $L^0(\text{Cap})$ to $L^0(\mathfrak{m})$.

Theorem 3.2.17 (Quasi-continuous representative of Sobolev vector fields). Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, \infty)$ space for some $K \in \mathbb{R}$. Then there exists a unique map $Q\bar{C}R : H_C^{1,2}(TX) \to QC(TX)$ such that $\bar{Pr} \circ Q\bar{C}R : H_C^{1,2}(TX) \to L^0(TX)$ coincides with the inclusion $H_C^{1,2}(TX) \subset L^0(TX)$. Moreover $Q\bar{C}R$ is linear and for every $v \in H_C^{1,2}(TX)$ it holds $|Q\bar{C}R(v)| = 0$

QCR(|v|).

3.2.3Tangent module over ∂E and Gauss-Green formula

In this section we recall the main results of [BPS, Chapter 2]: the existence and uniqueness of a tangent module over the boundary of a set with finite perimeter and the Gauss-Green formula for RCD spaces.

In order to do that we observe that the following proposition holds.

Proposition 3.2.18. Let X be an $\mathsf{RCD}(K, N)$ metric measure space and let $E \subset X$ be of finite perimeter. Then $|D\chi_E| \ll \text{Cap}$.

We will indicate with $\pi_{|D\chi_E|}$ the projection of $L^0(\text{Cap})$ on $L^0(|D\chi_E|)$.

Definition 3.2.19 (Trace operator over ∂E). We define the trace operator over the boundary of E as the function $\operatorname{tr}_E : W^{1,2}(X) \to L^0(|D\chi_E|)$ given by

$$\operatorname{tr}_E := \pi_{|D}\chi_{E|} \circ \operatorname{\mathsf{QCR}}$$
 .

Theorem 3.2.20 (Tangent module over ∂E). Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ space and let $E \subset X$ be a set of finite perimeter. Then there exists a unique couple $(L_E^2(TX), \tilde{\nabla})$ where $L_E^2(TX)$ is an $L^2(|D\chi_E|)$ -normed $L^{\infty}(|D\chi_E|)$ module and $\tilde{\nabla}$: Test $(X) \to L_E^2(TX)$ is a linear operator such that

- i) the equality $|\tilde{\nabla}f| = \operatorname{tr}_E(|\nabla f|)$ holds $|D\chi_E|$ -a.e. for every $f \in \operatorname{Test}(X)$;
- *ii)* the set

$$\left\{\sum_{i=1}^{n} \chi_{E_i} \tilde{\nabla} f_i : (E_i)_{i=1}^n \text{ is a Borel partition of X}, \ (f_i)_{i=1}^n \subset \text{Test}(X)\right\}$$

is dense in $L^2_E(TX)$.

Uniqueness is intended up to unique isomorphism.

The space $L^2_E(TX)$ is called tangent module over the boundary of the set E and the operator $\tilde{\nabla}$ is called gradient.

Similarly to how we defined the trace operator of functions over ∂E , we define the trace operator of vector fields. To do that we indicate with $\bar{\pi}_{|D\chi_E|}$ the projection of $L^0_{\text{Cap}}(TX)$ on $L^0_E(TX)$, the L^0 -completion of $L^2_E(TX)$.

Definition 3.2.21 (Trace operator of vector fields). We define the trace operator of vector fields over ∂E as the function $\bar{\operatorname{tr}}_E : H^{1,2}_C \cap L^{\infty}(TX) \to L^2_E(TX)$ defined by

$$\bar{\mathrm{tr}}_E := \bar{\pi}_{|D}\chi_{E|} \circ \mathsf{Q}\bar{\mathsf{C}}\mathsf{R}$$

Theorem 3.2.22 (Gauss-Green formula on RCD spaces). Let (X, d, \mathfrak{m}) be an RCD(K, N) space and let $E \subset X$ be a set of finite perimeter such that $\mathfrak{m}(E) < +\infty$. Then there exists a unique vector field $\nu_E \in L^2_E(TX)$ such that $|\nu_E| = 1$ holds Per (E, \cdot) -a.e. and

$$\int_{E} \operatorname{div}(v) \,\mathrm{d}\mathfrak{m} = -\int \langle \bar{\operatorname{tr}}_{E}(v), \nu_{E} \rangle \,\mathrm{d}\operatorname{Per}(E, \cdot)$$
(3.2.3)

for every $v \in H^{1,2}_C(T\mathbf{X}) \cap D(\operatorname{div})$ with $|v| \in L^{\infty}(\mathfrak{m})$.

A similar result for $\mathsf{RCD}(K, \infty)$ space can be found in [BCM].

3.3 Volume of the ends and harmonic functions

The concept of 'end' borns in the smooth category of Riemannian manifolds, but it can easily be adapted to metric spaces.

Definition 3.3.1. Let (X, d) be a metric space and $K \subset X$ compact. A set $E \subset X$ is called end of X with respect to K provided:

- -) E is an unbounded connected component of $X \setminus K$;
- -) for any $K' \supset K$ compact the set $E \setminus K'$ has only one unbounded connected component.

Suppose that (X, d) is equipped with a Radon measure $\mathfrak{m} \geq 0$. Then we say that an end E has infinite volume if $\mathfrak{m}(E) = +\infty$.

Notice that if E is an end of X with respect to K and $K' \supset K$ is compact, then the only unbounded connected component E' of $E \setminus K'$ is an end with respect to K'. Also, in this case E has infinite volume if and only if E' does.

Let E be an end of X with respect to K and let $p \in K$. We indicate with $E(R) := E \cap B_R(p)$ for every R > dist(E,p). Moreover we define $\partial E := \partial K \cap \overline{E}$ and $\partial E(R) := \partial B_R(p) \cap E$.

We indicate with $V_E(R)$ the volume of E(R) and with $V_E(\infty)$ the volume of the end E.

We conclude defining the first eigenvalue of the Laplacian λ_1 .

Definition 3.3.2 (First eigenvalue of the Laplacian). Let (X, d, \mathfrak{m}) be a metric measure space. We define

$$\lambda_1 := \inf \left\{ \frac{\int_{\mathbf{X}} |\mathrm{d}f|^2 \,\mathrm{d}\mathfrak{m}}{\int_{\mathbf{X}} |f|^2 \,\mathrm{d}\mathfrak{m}} : f \in W^{1,2}(\mathbf{X}), \ \int f^2 \,\mathrm{d}\mathfrak{m} \neq 0 \right\}.$$

Notice that the definition makes sense on arbitrary metric measure spaces, regardless of the linearity of the Laplacian (but we shall only work on infinitesimally Hilbertian spaces).

In this section, following the steps used by Li and Wang, we prove that, assuming $\lambda_1 > 0$, an end has infinite volume if and only if there exists a non-constant bounded harmonic function on it. In order to do this we begin studying some decay estimates for a class of harmonic functions.

In [GV, Section B.1] the following result has been established, the point being the continuity at ∂K^r (see also [BB]). Below for $K \subset X$ and r > 0 we put

$$\mathbf{K}^r := \{ x : \mathsf{d}(x, \mathbf{K}) < r \}.$$

We consider such enlargements to gain sufficient regularity of the boundary to be sure that the harmonic function given by the statement below is continuous up to the boundary.

Theorem 3.3.3. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ space with $K \in \mathbb{R}$ and $N \in [1, +\infty)$, and fix $K \subset X$ a bounded subset and r > 0. Let also $B \subset X$ be a ball containing K^r . Then there is $f \in W_0^{1,2}(B) \cap C(B)$ with $0 \le f \le 1$ on B that is harmonic on $B \setminus \overline{K^r}$ and equal to 1 in $\overline{K^r}$.

In particular, let E be an end of X, say with respect to a compact set K. Then

E is also an end with respect to the compact set $K' := \overline{K^1}$. (3.3.1)

This means that for $p \in K'$ and R > 0 large enough we can apply the above with $B = B_R(p)$ with R > 0 big enough to find $f_R \in W_0^{1,2}(B) \cap C(B)$ with $0 \le f_R \le 1$ on B that is harmonic on $B \setminus K'$ and equal to 1 in K'.

Thanks to the maximum principle (see Theorems 3.1.9 and 3.1.10) the functions f_R generated above are pointwise increasing in R, thus we can define:

Definition 3.3.4. The function $f_E : E \setminus K' \to [0, 1]$ is defined as the pointwise limit of the f_R 's described above.

It is easy to verify that f_E depends only on E and K', and not on the particular $p \in K'$ chosen, and that it is harmonic. The dependence of this function on K' is omitted from the notation for brevity. Also, in what follows we relabel E to be $E \setminus K'$.

Our first goal is to prove some key decay estimates for the function f_E .

In all the following for a fixed point $p \in X$ we indicate with $\mathsf{d}_p : X \to \mathbb{R}$ the function $\mathsf{d}(\cdot, p)$.

Lemma 3.3.5. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ metric measure space and E an end of it. Assume that $\lambda_1 > 0$. Then for the harmonic function f_E given by Definition 3.3.4 there exists a constant $C = C(E, K, \lambda_1)$ such that for every R large enough the following estimates hold:

$$\int_{E(R+1)\setminus E(R)} f_E^2 \,\mathrm{d}\mathfrak{m} \le C e^{-2\sqrt{\lambda_1}R},\tag{3.3.2}$$

$$\int_{E(R+1)\setminus E(R)} |\nabla f_E|^2 \,\mathrm{d}\mathfrak{m} \le C e^{-2\sqrt{\lambda_1}R},\tag{3.3.3}$$

$$\int_{E(R)} e^{2\sqrt{\lambda_1} \mathsf{d}_p} |\nabla f_E|^2 \, \mathrm{d}\mathfrak{m} \le CR.$$
(3.3.4)

Proof. Let $K \subset X$ be compact, put $K' := \overline{K^1}$ and, as before, say that E is an end with respect to K'. Also, let $p \in X$ and $R_0 > 0$ fixed so that $K' \subset B_{R_0}(p)$. Recall that we put $E(R) := E \cap B_R(p)$ and that the function $f_R \in W_0^{1,2} \cap C(B_R(p))$ has been defined after Theorem 3.3.3.

Step 0. We claim that for every $g \in LIP_{loc}(X)$ be identically 0 on K' and for every R large enough it holds

$$\int_{E(R)} |\nabla(gf_R)|^2 \,\mathrm{d}\mathfrak{m} = \int_{E(R)} |\nabla g|^2 f_R^2 \,\mathrm{d}\mathfrak{m}.$$
(3.3.5)

Using the Leibniz and chain rules we compute

$$\begin{split} \int_{E(R)} \left| \nabla(gf_R) \right|^2 \, \mathrm{d}\mathfrak{m} &= \int_{E(R)} \left| \nabla g \right|^2 f_R^2 \, \mathrm{d}\mathfrak{m} + \int_{E(R)} g^2 \left| \nabla f_R \right|^2 \, \mathrm{d}\mathfrak{m} \\ &+ \int_{E(R)} f_R \left\langle \nabla(g^2), \nabla f_R \right\rangle \, \mathrm{d}\mathfrak{m}. \end{split}$$

Now we notice that the integration by parts in

$$\int_{E(R)} f_R \left\langle \nabla(g^2), \nabla f_R \right\rangle \, \mathrm{d}\mathfrak{m} = -\int_{E(R)} g^2 \operatorname{div}(f_R \nabla f_R) \, \mathrm{d}\mathfrak{m}$$
$$= -\int_{E(R)} g^2 |\nabla f_R|^2 \, \mathrm{d}\mathfrak{m}$$

is justified by the fact that $\partial(E(R)) \subset (\partial K') \cup (\partial B_R(p))$ and the assumption $g \equiv 0$ on K' and $f_R \in W_0^{1,2}(B_R(p))$. The claim (3.3.5) follows.

The same line of thought proves that

$$\int_{E} |\nabla(gf_E)|^2 \,\mathrm{d}\mathfrak{m} = \int_{E} |\nabla g|^2 f_E^2 \,\mathrm{d}\mathfrak{m}$$
(3.3.6)

if g is as above and moreover in $W^{1,2}(X)$.

Step 1. Let $\xi := \exp(\sqrt{\lambda_1} \mathsf{d}_p)$. We claim that for some $C = C(E, \mathbf{K}, \lambda_1)$ we have

$$\int_{E} \xi^{2\delta} f_E^2 \,\mathrm{d}\mathfrak{m} \le \frac{C}{(1-\delta)^2} \qquad \text{for every } \delta \in (0,1). \tag{3.3.7}$$

Let $\varphi \in \text{LIP} \cap C_b(\mathbf{X})$ be defined as $\varphi := \hat{\varphi} \circ \mathsf{d}_p$, where

$$\hat{\varphi}(z) := \begin{cases} 0 & \text{if } z \le R_0, \\ \frac{z - R_0}{R_0} & \text{if } z \in [R_0, 2R_0], \\ 1 & \text{if } z \ge 2R_0, \end{cases}$$

and let $R > 2R_0$. By (3.3.5) and the Cauchy-Schwarz inequality we have that

$$\begin{split} \int_{E(R)} \left| \nabla(\varphi \xi^{\delta} f_{R}) \right|^{2} \mathrm{d}\mathfrak{m} &= \int_{E(R)} |\nabla(\varphi \xi^{\delta})|^{2} f_{R}^{2} \mathrm{d}\mathfrak{m} \\ &\leq (1+\varepsilon) \int_{E(R)} \varphi^{2} |\nabla \xi^{\delta}|^{2} f_{R}^{2} \mathrm{d}\mathfrak{m} \\ &\quad + \left(1 + \frac{1}{\varepsilon}\right) \int_{E(R)} \xi^{2\delta} |\nabla \varphi|^{2} f_{R}^{2} \mathrm{d}\mathfrak{m} \\ &\leq (1+\varepsilon) \delta^{2} \lambda_{1} \int_{E(R)} \varphi^{2} \xi^{2\delta} f_{R}^{2} \mathrm{d}\mathfrak{m} \\ &\quad + \left(1 + \frac{1}{\varepsilon}\right) \frac{1}{R_{0}^{2}} \int_{E(2R_{0}) \setminus E(R_{0})} \xi^{2\delta} f_{R}^{2} \mathrm{d}\mathfrak{m} \end{split}$$

By definition of λ_1 we have $\lambda_1 \int_{E(R)} \varphi^2 \xi^{2\delta} f_R^2 \, \mathrm{d}\mathfrak{m} \leq \int_{E(R)} |\nabla(\varphi \xi^{\delta} f_R)|^2 \, \mathrm{d}\mathfrak{m}$ and thus

$$\lambda_1 \left(1 - (1+\varepsilon)\delta^2 \right) \int_{E(R)} \varphi^2 \xi^{2\delta} f_R^2 \,\mathrm{d}\mathfrak{m} \le (1+\frac{1}{\varepsilon}) \frac{1}{R_0^2} \int_{E(2R_0) \setminus E(R_0)} \xi^{2\delta} f_R^2 \,\mathrm{d}\mathfrak{m},$$

and choosing $\varepsilon = \frac{1-\delta}{\delta}$, recalling that $0 \le f_R \le 1$ we obtain

$$\lambda_1 (1-\delta)^2 \int_{E(R)} \varphi^2 \xi^{2\delta} f_R^2 \, \mathrm{d}\mathfrak{m} \le \frac{1}{R_0^2} \int_{E(2R_0) \setminus E(R_0)} \xi^{2\delta} f_R^2 \, \mathrm{d}\mathfrak{m} \le C.$$
(3.3.8)

Since φ is positive and identically equal to 1 on $E(R) \setminus E(2R_0)$, we have that

$$\begin{split} \int_{E(R)} \xi^{2\delta} f_R^2 \, \mathrm{d}\mathfrak{m} &= \int_{E(R) \setminus E(2R_0)} \xi^{2\delta} f_R^2 \, \mathrm{d}\mathfrak{m} + \int_{E(2R_0)} \xi^{2\delta} f_R^2 \, \mathrm{d}\mathfrak{m} \\ &\leq \int_{E(R)} \varphi^2 \xi^{2\delta} f_R^2 \, \mathrm{d}\mathfrak{m} + C \stackrel{(3.3.8)}{\leq} \frac{C}{(1-\delta)^2} + C = \frac{C}{(1-\delta)^2} \end{split}$$

and letting $R \to \infty$ we get the claim (3.3.7).

Step 2. Put for simplicity

$$F(R) := \int_{E(R)} \xi^2 f_E^2 \,\mathrm{d}\mathfrak{m} \qquad \text{for every } R \ge R_0. \tag{3.3.9}$$

Let $R_0 < R_1 < R$. We claim that for any $t \in (0, R - R_1)$ it holds

$$\frac{2\sqrt{\lambda_1}t}{(R-R_1)^2}(F(R-t) - F(R_1)) \le \frac{2\sqrt{\lambda_1}(R_1 - R_0) + 1}{(R_1 - R_0)^2}(F(R_1) - F(R_0)) + \frac{1}{(R-R_1)^2}(F(R) - F(R_1)).$$
(3.3.10)

To see this let $\psi \in LIP(X)$ be defined as $\psi := \hat{\psi} \circ \mathsf{d}_p$, where

$$\hat{\psi}(z) := \begin{cases} \frac{z - R_0}{R_1 - R_0} & \text{if } z \in [R_0, R_1], \\ \frac{R - z}{R - R_1} & \text{if } z \in [R_1, R], \\ 0 & \text{otherwise.} \end{cases}$$

By definition of λ_1 , (3.3.6) and the identity $\nabla \xi = \sqrt{\lambda_1} \xi \nabla \mathsf{d}_p$ we get

$$\begin{split} \lambda_1 \int_E \psi^2 \xi^2 f_E^2 \, \mathrm{d}\mathfrak{m} &\leq \int_E |\nabla \left(\psi \xi f_E\right)|^2 \, \mathrm{d}\mathfrak{m} = \int_E |\nabla \left(\psi \xi\right)|^2 f_E^2 \, \mathrm{d}\mathfrak{m} \\ &= \int_E |\nabla \psi|^2 \xi^2 f_E^2 \, \mathrm{d}\mathfrak{m} + \lambda_1 \int_E \psi^2 \xi^2 f_E^2 \, \mathrm{d}\mathfrak{m} \\ &\quad + 2\sqrt{\lambda_1} \int_E \psi \xi^2 \langle \nabla \psi, \nabla \mathsf{d}_p \rangle f_E^2 \, \mathrm{d}\mathfrak{m}, \end{split}$$

that can be rewritten as

$$-2\sqrt{\lambda_1}\int_E \psi\xi^2 \langle \nabla\psi, \nabla \mathsf{d}_p \rangle f_E^2 \,\mathrm{d}\mathfrak{m} \leq \int_E |\nabla\psi|^2 \,\xi^2 f_E^2 \,\mathrm{d}\mathfrak{m}.$$

Using the explicit expression of ψ , this can be further rewritten as

$$\begin{split} &\frac{2\sqrt{\lambda_1}}{(R-R_1)^2} \int_{E(R)\setminus E(R_1)} (R-\mathsf{d}_p)\xi^2 f_E^2 \,\mathrm{d}\mathfrak{m} \\ &\leq \frac{2\sqrt{\lambda_1}}{(R_1-R_0)^2} \int_{E(R_1)\setminus E(R_0)} \underbrace{(\mathsf{d}_p-R_0)}_{(\leq R_1-R_0 \text{ on } E(R_1)\setminus E(R_0))} \xi^2 f_E^2 \,\mathrm{d}\mathfrak{m} \\ &\quad + \frac{1}{(R_1-R_0)^2} \int_{E(R_1)\setminus E(R_0)} \xi^2 f_E^2 \,\mathrm{d}\mathfrak{m} + \frac{1}{(R-R_1)^2} \int_{E(R)\setminus E(R_1)} \xi^2 f_E^2 \,\mathrm{d}\mathfrak{m} \\ &\leq \text{Right Hand Side of } (3.3.10). \end{split}$$

Then to get (3.3.10) notice that for $t \in (0, R - R_1)$ we have $t \leq R - \mathsf{d}_p$ on $E(R-t) \setminus E(R_1)$.

Step 3. We claim that there exists a constant $C = C(E, K, \lambda_1) > 0$ such that

$$F(R) \leq CR$$
 for all R large enough and F as in (3.3.9). (3.3.11)

To see this pick t = 1, $R_1 = R_0 + 1$ and replace R with R + 1 in (3.3.10) to get, after an easy manipulation, that

$$F(R) - F(R_0 + 1) \le CR^2 + \frac{1}{2\sqrt{\lambda_1}}(F(R+1) - F(R_0 + 1))$$

(observe that R_0 depends only on E and K). By iteration we get

$$F(R) - F(R_0 + 1) \le C \sum_{i=1}^{k} \frac{(R+i)^2}{2^{i-1}} + (2\sqrt{\lambda_1})^{-k} \left(F(R+k) - F(R_0 + 1) \right)$$

$$\le CR^2 + (2\sqrt{\lambda_1})^{-k} \left(F(R+k) - F(R_0 + 1) \right)$$

(3.3.12)

for every $k \in \mathbb{N}, k > 0$.

Now notice that for any $\delta \in (0, 1)$ we have

$$F(R+k) - F(R_0+1) \le e^{2\sqrt{\lambda_1}(R+k)(1-\delta)} \int_{E(R+k)\setminus E(R_0+1)} \xi^{2\delta} f_E^2 \,\mathrm{d}\mathfrak{m}$$

$$\stackrel{(3.3.7)}{\le} \frac{Ce^{2\sqrt{\lambda_1}(R+k)(1-\delta)}}{(1-\delta)^2},$$

thus picking δ so that $2\sqrt{\lambda_1}(1-\delta) < \log(2\sqrt{\lambda_1})$, by letting $k \to \infty$ in (3.3.12) we conclude that

$$F(R) \le CR^2. \tag{3.3.13}$$

To improve this, pick $t = \frac{R}{2}$ and $R_1 = R_0 + 1$ in (3.3.10) to obtain, after little manipulation, that

$$\frac{1}{R} \left(F(\frac{R}{2}) - F(R_0 + 1) \right) \le C \left(1 + \frac{1}{R^2} (F(R) - F(R_0 + 1)) \right) \stackrel{(3.3.13)}{\le} C$$

for every R large enough. This is (equivalent to) our claim (3.3.11).

Step 4. We claim that (3.3.2) holds. To see this, in (3.3.10) pick $t = \frac{2}{\sqrt{\lambda_1}}$, R + t in place of R and R - t in place of R_1 : with little manipulation we deduce that

$$\frac{1}{t^2} \left(F(R) - F(R-t) \right) \le \underbrace{\frac{C}{R} \left(F(R-t) - F(R_0) \right)}_{\le C \text{ by } (3.3.11)} + \frac{1}{4t^2} \underbrace{\left(F(R+t) - F(R-t) \right)}_{=(F(R+t) - F(R)) + (F(R) - F(R-t))}$$

and thus that $F(R) - F(R-t) \le C + \frac{1}{3}(F(R+t) - F(R))$. Iterating we get

$$F(R) - F(R-t) \le C \sum_{i=0}^{k-1} \frac{1}{3^i} + \frac{1}{3^k} (F(R+kt) - F(R+(k-1)t))$$

for every $k \in \mathbb{N}, k > 0$.

Recalling (3.3.11) and letting $k \to \infty$ we conclude that $F(R) - F(R-t) \leq C$ for all R large enough (recall that here $t = \frac{2}{\sqrt{\lambda_1}}$). The claim (3.3.2) easily follows.

Step 5. We claim that (3.3.3) and (3.3.4) hold.

Let $R > R_0 + 1$ and $\zeta \in \text{LIP}_{bs}(X)$ be defined as $\zeta := \hat{\zeta} \circ \mathsf{d}_p$, where

$$\hat{\zeta}(z) := \begin{cases} z - (R - 1) & \text{if } z \in [R - 1, R], \\ 1 & \text{if } z \in [R, R + 1], \\ R + 2 - z & \text{if } z \in [R + 1, R + 2], \\ 0 & \text{otherwise.} \end{cases}$$

Then using that f_E is harmonic we get the standard estimate

$$\int_{E} \zeta^{2} |\nabla f_{E}|^{2} d\mathfrak{m} = -2 \int_{E} \zeta f_{E} \langle \nabla \zeta, \nabla f_{E} \rangle d\mathfrak{m}$$
$$\leq \frac{1}{2} \int_{E} \zeta^{2} |\nabla f_{E}|^{2} d\mathfrak{m} + 2 \int_{E} |\nabla \zeta|^{2} f_{E}^{2} d\mathfrak{m}$$

that gives $\int_E \zeta^2 |\nabla f_E|^2 \, \mathrm{d}\mathfrak{m} \leq 4 \int_E |\nabla \zeta|^2 f_E^2 \, \mathrm{d}\mathfrak{m} \leq 4 \int_{E(R+2)\setminus E(R-1)} f_E^2 \, \mathrm{d}\mathfrak{m}$. Thus

$$\int_{E(R+1)\setminus E(R)} |\nabla f_E|^2 \,\mathrm{d}\mathfrak{m} \leq \int_E \zeta^2 |\nabla f_E|^2, \,\mathrm{d}\mathfrak{m}$$
$$\leq 4 \int_{E(R+2)\setminus E(R-1)} f_E^2 \,\mathrm{d}\mathfrak{m} \stackrel{(3.3.2)}{\leq} C e^{-2\sqrt{\lambda_1}R},$$

as desired. Then (3.3.4) is a direct consequence of (3.3.3).

These estimates allow us to deduce the following important dichotomy result. In proving it we shall use the main results shown in Section 3.2: the Gauss-Green formula (Theorem 3.2.22) and the coarea formula (Theorem 3.2.2).

Theorem 3.3.6. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ metric measure space and let E be an end of X. Assume that $\lambda_1 > 0$. Then exactly one of the following holds:

i) there exists a bounded non-constant harmonic function on E and E has exponential volume growth, more precisely

$$V_E(R) \ge C \exp\left(2\sqrt{\lambda_1}R\right)$$
 (3.3.14)

for every R large enough and some constant C > 0;

ii) every bounded harmonic function on E is constant and E has exponential volume decay, more precisely

$$V_E(\infty) - V_E(R) \le C \exp\left(-2\sqrt{\lambda_1}R\right)$$
 (3.3.15)

for every R large enough and some constant C > 0.

Proof.

Step 1. Assume that every bounded harmonic function on E is constant. Then this is the case for the function f_E given by Definition 3.3.4. We have then that f_E is equal to 1 a.e. on E, and by (3.3.2) follows that

$$V_E(R+1) - V_E(R) \le C \exp\left(-2\sqrt{\lambda_1}R\right).$$

Summing up we obtain

$$V_E(\infty) - V_E(R) \le C \sum_{i=0}^{+\infty} \exp\left(-2\sqrt{\lambda_1}(R+i)\right) \le C \exp\left(-2\sqrt{\lambda_1}R\right).$$

Step 2. Assume now that there exists a bounded non-constant harmonic function u on E. By translation and rescaling we can assume that $u \ge 0$ on X, $u \ge 1$ on K' and u < 1 somewhere in E (recall (3.3.1)). Then by the maximum principle and the construction of f_E it follows that f_E is not constant as well.

Now notice that by the spherical version of the Bishop-Gromov inequality we have that $B_R(p)$ has finite perimeter for any R > 0. It follows that for any $R > R_0$ the set $E^R := E \setminus E(R)$ has finite perimeter with $Per(E^R, \cdot)$ being the restriction of $Per(B_R(p), \cdot)$ to E.

We claim that there is c > 0 such that

$$\int |\nabla f_E| \,\mathrm{d}\operatorname{Per}(E^r, \cdot) \ge c \qquad \text{for every } r > R_0, \qquad (3.3.16)$$

where here, with a slight abuse of notation, we are denoting by $\nabla f_E \in L^0(TX, \operatorname{Per}(E^r(\cdot)))$ the trace of $\nabla f_E \in W^{1,2}_{C,loc}(E)$ in the sense of Definition 3.2.21.

Let us show how from (3.3.16) we can conclude. Denoting with P(r) the perimeter $Per(E^r, X)$, by (3.3.16) and the Cauchy-Schwarz inequality we immediately have $\frac{1}{P(r)} \leq \frac{1}{c} \int |\nabla f_E|^2 dPer(E^r, \cdot)$ and therefore

$$1 \leq \int_{R}^{R+1} P(r) \, \mathrm{d}r \int_{R}^{R+1} \frac{1}{P(r)} \, \mathrm{d}r$$

$$\stackrel{*}{\leq} \frac{1}{c} \left(V_{E}(R+1) - V_{E}(R) \right) \int_{E(R+1) \setminus E(R)} |\nabla f_{E}|^{2} \, \mathrm{d}\mathfrak{m} \quad (3.3.17)$$

(by (3.3.3))
$$\leq C \left(V_{E}(R+1) - V_{E}(R) \right) e^{-2R\sqrt{\lambda_{1}}},$$

where in the starred inequality we used the coarea formula to deduce

$$V_E(R+1) - V_E(R) = \mathfrak{m}(E(R+1) \setminus E(R))$$
$$= \int_{E(R+1) \setminus E(R)} |\nabla \mathsf{d}_p| \, \mathrm{d}\mathfrak{m} = \int_R^{R+1} P(r) \, \mathrm{d}r$$

and the fact that, if we call $\tilde{\nabla} f_E \in L^0_{\text{Cap}}(TX)$ the quasi-continuous representative of $\nabla f_E \in W^{1,2}_{C,loc}(E)$ (set to 0 outside E, say) and $\operatorname{tr}_r \nabla f_E$ the trace of ∇f_E in $L^0(TX, \operatorname{Per}(E^r, \cdot))$, then, by definition, $\operatorname{tr}_r \nabla f_E$ is the equivalence class of $\tilde{\nabla} f_E$ in in $L^0(TX, \operatorname{Per}(E^r, \cdot))$ and thus we have $|\operatorname{tr}_r \nabla f_E| = |\tilde{\nabla} f_E|$ $\operatorname{Per}(E^r, \cdot)$ -a.e.. This and the trivial identity $|\tilde{\nabla} f_E| = |\nabla f_E|$ m-a.e. justifies the computation above.

Since (3.3.17) is equivalent to the claim, we are left to prove (3.3.16). Since f_E is harmonic, putting $F(r) := \int \langle \nabla f_E, \nu_{E(r)} \rangle \,\mathrm{d}\operatorname{Per}(E^r, \cdot)$ for brevity, by the Gauss-Green formula we have that, for every $r_2 > r_1 > R_0$ it holds

$$F(r_2) - F(r_1) = \int \langle \nabla f_E, \nu_{E(r_2) \setminus E(r_1)} \rangle \,\mathrm{d}\operatorname{Per}(E(r_2) \setminus E(r_1), \cdot)$$
$$= -\int_{E(r_2) \setminus E(r_1)} \Delta f_E \,\mathrm{d}\mathfrak{m} = 0,$$

thus $r \mapsto F(r)$ is constant on $(R_0, +\infty)$ and since clearly

$$|F(r)| \leq \int |\nabla f_E| \,\mathrm{d}\operatorname{Per}(E^r, \cdot),$$

to get (3.3.16) it suffices to prove that $F(r) \neq 0$ for some $r > R_0$.

To see this, fix $R > R_0$ and notice that by the strong maximum principle and the fact that f_E is not constant, there must be a > 0 such that $\sup_{\partial E^{\bar{R}}} f_E \leq 1 - a$. In particular, taking into account the continuity of f_E and the Sobolev-to-Lipschitz property it easily follows that there is $b \in (0, \frac{a}{2})$ such that

$$\int_{E \cap B_{\bar{R}}(p) \cap \{1-2b < f_E < 1-b\}} |\nabla f_E|^2 \, \mathrm{d}\mathfrak{m} > 0.$$
(3.3.18)

Put $\rho := \varphi \circ f_E$ where

$$\varphi(z) := \begin{cases} 1, & \text{if } z \leq 1-2b, \\ 0, & \text{if } z \geq 1-b, \\ \text{affine and continuous on } [1-2b, 1-b]. \end{cases}$$

and notice that we have

$$\begin{split} \int_{E(\bar{R})} \operatorname{div}(\rho \nabla f_E) \, \mathrm{d}\mathfrak{m} &= \int_{E(\bar{R})} \langle \nabla \rho, \nabla f_E \rangle \, \mathrm{d}\mathfrak{m} \\ &= -\frac{1}{b} \int_{E \cap B_{\bar{R}}(p) \cap \{1-2b < f_E < 1-b\}} |\nabla f_E|^2 \, \mathrm{d}\mathfrak{m} \overset{(3.3.18)}{<} 0. \end{split}$$

We now claim that there is a finite perimeter set \tilde{E} such that $\tilde{E}\Delta E(\bar{R}) \subset f_E^{-1}((1-b,1))$. Assuming we have such \tilde{E} , we conclude with

$$0 > \int_{E(\bar{R})} \operatorname{div}(\rho \nabla f_E) \, \mathrm{d}\mathfrak{m} = \int_{\tilde{E}} \operatorname{div}(\rho \nabla f_E) \, \mathrm{d}\mathfrak{m}$$
$$= \int \rho \langle \nabla f_E, \nu_{\tilde{E}} \rangle \, \mathrm{d}\operatorname{Per}(\tilde{E}, \cdot) = -F(\bar{R}),$$

where in the last step we used the fact that $\rho \equiv 1$ on $\partial E^{\bar{R}} = \partial \tilde{E} \cap f_E^{-1}((0, 1 - b])$ and $\rho \equiv 0$ on $\partial \tilde{E} \setminus f_E^{-1}((0, 1 - b])$. It thus remains to show the existence of such \tilde{E} . To see this, recall that balls have finite perimeter (by the spherical version of the Bishop-Gromov inequality (1.5.11)) and that ∂E is compact. Then using the continuity of f_E we can find r > 0 such that for any $x \in \partial E$ we have $B_r(x) \subset \{f_E \geq 1 - b\}$ and by compactness a finite number of points x_1, \ldots, x_n such that $\partial E \subset \bigcup_i B_r(x_i)$. It is then clear that the set $\tilde{E} := E \setminus \bigcup_i B_r(x_i)$ does the job.

As a direct consequence of the above result we obtain the following corollaries.

Corollary 3.3.7. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ space with $\lambda_1 > 0$ and with at least two ends with infinite volume.

Then there exists a bounded non-constant harmonic function on X.

Proof. Let E_1, E_2 be the two given ends with infinite volume and consider the associated functions f_{E_1}, f_{E_2} as in Definition 3.3.4 (possibly enlarging R_0 we can take $K' := B_{R_0}(p)$), which by Theorem 3.3.6 are bounded non-constant and harmonic functions. Recall that $f_{E_i} \equiv 1$ on ∂E_i and ess inf $f_{E_i} = 0$ for i = 1, 2.

Then fix $R > R_0 + 1$ and let $C_i := \{x \in B_R(p) : \mathsf{d}(x, E_i \cap \partial B_{R+1}(p)) \ge 1\}$. Then a simple variant of Theorem 3.3.3 gives the existence of a function $f_R \in (W_0^{1,2} \cap C)(B_{R+2}(p))$ with values in [0,1] that is harmonic in $B_{R+2}(p) \setminus (C_1 \cup C_2) \supset B_R(p)$, equal to 1 on C_1 and equal to 0 on C_2 .

We claim that on $B_R \cap E_2$ we have $f_R \leq f_{E_2}$. To see this notice that the boundary $\partial(B_R \cap E_2)$ is contained in the disjoint union of ∂E_2 and C_2 and by construction we have $f_{E_2} = 1$ on ∂E_2 and $f_R = 0$ on C_2 . Since both functions have values in [0, 1], are harmonic in $B_R \cap E_2$ and continuous up to the boundary of such set, by the maximum principle our claim follows.

Analogously we can prove that $f_R \ge 1 - f_{E_1}$ on $B_R \cap E_1$.

We now want to send $R \uparrow +\infty$ and find, possibly after passing to a subsequence, a limit harmonic function on the whole X. This is possibile thanks to the Lipschitz estimates (3.1.3), that grants that for some C(R) the

Lipschitz constant of $f_{R'}$ on $B_{R/2}(p)$ is bounded from above by C(R) for any R' > R. By Arzelá-Ascoli's theorem, this suffices to find a limit function f and it is then easy to see that this function is harmonic and bounded (in fact with values in [0, 1]).

Also, a direct consequence of the construction and of the previous claims is that $f \leq f_{E_2}$ on E_2 and $f \geq 1 - f_{E_1}$ on E_1 . If we knew that $\inf_{E_i} f_{E_i} = 0$ for i = 1, 2 the conclusion would directly follow, as $\inf f \leq \inf_{E_2} f_{E_2} = 0$ and $\sup f \geq \sup_{E_1} 1 - f_{E_1} = 1 - \inf_{E_1} f_{E_1} = 1$, proving that f is not constant.

Thus it remains to show that $\inf_{E_1} f_{E_1} = 0$ (the argument for E_2, f_{E_2} being analogous). Say not. Then by construction $a := \inf_{E_1} f_{E_1} > 0$. Define the new function $\tilde{f}_{E_1} := 1 - \frac{1 - f_{E_1}}{1 - a}$ and notice that it is still harmonic and continuous on E_1 with boundary value equal to 1 and that it is still positive. Thus by the maximum principle the function \tilde{f}_{E_1} bounds from above each of the harmonic functions defined on $B_R(p) \cap E_1$ that are used in the definition of f_{E_1} . It follows by the definition of \tilde{f}_{E_1} has this ensures that $\tilde{f}_{E_1} < f_{E_1}$ on E_1 .

A similar argument allows us to find a non-constant positive harmonic function on X in the case of one end with infinite volume and one with finite volume.

Corollary 3.3.8. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ space with $\lambda_1 > 0$ and with at least one end with infinite volume and one end with finite volume.

Then there exists a non-constant positive harmonic function on X.

Proof. Let E_1 be an end with infinite volume and E_2 be an end with finite volume both with respect to a compact $K \subset X$, and fix $p \in K$. Taking $R_0 > 0$ large enough we can assume, as before, $K' := \overline{B_{R_0}}(p) \supset \overline{K^1}$.

Let f_{E_1} and f_R be defined as in the proof of Corollary 3.3.7 (we note that the definition of f_R does not depend on the volume of the ends, so the same argument holds in this case too): f_{E_1} is defined as in Definition 3.3.4, and for R large enough $f_R \in (W_0^{1,2} \cap C)(\overline{B_R}(p))$ is harmonic in $B_R(p)$ identically equal to 1 on $\partial E_2(R)$ and identically equal to 0 on $\partial E_1(R)$ and on every other eventual end $\partial E_i(R)$ with i > 2.

To be precise, we should modify $B_R(p)$ near its boundary as in the proof of Corollary 3.3.7 to achieve the regularity needed to apply Theorem 3.3.3. We omit these technical details here referring instead to the proof of Corollary 3.3.7.

We fix $\varepsilon > 0$ and take $(\partial E_2)^{\varepsilon} := \{p \in X : \mathsf{d}(p, \partial E_2) \le \varepsilon\}$. Thanks to the strong maximum principle, the compactness of $(\partial E_2)^{\varepsilon}$ and the continuity of f_R we have that $c_R := \max_{(\partial E_2)^{\varepsilon}} f_R > 0$ for every R large enough. Moreover, by

(3.1.3), the functions $\log(f_R/c_R)$ are locally equi-Lipschitz, and since they are equibounded on $(\partial E_2)^{\varepsilon}$ (by the definition of c_R) we conclude by Arzelà-Ascoli's theorem that there exists a subsequence of $(f_R/c_R)_R$ that converges uniformely on compact sets to a function $u: X \to \mathbb{R}$.

The function u is clearly positive and harmonic. We observe that it is also non-constant. Indeed, as in the proof of Corollary 3.3.7 it is easy to see that $u \leq f_{E_1}$ on E_1 , then $\inf_X u = 0$, moreover, by the definition of c_R we conclude that $u \geq 1$ on $(\partial E)^{\varepsilon}$.

We also note that on every other end E_i with i > 2, thanks to the maximum principle, $f_R < c_R$, then $u \le 1$ on every end E_i with i > 2 (if the space has more than two ends).

Corollaries 3.3.7 and 3.3.8 should be coupled with the following simple and general result, stating that if $\lambda_1 > 0$ (an assumption that is present in Theorems 0.0.7 and 0.0.8), then the space has at least one end of infinite volume.

Proposition 3.3.9. Let (X, d, \mathfrak{m}) be a metric measure space with $\lambda_1 > 0$. Then X has at least one end with infinite volume.

Proof. By the definition of λ_1 it follows immediately that if $\mathfrak{m}(X) < \infty$ then $\lambda_1 = 0$, this means that $\mathfrak{m}(X) = \infty$.

Assume by contradiction that X has no ends with infinite volume. Then there exists a compact K which has positive volume and cuts the space in ends with finite volume. Let $\{E_i\}_{i\in\mathbb{N}}$ be the set of the ends with respect to K. Since $\mathfrak{m}(E_i) < \infty$ for every $i \in \mathbb{N}$ then for every $\varepsilon > 0$ there exist a couple of radii $r_{i,\varepsilon}$, $R_{i,\varepsilon} > r_{i,\varepsilon} + 2$ such that $V_{E_i}(R_{i,\varepsilon}) - V_{E_i}(r_{i,\varepsilon}) \leq \frac{\varepsilon}{2^i}$. Taking now the function

$$f_{\varepsilon}(x) := \begin{cases} 1 & \text{for } x \in \mathcal{K} \cup \left(\bigcup(E_i(r_{i,\varepsilon}))\right), \\ \frac{\mathsf{d}(x,p) - r_{i,\varepsilon}}{R_{i,\varepsilon} - r_{i,\varepsilon}} & \text{for } x \in E_i(R_{i,\varepsilon}) \setminus E_i(r_{i,\varepsilon}), \\ 0 & \text{for } x \in E_i \setminus E_i(R_{i,\varepsilon}), \end{cases}$$

we conclude that

$$\frac{\int_{\mathbf{X}} |\nabla f_{\varepsilon}|^2 \, \mathrm{d} \mathfrak{m}}{\int_{\mathbf{X}} |f_{\varepsilon}|^2 \, \mathrm{d} \mathfrak{m}} \leq \frac{\varepsilon}{\mathfrak{m}(\mathbf{K})} \xrightarrow[\varepsilon \to 0]{} 0.$$

г			

3.4 Rigidity theorem for spaces with positive spectrum

We focus now on Theorem 0.0.7, which states that if a space with strictly positive λ_1 has two ends with infinite volume then it splits (see Definitions 3.3.2 and 3.3.1 for the definitions of λ_1 and ends in metric measure spaces). We recall here its precise statement.

Theorem 3.4.1. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(-(N-1), N)$ space with $N \ge 3$ and $\operatorname{supp}(\mathfrak{m}) = X$, and assume that the first eigenvalue of the Laplacian λ_1 is $\ge N-2$. Then one of the following holds:

- i) X has only one end with infinite volume;
- ii) X is isomorphic as metric measure space to a warped product space $\mathbb{R} \times_w X'$, where X' is a compact $\mathsf{RCD}(-(N-1), N)$ space and the warping functions are

 $w_{\mathsf{d}}(t) := \cosh(t)$ and $w_{\mathfrak{m}}(t) := \cosh^{N-1}(t).$

Moreover, in this case $\lambda_1 = N - 2$.

Given the statement of Theorem 3.4.1 and remembering the results of Section 3.3, let us fix the following notations and assumptions that we will use along all the proof of this result.

Assumption 3.4.2. (X, d, \mathfrak{m}) is an $\mathsf{RCD}(-(N-1), N)$ space with $N \ge 3$ and $\lambda_1 \ge N-2$. We also assume that $\mathrm{supp}(\mathfrak{m}) = X$ and that X has two ends with infinite volume.

Finally, we shall denote by u a fixed bounded and non-constant harmonic function on X. The existence of such u is granted by Corollary 3.3.7.

3.4.1 Properties of the bounded harmonic function

We start with the following simple regularity statement.

Lemma 3.4.3. With the same assumptions and notation as in Assumption 3.4.2 the following holds.

The function u is in $\text{Test}_{loc}(X)$ and globally Lipschitz.

Proof. By the Cheng-Yau gradient estimate (3.1.3) and the fact that u is bounded it follows that it is globally Lipschitz. This information together with the fact that $\Delta u \equiv 0$ suffices to ensure that $u \in \text{Test}_{loc}(X)$.

The rigidity result we are going to prove in this section is ultimately a consequence, as customary, of the equality in the improved Bochner inequality (see inequality (3.1.4)). Before coming to that it is useful to recall the following result, that in the RCD setting has been proved in [GV]. We provide anyway the complete proof because we shall be interested in the equality case, that was not explicitly studied in [GV].

Lemma 3.4.4 (Generalized refined Kato inequality). Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ space with $K \in \mathbb{R}$ and $N \in [1, +\infty)$. Then for any $u \in H^{2,2}_{loc}(X)$ it holds

$$\frac{t + \dim(\mathbf{X})}{t + \dim(\mathbf{X}) - 1} |\mathbf{d}| \mathbf{d}u||^2 \le |\operatorname{Hess}(u)|_{\mathsf{HS}}^2 + \frac{(\operatorname{tr}\operatorname{Hess}(u))^2}{t} \quad \mathfrak{m}\text{-}a.e. \quad (3.4.1)$$

for every t > 0, where dim(X) $\in \mathbb{N} \cap [1, N]$ is the (constant) dimension of $L^0(TX)$ (recall Theorem 3.1.12). Equality holds \mathfrak{m} -a.e. for a given t if and only if

$$\operatorname{Hess}(u) = \alpha \big(\operatorname{Id} - (t + \dim(X)) e_1 \otimes e_1 \big), \qquad (3.4.2)$$

for some function $\alpha \in L^2_{loc}(\mathbf{X})$, where $\mathbf{e}_1 \in L^0(T\mathbf{X})$ is a pointwise unitary vector that is equal to $\frac{\nabla u}{|\nabla u|}$ on $|\nabla u| > 0$.

Proof. We indicate with n the dimension dim(X). Let A be a symmetric $n \times n$ real matrix. We claim that for every t > 0 and every $v \in \mathbb{R}^n$ we have:

$$\frac{t+n}{t+n-1}|A \cdot v|^2 \le |v|^2 |A|_{\mathsf{HS}}^2 + \frac{|v|^2 (\operatorname{tr} A)^2}{t}.$$
(3.4.3)

Indeed, by the spectral theorem there is an orthonormal base e_1, \ldots, e_n of \mathbb{R}^n so that A is diagonal with respect to such base, with diagonal entries $\alpha_1, \ldots, \alpha_n$. We can also assume that $|\alpha_1| \ge |\alpha_i|$ for $i = 2, \ldots, n$. Then, applying twice the Cauchy-Schwarz inequality we obtain

$$\begin{aligned} |v|^{2} \left(\frac{(\alpha_{1} + \alpha_{2} + \dots + \alpha_{n})^{2}}{t} + \alpha_{1}^{2} + \alpha_{2}^{2} + \dots + \alpha_{n}^{2} \right) \\ &\stackrel{(1)}{\geq} |v|^{2} \left(\frac{(\alpha_{1} + \alpha_{2} + \dots + \alpha_{n})^{2}}{t} + \frac{(\alpha_{2} + \alpha_{3} + \dots + \alpha_{n})^{2}}{n - 1} + \alpha_{1}^{2} \right) \\ &\stackrel{(2)}{\geq} |v|^{2} \left(\frac{\alpha_{1}^{2}}{t + n - 1} + \alpha_{1}^{2} \right) \\ &\stackrel{(3)}{\geq} \frac{t + n}{t + n - 1} |A \cdot v|^{2}, \end{aligned}$$

which is (3.4.3). Equality holds in (3.4.3) if and only if the inequalities above are all equality. Equality in (1) holds if and only if $\alpha_2 = \cdots = \alpha_n =: \alpha$,

equality in (2) if and only if $\alpha = \alpha_1(1 - (t + n))$. Finally, equality in (3) holds if and only if v is a multiple of e_1 . In other words, equality in (3.4.3) holds if and only if

$$A = \alpha (\mathrm{Id} - (t+n)\mathbf{e}_1 \otimes \mathbf{e}_1).$$

We come to (3.4.1) and its equality case and start noticing that $d|du| = \chi_{\{|\nabla u|>0\}}$ Hess $u(\frac{\nabla u}{|\nabla u|})$ (for $u \in \text{Test}_{loc}(X)$ this follows from Proposition 1.6.13 and the chain rule - see also [DGP, Lemma 2.5], then the general case comes by approximation). Now the conclusion follows picking a suitable pointwise orthonormal base $e_1, \ldots, e_{\dim(X)} \in L^0(TX)$ of $L^0(TX)$ that diagonalizes Hess u.

Proposition 3.4.5. With the same assumptions and notation as in Assumption 3.4.2 the following hold:

- *i*) $\lambda_1 = N 2;$
- *ii)* |du| *is locally Lipschitz (i.e. it has a locally Lipschitz representative) and* strictly positive;
- iii) Putting $e_1 := \frac{\nabla u}{|\nabla u|}$ (this is well defined by item (ii)) we have

$$\operatorname{Hess}(u) = \alpha (\operatorname{Id} - Ne_1 \otimes e_1), \qquad (3.4.4)$$

for some $\alpha \in L^2_{loc}(\mathbf{X})$;

iv) we have $|du|^{\frac{N-2}{N-1}} \in D(\Delta_{loc})$ with

$$\Delta(|\mathrm{d}u|^{\frac{N-2}{N-1}}) = -(N-2)|\mathrm{d}u|^{\frac{N-2}{N-1}}; \qquad (3.4.5)$$

v) u is an open map, i.e. $u(U) \subset \mathbb{R}$ is open for any $U \subset X$ open.

Proof. The required rigidity will follow by closely inspecting the proof of [GV, Theorem 3.4]. Start noticing that since we know from Lemma 3.4.3 above that $u \in \text{Test}_{loc}(X)$, we can apply the improved Bochner inequality 3.1.4 (say $N > \dim(X)$; the case $N = \dim(X)$ follows along similar lines recalling that in this case tr Hess $u = \Delta u = 0$ and that the last addend in the Bochner inequality below is 0 - see [Han]) recalling that $\Delta u = 0$ to get

$$\mathbf{\Delta}_{\frac{|\mathrm{d}u|^2}{2}}^{\frac{|\mathrm{d}u|^2}{2}} \ge \left(|\operatorname{Hess} u|^2_{\mathsf{HS}} - (N-1)|\mathrm{d}u|^2 + \frac{(\operatorname{tr} \operatorname{Hess} u)^2}{N - \dim(\mathrm{X})}\right)\mathfrak{m}.$$

Kato's inequality as in Lemma 3.4.1 with $t = N - \dim(X)$ can be written as

$$|\operatorname{Hess} u|_{\mathsf{HS}}^2 = \frac{N}{N-1} |\mathrm{d}|\mathrm{d} u||^2 - \frac{(\operatorname{tr} \operatorname{Hess} u)^2}{N-\dim(\mathbf{X})} + F, \qquad \text{for some } F \ge 0 \ \mathfrak{m}\text{-a.e.}$$

and the equality case reads as

$$F = 0 \mathfrak{m}$$
-a.e. \Rightarrow identity (3.4.4) holds. (3.4.6)

We thus have

$$\Delta \frac{|\mathrm{d}u|^2}{2} \ge \left(\frac{N}{N-1} |\mathrm{d}|\mathrm{d}u||^2 - (N-1)|\mathrm{d}u|^2 + F\right)\mathfrak{m}$$
(3.4.7)

Now let us fix $\beta, \varepsilon > 0$ and put $\varphi(z) := \varphi_{\beta,\varepsilon}(z) := (z + \varepsilon)^{\beta}$. Then (3.4.7) and basic calculus rules (see also the proof of [GV, Theorem 3.4]) give

$$\begin{aligned} \mathbf{\Delta}(\varphi(|\mathrm{d}u|^2)) &\geq \left(\beta(|\mathrm{d}u|^2 + \varepsilon)^{\beta - 1}(2F - 2(N - 1)|\mathrm{d}u|^2) \\ &+ 2\beta|\mathrm{d}|\mathrm{d}u||^2 \left(\frac{N}{N - 1} + \frac{2(\beta - 1)|\mathrm{d}u|^2}{|\mathrm{d}u|^2 + \varepsilon}\right)\right) \mathfrak{m}.\end{aligned}$$

Now pick $\beta := \frac{N-2}{2(N-1)}$ and let $\varepsilon \downarrow 0$: the rightmost addend goes to 0 and putting for brevity $f := |\nabla u|^{\frac{N-2}{N-1}}$ (and summing $\lambda_1 f \mathfrak{m}$ to both sides) we are left with

$$\Delta f + \lambda_1 f \mathfrak{m} \ge \left((\lambda_1 - (N-2))f + \frac{N-2}{2(N-1)}F |\mathrm{d}u|^{-\frac{N}{N-1}} \right) \mathfrak{m} \ge 0 \qquad (3.4.8)$$

(notice that the passage to the limit is justified also by the fact that $f \in W_{loc}^{1,2}(\mathbf{X})$, as established in [GV, Theorem 3.4] - the proof of this follows from the computations we are repeating here).

To deduce the desired rigidity from the above, start observing that for any $\varphi \in \text{LIP}_{bs}(\mathbf{X})$ we have

$$\int \varphi^2 f \,\mathrm{d} \mathbf{\Delta} f = \int -\varphi^2 |\mathrm{d} f|^2 - 2f\varphi \langle \mathrm{d} \varphi, \mathrm{d} f \rangle \,\mathrm{d} \mathfrak{m} = \int -|\mathrm{d}(\varphi f)|^2 + f^2 |\mathrm{d} \varphi|^2 \,\mathrm{d} \mathfrak{m},$$

and thus using that $\lambda_1 \int f^2 \varphi^2 \, \mathrm{d}\mathfrak{m} \leq \int |\mathrm{d}(\varphi f)|^2 \, \mathrm{d}\mathfrak{m}$ we get

$$\int \varphi^2 f \,\mathrm{d}(\mathbf{\Delta} f + \lambda_1 f \mathfrak{m}) \le \int f^2 |\mathrm{d}\varphi|^2 \,\mathrm{d}\mathfrak{m}. \tag{3.4.9}$$

Notice that (3.4.8) and the assumption $\lambda_1 \geq N-2$ tell in particular that $\Delta f + \lambda_1 f \mathfrak{m} \geq 0$, then fix R > 0, let $\varphi := (1 - R^{-1} \mathsf{d}(\cdot, B_R(p)))^+$ in the above then let $R \uparrow \infty$ to get

$$\int f \,\mathrm{d}(\mathbf{\Delta} f + \lambda_1 f \mathfrak{m}) \leq \liminf_{R \to +\infty} \frac{1}{R^2} \int_{B_{2R}(p) \setminus B_R(p)} f^2 \,\mathrm{d}\mathfrak{m}.$$
(3.4.10)

Let us prove that the right hand side is zero. Putting $A_R := B_{2R}(p) \setminus B_R(p)$, from Holder's inequality we get

$$\begin{aligned} \int_{A_R} f^2 \, \mathrm{d}\mathfrak{m} &\leq \left(\int_{A_R} |\mathrm{d}u|^2 e^{2\sqrt{\lambda_1} \mathsf{d}_p} \, \mathrm{d}\mathfrak{m} \right)^{\frac{N-2}{N-1}} \\ &\quad \cdot \left(\int_{A_R} e^{-2(N-2)\sqrt{\lambda_1} \mathsf{d}_p} \, \mathrm{d}\mathfrak{m} \right)^{\frac{1}{N-1}} \\ (\mathrm{by} \ (3.3.4)) &\quad \leq CR^{\frac{N-2}{N-1}} \Big(\int_R^{2R} e^{-2(N-2)\sqrt{\lambda_1} \mathsf{d}_p} s(\mathsf{d}_p) \, \mathrm{d}\mathfrak{m} \Big)^{\frac{1}{N-1}} \\ (\mathrm{by} \ (1.5.11)) &\quad \leq CR^{\frac{N-2}{N-1}} \Big(\int_R^{2R} e^{(-2(N-2)\sqrt{\lambda_1}+N-1)\mathsf{d}_p} \, \mathrm{d}\mathfrak{m} \Big)^{\frac{1}{N-1}} \leq CR, \end{aligned}$$

where in the last inequality we used the fact that $-2(N-2)\sqrt{\lambda_1}+N-1 \leq 0$, that in turn follows from $\lambda_1 \geq N-2$ and $N \geq 3$.

Therefore from (3.4.10) we see that $\int f d(\Delta f + \lambda_1 f \mathfrak{m}) \leq 0$ and thus from (3.4.8) we conclude that

$$(\lambda_1 - (N-2)) \int f^2 \,\mathrm{d}\mathfrak{m} + \int |\mathrm{d}u|^{-\frac{2}{N-1}} F \,\mathrm{d}\mathfrak{m} \le 0.$$

Since $\lambda_1 \ge N-2$ and f is not identically 0 (as u is not constant), we deduce that $\lambda_1 = N-2$, i.e. item (i) holds. Also, since $|du|^{-\frac{2}{N-1}} \ge \text{LIP}(u)^{-\frac{2}{N-1}} > 0$, we see that F = 0 m-a.e., and thus by (3.4.6) item (iii) holds as well.

Now notice that we now know that

$$\Delta f = -(N-2)f\mathfrak{m} \tag{3.4.11}$$

and by the compatibility of the concepts of measure-valued and L^2 -valued Laplacian (see Proposition 3.1.2), item (iv) follows. Moreover, we also deduce that $\Delta f \leq 0$, hence f is superharmonic on any bounded open subset of X. Since X is locally doubling and supports a local weak Poincaré inequality, point (ii) follows from the weak Harnack inequality (see e.g. [BB, Chapter 8]) recalling, once again, that f is not identically 0. Even more, knowing (3.4.11) and the Bochner inequality it is not hard to see that

$$\mathbf{\Delta}\frac{|\mathrm{d}f|^2}{2} \ge \langle \mathrm{d}f, \mathrm{d}\Delta f \rangle - (N-1)|\mathrm{d}f|^2 = -(2N-3)|\mathrm{d}f|^2,$$

thus the same arguments just expressed tell that |df| is locally bounded from above, i.e. that f is locally Lipschitz (by the Sobolev-to-Lipschitz property). Hence the same holds for $|du| = f^{\frac{N-1}{N-2}}$.

It remains to prove that u is open. Fix $\bar{x} \in X$, let $\xi : X \to [0, 1]$ be a Lipschitz cut-off function with compact support identically 1 on $B_2(\bar{x})$.

102 Chapter 3. Applications to spaces with positive spectrum

Let (Fl_t) be the Regular Lagrangian Flow of $\xi \nabla u$, that is easily seen to exists as $|\xi \nabla u| \in L^{\infty}$, $\operatorname{div}(\xi \nabla u) = \langle \nabla \xi, \nabla u \rangle \in L^{\infty}$ and $\xi \nabla u \in W_C^{1,2}(TX)$ by Proposition 1.6.16. By the defining property (1.8.3) and the finite speed of propagation that follows from (1.8.5), we see that there is T > 0 such that $\partial_t u(\mathsf{Fl}_t(x)) = |\mathrm{d}u|^2(\mathsf{Fl}_t(x))$ holds for \mathfrak{m} -a.e. $x \in B_1(\bar{x})$ and $t \in [-T, T]$. For these x, t, recalling also item (*ii*), we see that $\partial_t u(\mathsf{Fl}_t(x)) \geq c$ for c := $\inf_{B_2(\bar{x})} |\mathrm{d}u| r > 0$. Fix x for which this holds for a.e. $t \in [-T, T]$ and use the continuity of u to deduce that the image under u of $\{\mathsf{Fl}_t(x) : t \in [-T, T]\}$ contains [u(x) - Tc, u(x) + Tc]. Picking x sufficiently close to \bar{x} we conclude that $u(B_1(\bar{x}))$ contains a neighbourhood of u(x) and repeating the argument with $B_r(\bar{x}), r \ll 1$, in place of $B_1(\bar{x})$ we conclude.

3.4.2 |du| as a function of u

In this section we prove that the minimal weak upper gradient of u is of the form $\varphi \circ u$ for a suitable smooth function φ . Notice that from (3.4.4) it follows that

$$d|du| = \operatorname{Hess} u(\frac{\nabla u}{|du|}) = \alpha(1-N)\frac{du}{|du|},$$

i.e. d|du| = hdu for some function h. In the smooth category, this suffices to conclude that, locally, |du| is a function of u, a natural line of thought being: let U be a small open set such that level sets of u are smooth-path-connected in U (recall that |du| > 0), then pick $x, y \in U$ with u(x) = u(y) and find a smooth curve γ joining them, with values in U and with $t \mapsto u(\gamma_t)$ constant. Then $0 = \partial_t u(\gamma_t) = du(\gamma'_t)$ and thus $\partial_t(|du|(\gamma_t)) = d|du|(\gamma'_t) = h(\gamma_t)du(\gamma'_t) = 0$, proving that |du| is also constant along γ and thus that, on U, the value of |du| depends solely on that of u.

We are going to prove an analogous statement in our setting: roughly said, the underlying idea is the same just exposed, but the technicalities are much more involved.

Proposition 3.4.6. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ space and fix two functions $f, g \in \mathrm{LIP}_{loc}(X)$ such that $\mathrm{d}g = \mathrm{hd}f$ for some $h : X \to \mathbb{R}$.

Also, assume that $f \in D(\Delta_{loc}) \cap W^{2,2}_{loc}(\mathbf{X})$ is with $\frac{1}{|\mathrm{d}f|}, |\Delta f|, |\mathrm{Hess}\, f|_{\mathsf{HS}} \in L^{\infty}_{loc}(\mathbf{X}).$

Then for every $x \in X$ there are a neighbourhood U and a Lipschitz function $\varphi : \mathbb{R} \to \mathbb{R}$ such that $g = \varphi \circ f$ on U.

Proof. Fix $\bar{x} \in X$ and let $\eta : X \to [0, 1]$ be Lipschitz, with bounded support and identically 1 on $B_3(\bar{x})$. Then by direct computation we see that for the vector field $v:=\eta \frac{\nabla f}{|\nabla f|^2}$ we have

$$\begin{aligned} \operatorname{div}(v) &= \eta \frac{\Delta f}{|\nabla f|^2} - 2\eta \frac{\langle \nabla |\nabla f|, \nabla f \rangle}{|\nabla f|^3} + \frac{\langle \nabla \eta, \nabla f \rangle}{|\nabla f|^2} \\ \nabla v &= \eta \frac{\operatorname{Hess}(f)}{|\nabla f|^2} - 2\eta \frac{\nabla |\nabla f| \otimes \nabla f}{|\nabla f|^3} + \frac{\nabla \eta \otimes \nabla f}{|\nabla f|^2} \end{aligned}$$

so that our assumptions on f and η grant that v has bounded divergence and bounded covariant derivative. It follows from Theorem 1.8.7 that the Regular Lagrangian Flow FI : $\mathbb{R} \times X \to X$ of v is Lipschitz in space and time. By definition of Regular Lagrangian Flow and of v the equation $\partial_t(f(\mathsf{Fl}_t(x))) =$ $\eta(\mathsf{Fl}_t(x))$ holds for **m**-a.e. x and a.e. t, but thanks to the continuity of η , FI and f it is easy to see that in fact for any $x \in X$ the map $t \mapsto f(\mathsf{Fl}_t(x))$ is C^1 with derivative equal to $\eta(\mathsf{Fl}_t(x))$.

Now consider the map $H : X \times \mathbb{R} \to X$ defined as $H(x,t,s) := \mathsf{Fl}_{t+s-f(x)}(x)$. Then this is clearly Lipschitz. Also, let $B \subset X$ be bounded and $I \subset \mathbb{R}$ be a bounded interval. We claim that for some C > 0 we have

$$H_*(\mathfrak{m}_{|B} \times \delta_t \times \mathcal{L}^1_{|I}) \le C\mathfrak{m} \quad \text{for every } t \in I.$$
(3.4.12)

Indeed, since f is Lipschitz there is T > 0 such that $|t + s - f(x)| \leq T$ holds for any $(x, t, s) \in B \times I \times I$. It follows that for any $\varphi \in C_b(X)$ non-negative it holds

$$\int \varphi \, \mathrm{d}H_*(\mathfrak{m}_{|B} \times \delta_t \times \mathcal{L}^1_{|I}) = \int_B \int_I \varphi(\mathsf{Fl}_{t+s-f(x)}) \, \mathrm{d}s \, \mathrm{d}\mathfrak{m}(x)$$
$$\leq \int_B \int_{-T}^T \varphi(\mathsf{Fl}_r) \, \mathrm{d}r \, \mathrm{d}\mathfrak{m}(x) \leq 2T \mathsf{Comp}(\mathsf{Fl}) \int \varphi \, \mathrm{d}\mathfrak{m},$$

thus proving our claim (3.4.12). Notice also that since $|v| \in L^{\infty}$, the estimate (1.8.5) trivially ensures that

 $\exists \bar{t} > 0 \text{ so that for any } x \in B_2(\bar{x}) \text{ we have } \mathsf{Fl}_t(x) \in B_3(\bar{x}) \text{ for any } t \in [-\bar{t}, \bar{t}].$ (3.4.13)

Put
$$\bar{r} := \min\{\frac{1}{2}, \frac{t}{8}(\operatorname{LIP}(f_{|B_3(\bar{x})}))^{-1}\}$$
. We claim that
there is $\varphi : \mathbb{R} \to \mathbb{R}$ such that $g = \varphi \circ f$ on $B_{\bar{r}}(\bar{x})$ (3.4.14)

and argue by contradiction. If not, there are $x_0, x_1 \in B_{\bar{r}}(\bar{x})$ with $f(x_0) = f(x_1)$ and $g(x_0) < g(x_1)$. By continuity we can find neighbourhoods $U_0, U_1 \subset B_{\bar{r}}(\bar{x})$ of x_0, x_1 respectively such that $\inf_{U_1} g > \sup_{U_0} g$. Then we can find other neighbourhoods $V_i \subset U_i$ of $x_i, i = 0, 1$, and $\bar{s} \in (0, \frac{\bar{t}}{2})$ such that $\operatorname{Fl}_s(x) \in U_i$ for every $x \in V_i, s \in [-\bar{s}, \bar{s}], i = 0, 1$.

Let $\mu_i := \mathfrak{m}(V_i)^{-1}\mathfrak{m}_{|_{V_i}}$, i = 0, 1 and let π be the only (by [GRS]) optimal geodesic plan joining μ_0 to μ_1 . Then [R12a] ensures that π is a test plan and clearly it is concentrated on geodesics taking values in $B_{2\bar{r}}(\bar{x}) \subset B_1(\bar{x})$. Also, let $G : C([0,1], X) \times \mathbb{R} \to C([0,1], X)$ be given by $G(\gamma, s)_t := H(\gamma_t, f(x_0), s)$ and put $\hat{\pi} := G_*(\pi \times (\frac{1}{2\bar{s}}\mathcal{L}^1_{|_{[-\bar{s},\bar{s}]}}))$. From the fact that H is Lipschitz and (3.4.12) it directly follows that $\hat{\pi}$ is a test plan as well and the construction also ensures that $(e_i)_*\hat{\pi}$ is concentrated on V_i , i = 0, 1.

We make now the intermediate claim:

$$\hat{\pi}$$
 is concentrated on curves along which f is constant. (3.4.15)

To see this, notice that $\boldsymbol{\pi}$ is concentrated on geodesics having endpoints in $B_{\bar{r}}(\bar{x})$ and thus that it is sufficient to prove that for any such geodesic γ , any $t \in [0, 1]$ and $s \in [-\bar{s}, \bar{s}]$ we have

$$f(\mathsf{Fl}_{f(x_0)+s-f(\gamma_0)}(\gamma_t)) = f(x_0) + s.$$

Since $r \mapsto f(\mathsf{Fl}_r(\gamma_t))$ is C^1 with derivative $\eta(\mathsf{Fl}_r(\gamma_t))$, the claim follows if we show that $\mathsf{Fl}_r(\gamma_t) \in B_3(\bar{x}) \subset \{\eta = 1\}$ for any $|r| \leq |f(x_0) + s - f(\gamma_0)|$. To see this, notice that γ takes values in $B_{2\bar{r}}(\bar{x}) \subset B_2(\bar{x})$ and thus $|f(\gamma_0) + s - f(\gamma_t)| \leq \bar{s} + \operatorname{LIP}(f_{|B_3(\bar{x})})\mathsf{d}(\gamma_0, \gamma_t) \leq \bar{t}$ for any $t \in [0, 1]$ and $s \in [-\bar{s}, \bar{s}]$. The claim then follows from (3.4.13).

From (3.4.15) and the definition of speed of a test plan (see Theorem 1.4.3) we have that

$$0 = f \circ \mathbf{e}_s - f \circ \mathbf{e}_t = \int_t^s [\mathbf{e}_r^* \mathrm{d}f](\hat{\boldsymbol{\pi}}_r') \,\mathrm{d}r \qquad \hat{\boldsymbol{\pi}}\text{-a.e}$$

for every $t, s \in [0, 1]$, t < s. By Fubini's theorem this implies that for a.e. $t \in [0, 1]$ the identity $[e_t^* df](\hat{\pi}'_t) = 0$ holds $\hat{\pi}$ -a.e.. Now we can use our assumption dg = hdf (noticing that $|h| \in L^{\infty}_{loc}(\mathbf{X})$ as a consequence of $|dg|, \frac{1}{|df|} \in L^{\infty}_{loc}(\mathbf{X})$) to deduce that

$$g \circ e_1 - g \circ e_0 = \int_0^1 [e_r^* dg](\hat{\pi}_r') dr = \int_0^1 h \circ e_r [e_r^* df](\hat{\pi}_r') dr = 0$$
 $\hat{\pi}$ -a.e..

This latter identity, however, is in contradiction with the fact that

$$\int g \circ e_0 \,\mathrm{d}\hat{\pi} = \int g \,\mathrm{d}(e_0)_* \hat{\pi} \le \sup_{V_0} g < \inf_{V_1} g \le \int g \,\mathrm{d}(e_1)_* \hat{\pi} = \int g \circ e_1 \,\mathrm{d}\hat{\pi},$$

thus proving the claim (3.4.14).

3.4 Rigidity theorem for spaces with positive spectrum

Property (3.4.14) defines the real valued function φ on the connected being the continuous image of a connected - set $I := f(B_{\bar{r}}(\bar{x})) \subset \mathbb{R}$. To conclude the proof it is therefore enough to show that $\varphi : I \to \mathbb{R}$ is locally Lipschitz, with a control on the local Lipschitz constant independent on the chosen neighbourhood.

Thus let $x \in B_{\bar{r}}(\bar{x})$, put $\alpha := f(x)$ and notice that for $|t| \ll 1$ we have $\mathsf{Fl}_t(x) \in B_{\bar{r}}(\bar{x})$ as well with - by the above discussion - $f(\mathsf{Fl}_t(x)) = \alpha + t$. Conclude noticing that

$$\begin{aligned} |\varphi(\alpha+t) - \varphi(\alpha)| &= |\varphi(f(\mathsf{Fl}_t(x))) - \varphi(f(x))| \\ &= |g(\mathsf{Fl}_t(x)) - g(x)| \le \mathrm{LIP}(g_{|B_{\overline{r}}(\overline{x})}) \|v\|_{L^{\infty}} |t|, \end{aligned}$$

where in the last inequality we used the fact that the speed of the curve $s \mapsto \mathsf{Fl}_s(x)$ is uniformly bounded by $||v||_{L^{\infty}}$.

In general this last result cannot be globabilized. In our case, however, this is possible thanks to the global properties of the function u we proved in Section 3.4.1.

Proposition 3.4.7. With the same notation and assumptions as in Assumption 3.4.2 the following holds.

There exists a function $\varphi \in C^{\infty}_{loc}(I)$ such that $|du| = \varphi \circ u$, where I := u(X), and it satisfies

$$\frac{1}{1-N}\varphi\varphi'' = 1 - \frac{1}{(1-N)^2}(\varphi')^2.$$
(3.4.16)

Proof. For $x \in X$ let U_x and φ_x be the neighbourhood and the function given by Proposition 3.4.6 above. By item (*ii*) in Proposition 3.4.5 we know that φ_x is strictly positive. Now suppose for a moment that we already know that φ_x is $C_{loc}^{1,1}$ for every $x \in X$.

We know that $|du| = \varphi_x \circ u$ holds on U_x , thus the regularity of φ_x and its positivity justify the chain rules

$$\begin{aligned} \Delta(|\mathrm{d}u|^{\frac{N-2}{N-1}}) &= \mathrm{div}(\frac{N-2}{N-1}(\varphi_x^{-\frac{1}{N-1}}\varphi_x') \circ u\nabla u) \\ &= \left(-\frac{N-2}{(N-1)^2}\varphi_x^{\frac{N-2}{N-1}}(\varphi_x')^2 + \frac{N-2}{N-1}\varphi_x^{1+\frac{N-2}{N-1}}\varphi_x''\right) \circ u \end{aligned}$$

having used also the fact that u is harmonic.

Since (3.4.5) can now be written as $\Delta(|du|^{\frac{N-2}{N-1}}) = -(N-2)\varphi_x^{\frac{N-2}{N-1}} \circ u$ we conclude that φ_x satisfies (3.4.16). It then follows by standard bootstrapping that φ_x is C_{loc}^{∞} , as in the statement.

The ODE (3.4.16) also gives, by ODE uniqueness, the desired rigidity, as it is clear that any two solutions $\varphi_x, \varphi_{x'}$ defined on some intervals $I_x, I_{x'} \subset$

106 Chapter 3. Applications to spaces with positive spectrum

(0,1) that coincide in some non-trivial interval $I \subset I_x \cap I_{x'}$ must coincide on the whole $I_x \cap I_{x'}$ and that the natural 'glued' function defined on $I_x \cup$ $I_{x'}$ is still a solution. Thus fix $\bar{x} \in X$, let φ be the maximal solution of (3.4.16) extending $\varphi_{\bar{x}}$ and let $A \subset X$ the set of those x's such that φ_x is the restriction of φ to some subinterval of its domain of definition. Since U_x is a neighbourhood of x and φ_x is defined on the open set $u(U_x)$ (by item (v) in Proposition 3.4.5), it is clear that A is both open and closed. Since it is not empty and X is connected (being geodesic) we conclude that A = X.

empty and X is connected (being geodesic) we conclude that A = X. We now prove that φ_x is $C_{loc}^{1,1}$ and to this aim we start claiming that for $\eta \in W^{1,2}(X)$ with support compact and contained in U_x the measure $\mu := u_*(\eta \mathfrak{m})$ is absolutely continuous with respect to \mathcal{L}^1 with density that is also absolutely continuous. Indeed, let ζ be a Lipschitz cut-off function with compact support and identically 1 on a neighbourhood of $\operatorname{supp}(\eta)$ and let (Fl_t) be the Regular Lagrangian Flow of $\zeta \nabla u$ (whose existence and uniqueness follow from $|\zeta \nabla u| \in L^{\infty}$, div $(\zeta \nabla u) = \langle \nabla \zeta, \nabla u \rangle \in L^{\infty}$ and $\zeta \nabla u \in W_C^{1,2}(TX)$ by Proposition 1.6.16). Then by Proposition 1.8.4 we see that $\frac{u \circ \operatorname{Fl}_t - u}{t} \to$ $|du|^2$ in $L^2(\operatorname{supp}(\eta), \mathfrak{m})$ as $t \to 0$. Notice also that item (*ii*) in Proposition 3.4.5 yields that $\frac{1}{|du|^2} \in W_{loc}^{1,2} \cap L_{loc}^{\infty}(X)$ (for Sobolev regularity recall that $u \in H_{loc}^{2,2}(X)$ and [G18b, Proposition 3.3.22]), thus the following computation is justified for any $\xi \in C_c^{\infty}(\mathbb{R})$:

$$-\int \xi' \,\mathrm{d}\mu = -\int \xi' \circ u \,\eta \,\mathrm{d}\mathfrak{m} = -\lim_{t \to 0} \int \frac{\xi \circ u \circ \mathsf{Fl}_t - \xi \circ u}{t} \frac{\eta}{|\mathrm{d}u|^2} \,\mathrm{d}\mathfrak{m}$$
$$= \int \xi \circ u \,\langle \mathrm{d}\Big(\frac{\eta}{|\mathrm{d}u|^2}\Big), \mathrm{d}u \rangle \,\mathrm{d}\mathfrak{m} = \int \xi \,\mathrm{d}\nu,$$

where $\nu := u_* \left(\langle \mathrm{d} \left(\frac{\eta}{|\mathrm{d}u|^2} \right), \mathrm{d}u \rangle \mathfrak{m} \right).$

This proves that the distributional derivative of μ is a Radon measure, and thus that $\mu \ll \mathcal{L}^1$ with BV density. To prove that the density is absolutely continuous it suffices to prove that $\nu \ll \mathcal{L}^1$. But this is obvious, as a direct consequence of what just proved is that $u_*\mathfrak{m} \ll \mathcal{L}^1$.

We are now ready to show that $\varphi = \varphi_x$ is $C_{loc}^{1,1}(u(U_x))$. We know from Proposition 3.4.6 that it is locally Lipschitz and that in U_x we have $|du| = \varphi \circ u$. Since we already recalled that $|du| \in W_{loc}^{1,2}$ we see that $d|du| = \varphi' \circ u \, du$. Also, we know from (3.4.5) and the chain rule for the Laplacian that in U_x we have $\Delta(|du|) = \psi \circ u$ for some locally bounded function ψ , that we can rewrite as $\operatorname{div}(\varphi' \circ u \, du) = \psi \circ u$. Now we take $\eta \in \operatorname{Test}(X)$ with support in U_x and $\xi \in C^{\infty}(\mathbb{R})$ and observe that

$$\int \langle \mathrm{d}(\xi \circ u \,\eta), \varphi' \circ u \,\mathrm{d}u \rangle \,\mathrm{d}\mathfrak{m} = -\int (\psi \,\xi) \circ u \,\eta \,\mathrm{d}\mathfrak{m} = -\int \psi \,\xi \rho \,\mathrm{d}\mathcal{L}^1,$$

where ρ is the density of $u_*(\eta \mathfrak{m})$. On the other hand we also have

$$\begin{split} \int \langle \mathrm{d}(\xi \circ u \,\eta), \varphi' \circ u \,\mathrm{d}u \rangle \,\mathrm{d}\mathfrak{m} &= \int (\xi' \varphi') \circ u \,|\mathrm{d}u|^2 \eta + (\varphi' \xi) \circ u \langle \mathrm{d}\eta, \mathrm{d}u \rangle \,\mathrm{d}\mathfrak{m} \\ &= \int \xi' \varphi' \rho_1 + \xi \varphi' \rho_2 \,\mathrm{d}\mathcal{L}^1, \end{split}$$

where ρ_1, ρ_2 are the densities of $u_*(|\mathrm{d}u|^2\eta\mathfrak{m})$ and $u_*(\langle \mathrm{d}\eta, \mathrm{d}u \rangle \mathfrak{m})$, respectively. By the arbitrariness of ξ we proved that the distributional derivative of $\varphi' \rho_1$ is equal to $\varphi' \rho_2 + \psi \rho \in L^1_{loc}$, and thus that $\varphi' \rho_1$ is (more precisely: has a representative that is) absolutely continuous. Since $|\mathrm{d}u|^2\eta$ is in $W^{1,2}(X)$ and has support in U_x , what previously proved shows that ρ_1 is also absolutely continuous.

We thus deduce that φ' is locally absolutely continuous on $\{\rho_1 > 0\} = u(\{\eta > 0\})$. Taking $\eta = \eta_n$ for (η_n) such that of $\bigcup_n \{\eta_n > 0\} = U_x$ we conclude.

3.4.3 Proof of Theorem 3.4.1

We are now ready to prove Theorem 3.4.1 combining the results of Chapter 2 and Sections 3.3, 3.4.1 and 3.4.2.

Proof of Theorem 3.4.1. Since $\lambda_1 \geq N-2 > 0$, we know from Proposition 3.3.9 that X has at least one end of infinite measure. Assume it has at least two of these. Then Corollary 3.3.7 gives the existence of a non-constant bounded harmonic function u on the whole X and then Proposition 3.4.7 that

$$|\mathrm{d}u| = \varphi \circ u \tag{3.4.17}$$

for a positive smooth function φ on u(X) that satisfies

$$\frac{1}{1-N}\varphi\varphi'' = 1 - \frac{1}{(1-N)^2}(\varphi')^2.$$
(3.4.18)

Then Proposition 3.4.5 tells that $|du| \neq 0$ a.e., and that putting $e_1 := \frac{\nabla u}{|\nabla u|}$ we have

Hess
$$u = \zeta \circ u |\mathrm{d}u| \mathrm{Id} - N\zeta \circ u |\mathrm{d}u| \mathrm{e}_1 \otimes \mathrm{e}_1$$
 for $\zeta = \frac{\varphi'}{1-N}$. (3.4.19)

We can therefore apply Lemma 2.1.1: let η be so that $\eta' = \frac{1}{\varphi} (\eta$ is defined up to an additive constant: the value of such constant will be chosen in a moment) and define the function $\mathbf{b} := \eta \circ u$. Then $\mathbf{b} \in H^{2,2}_{loc}(\mathbf{X})$ with

$$\begin{aligned} |\mathrm{db}| &\equiv 1, \quad (3.4.20) \\ \mathrm{Hess} \, \mathbf{b} &= \psi_{\mathsf{d}} \circ \mathbf{b} \big(\mathrm{Id} - \mathbf{e}_1 \otimes \mathbf{e}_1 \big), \quad \text{for} \quad \psi_{\mathsf{d}} &:= \zeta \circ \eta^{-1} \quad (3.4.21) \\ \Delta \mathbf{b} &= \psi_{\mathfrak{m}} \circ \mathbf{b}, \quad \text{for} \quad \psi_{\mathfrak{m}} &:= (N-1)\zeta \circ \eta^{-1} = (N-1)\psi_{\mathsf{d}}. \end{aligned}$$

To find explicitly ψ_{d} and $\psi_{\mathfrak{m}}$ notice that

$$\zeta'\varphi \stackrel{(3.4.19)}{=} \frac{\varphi''\varphi}{1-N} \stackrel{(3.4.18)}{=} 1 - \frac{(\varphi')^1}{(N-1)^2} \stackrel{(3.4.19)}{=} 1 - \zeta^2 \tag{3.4.23}$$

(3.4.22)

and thus

$$\psi'_{\mathsf{d}} = (\zeta \circ \eta^{-1})' = (\zeta' \tfrac{1}{\eta'}) \circ \eta^{-1} = (\zeta' \varphi) \circ \eta^{-1} \overset{(3.4.23)}{=} 1 - \zeta^2 \circ \eta^{-1} = 1 - \psi_{\mathsf{d}}^2$$

proving that $\psi_{\mathsf{d}}(z) = \tanh(z+c)$ for some $c \in \mathbb{R}$. Since $\psi_{\mathsf{d}} \circ \eta$ is equal to the given function ζ , we see that replacing η with $\tilde{\eta} := \eta + \alpha$ for $\alpha \in \mathbb{R}$ means replacing ψ_{d} with $\tilde{\psi}_{\mathsf{d}} = \psi_{\mathsf{d}}(\cdot - \alpha)$. Thus we can, and will, choose η so that $\psi_{\mathsf{d}} = \tanh$.

It follows by the defining properties (2.2.9b), (2.2.10) that $w_{d} = \cosh$. By (3.4.22) we also have $\psi_{\mathfrak{m}} = (N-1) \tanh$ and thus $w_{\mathfrak{m}} = \cosh^{N-1}$.

Then Theorem 2.5.3 gives the required warped product structure, where the fiber is the space $X' := b^{-1}(0)$ equipped with the distance d' and the measure \mathfrak{m}' defined in (2.2.3) and (2.2.4) respectively. Moreover, we can apply Proposition 2.5.4 with $\overline{z} = 0$ to deduce that $(X', \mathfrak{d}', \mathfrak{m}')$ is $\mathsf{RCD}(-(N-1), N)$.

It remains to prove that X' is compact. Say not. We are going to prove that in this case X has at most one end, thus contradicting the assumption made at the beginning of the proof.

Let $K \subset X$ be compact and let $K' := \Pr^{-1}(\Pr(K)) \cap b^{-1}(b(K))$. Since $T(K') = b(K) \times \Pr(K) \subset \mathbb{R} \times_w X'$ we see that the 'rectangle' K' is also compact. Let $x_0, x_1 \in X \setminus K'$. Then for i = 0, 1 either $\Pr(x_i) \notin \Pr(K')$ or $b(x_i) \notin b(K')$ (or both). Say $b(x_i) \notin b(K')$ for i = 0, 1 and use the assumption that X' is not compact to find $z \in X' \setminus \Pr(K')$. Then the curve $t \mapsto \operatorname{Fl}_t(z)$ (here Fl is the flow of ∇b) does not meet K' and, moreover, there are $t_0, t_1 \in \mathbb{R}$ with $b(\operatorname{Fl}_{t_i}(z)) = b(x_i), i = 0, 1$. Recalling that the level sets of b are path connected, we can find curves joining x_i and $\operatorname{Fl}_{t_i}(z)$ lying entirely on level sets. The assumption $b(x_i) \notin b(K')$ ensures that these curves do not meet K', thus the path obtained by gluing these curves and $t \mapsto \operatorname{Fl}_t(z)$ produces a curve from x_0 to x_1 that does not meet K', showing that x_0, x_1 belong to the same connected component of X \ K'. Since an analogous construction can be made if one, or both, of the x_i 's are with $\Pr(x_i) \notin \Pr(K')$, we see that X \ K' has only one connected component, providing the desired contradiction. \Box
3.5 Rigidity theorem for spaces with maximal spectrum

By Cheng's Theorem the first eigenvalue of the Laplacian of an *n*-dimensional Riemannian manifold with Ricci curvature bounded from below by -(n-1) is bounded from above by $\frac{(n-1)^2}{4}$. The same result holds in the non-smooth setting: it is a consequence of the following result proved in metric measure space by Sturm (see [S94, Theorem 5]).

Theorem 3.5.1. Let $v_*(r) := \inf_{x \in X} \frac{\mathfrak{m}(B_r(x))}{B_1(x)}$. If there exists $k \ge 0$ such that

$$\liminf_{r \to +\infty} \frac{1}{r} \log(v_*(r)) \le k,$$

then $\lambda_1 \in \left[0, \frac{k^2}{4}\right]$.

Theorem 3.5.2. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ with K < 0 and $N \in (1, +\infty)$. Then

$$\lambda_1 \le \frac{-(N-1)K}{4}.$$

Proof. By the Bishop-Gromov inequality (1.5.12) we can estimate $v_*(r)$ from above with $Ce^{r\sqrt{-(N-1)K}}$. Theorem 3.5.1 concludes the proof.

Its proof easily follows combining the Bishop-Gromov inequality (1.5.12) with the following theorem (see [S94, Theorem 5]).

Theorem 3.5.2 motivates the study of the case $\lambda_1 = \frac{(N-1)^2}{4}$, i.e. the spaces in which λ_1 is the maximum admissible.

3.5.1 Isomorphism with the warped product

The proof of Theorem 0.0.8 is very similar to the one of Theorem 3.4.1 saw in the previous section. We recall the precise statement of the theorem.

Theorem 3.5.3. Let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(-(N-1), N)$ space with N > 3and $\operatorname{supp}(\mathfrak{m}) = X$, and assume that the first eigenvalue of the Laplacian λ_1 is equal to $\frac{(N-1)^2}{4}$. Then one of the following holds:

- i) X has only one end;
- ii) X is isomorphic as metric measure space to a warped product space $\mathbb{R} \times_w$ X', where X' is a compact $\mathsf{RCD}(0, N)$ space and the warping functions are

 $w_{\mathsf{d}}(t) := e^t$ and $w_{\mathfrak{m}}(t) := e^{Nt}$.

To begin we observe that $\lambda_1(X) > N - 2$, then, by Theorem 3.4.1, the space has only one end with infinite volume. We fix then the following assumptions.

Assumption 3.5.4. (X, d, \mathfrak{m}) is an $\mathsf{RCD}(-(N-1), N)$ space with N > 3and $\lambda_1 = \frac{(N-1)^2}{4}$. We also assume that $\mathrm{supp}(\mathfrak{m}) = X$ and that X has at least two ends.

Finally, we shall denote by u the positive and non-constant harmonic function on X given by Corollary 3.3.8.

As we did in Proposition 3.4.5, we prove that the positive non-constant harmonic function u satisfies the equality in the Bochner+Kato inequality (3.4.7) with F = 0, and from this we deduce its Hessian.

As in the proof of Proposition 3.4.5, an inequality (the equivalent of (3.4.8)) comes from the Bochner inequality, and the other follows by the definition of λ_1 and the estimates (3.3.14) and (3.3.15) (as in (3.4.10)). The first inequality in this case has been studied by H.-C. Zhang and X.-P. Zhu in [ZZ], where they proved a sharp estimate for the gradient of harmonic functions.

Theorem 3.5.5 (Li-Yau gradient estimate). Let $K \leq 0$ and $N \in (1, +\infty)$, and let (X, d, \mathfrak{m}) be an $\mathsf{RCD}(K, N)$ metric measure space. Taking u positive harmonic function on X it holds

$$|\mathrm{d} u| \leq \sqrt{-K(N-1)}u.$$

Proposition 3.5.6 (Properties of the Busemann function). With the same assumptions and notation as in Assumption 3.5.4 the following hold:

- i) |du| = (N-1)u holds \mathfrak{m} -a.e., in particular |du| is locally Lipschitz and strictly positive;
- ii) putting $e_1 := \frac{\nabla u}{|\nabla u|}$, which is well defined by the above, we have

$$\text{Hess}(u) = -(N-1)u(\text{Id} - Ne_1 \otimes e_1).$$
 (3.5.1)

Proof. Also in this case it is easy to see, as in Lemma 3.4.3, that $u \in \text{Test}_{loc}(X)$ (since here u is not globally bounded we do not have the global Lipschitzianity in this case).

Starting with item (i), the inequality $|du| \le (N-1)u$ is given by Theorem 3.5.5, so we only have to prove that the opposite inequality holds.

Let $f := \sqrt{u}$. Then, by the chain rule for the Laplacian (3.1.1) it holds

$$\Delta f = -\frac{1}{4}u^{-\frac{3}{2}}|\mathrm{d}u|^2 \mathfrak{m} \ge -\frac{(N-1)^2}{4}f\mathfrak{m}, \qquad (3.5.2)$$

which means that, as in the proof of Proposition 3.4.5, $\Delta f + \lambda_1 f \mathfrak{m} \geq 0$.

Moreover, repeating the same computations as in (3.4.9) we conclude that

$$\int \varphi^2 f \,\mathrm{d}(\mathbf{\Delta} f + \lambda_1 f \mathfrak{m}) \le \int f^2 |\mathrm{d}\varphi|^2 \,\mathrm{d}\mathfrak{m}. \tag{3.5.3}$$

As in the proof of Proposition 3.4.5 the aim is to choose a sequence of functions $(\varphi_R) \rightarrow \hat{1}$ that sends the right hand side of (3.5.3) to 0. We start noticing that we can ignore every end with finite volume on which $u \leq 1$ (so every end with finite volume except for E_2), since we can choose φ_R similarly to what we did in the proof of Proposition 3.3.9 and see that their contribution in the right hand side of (3.5.3) is negligible.

In the following, for brevity and simplicity of the notation, we assume that the space has only the two ends E_1 and E_2 (noting that we can define φ_R independently on every end, so the argument above is enough to conclude in presence of other ends with finite volume).

Take φ_R defined as $\varphi_R := (1 - R^{-1} \mathsf{d}(\cdot, B_R(p)))^+$. Letting $R \uparrow +\infty$ we get

$$\int f \,\mathrm{d}(\mathbf{\Delta} f + \lambda_1 f \mathfrak{m}) \leq \liminf_{R \to +\infty} \frac{1}{R^2} \int_{B_{2R}(p) \setminus B_R(p)} f^2 \,\mathrm{d}\mathfrak{m}.$$
(3.5.4)

We prove that the right hand side is zero. More precisely, we claim that there exists a constant C > 0 such that for every R > 0 large enough it holds

$$\int_{B_R(p)} f^2 \,\mathrm{d}\mathfrak{m} = \int_{B_R(p)} u \,\mathrm{d}\mathfrak{m} \le CR. \tag{3.5.5}$$

On the end with infinite volume E_1 we have that

$$\int_{E_1(R+1)\setminus E_1(R)} u \,\mathrm{d}\mathfrak{m} \le \left(\int_{E_1(R+1)\setminus E_1(R)} u^2 \,\mathrm{d}\mathfrak{m}\right)^{\frac{1}{2}} \sqrt{V_{E_1}(R+1)}$$

(3.3.2) + (3.3.14) $\le C e^{-\frac{N-1}{2}R} e^{\frac{N-1}{2}R} = C,$

and summing up on R

$$\int_{E_1(R)} f^2 \,\mathrm{d}\mathfrak{m} \le CR. \tag{3.5.6}$$

Focusing now on E_2 , we note that, by Theorem 3.5.5 and the Sobolev-to-Lipschitz property it holds

$$u \le C + e^{(N-1)\mathsf{d}_p}$$
 m-a.e.,

then for every R large enough

$$\int_{E_2(R+1)\setminus E_2(R)} u \, \mathrm{d}\mathfrak{m} \leq \int_{E_2(R+1)\setminus E_2(R)} C + e^{(N-1)\mathsf{d}_p} \, \mathrm{d}\mathfrak{m} \\
\leq (C + e^{(N-1)(R+1)})(V_{E_2}(R+1) - V_{E_2}(R)) \\
\leq (C + e^{(N-1)(R+1)})(V_{E_2}(\infty) - V_{E_2}(R)) \\
(3.3.15) \leq C(C + e^{(N-1)(R+1)})e^{-(N-1)R} = C.$$

Summing up on R we obtain that

$$\int_{E_2(R)} f^2 \,\mathrm{d}\mathfrak{m} \le CR,\tag{3.5.7}$$

and this, together with (3.5.6) concludes the proof of (3.5.5).

We conclude that $\Delta f = -\lambda_1 f \mathfrak{m}$, then, since the equality holds in (3.5.2), we have that

$$|du| = (N-1)u. (3.5.8)$$

From an easy computation, we see that u satisfies an equality in the Bochner+Kato inequality, i.e. (3.4.7) holds with F = 0: indeed, using the chain rule for the Laplacian (3.1.1), $\Delta u \equiv 0$ and (3.5.8) we have

$$\mathbf{\Delta}^{\frac{|\mathbf{d}u|^2}{2}} = \frac{(N-1)^2}{2} \mathbf{\Delta}(u^2) = (N-1)^2 |\mathbf{d}u|^2 \mathfrak{m}$$

and

$$(\frac{N}{N-1} |\mathbf{d}| \mathbf{d}u||^2 - (N-1) |\mathbf{d}u|^2) \mathfrak{m} = (N(N-1) |\mathbf{d}u|^2 - (N-1) |\mathbf{d}u|^2) \mathfrak{m}$$

= $(N-1)^2 |\mathbf{d}u|^2 \mathfrak{m}.$

Arguing as in the proof of Proposition 3.4.5 we conclude that the Hessian of u is of the form $\alpha(\operatorname{Id} - Ne_1 \otimes e_1)$ for $e_1 := \frac{\nabla u}{|\nabla u|}$ and

$$\alpha = \frac{\langle \mathbf{d} | \mathbf{d} u |, \mathbf{d} u \rangle}{(1 - N) | \mathbf{d} u |} = -(N - 1)u.$$

We observe that in this case we do not need to repeat the argument of Section 3.4.2, since we already have that |du| = (N - 1)u, then we can conclude the proof of the "splitting" in Theorem 3.5.3 arguing as in Section 3.4.3.

Proposition 3.5.7. With the same assumptions and notation as in Assumption 3.5.4 the following holds: the space X is isomorphic as metric measure space to a warped product space $\mathbb{R} \times_w X'$, where X' is compact and the warping functions are

 $w_{\mathsf{d}}(t) := e^t, \qquad and \qquad w_{\mathfrak{m}}(t) := e^{(N-1)t}.$

Proof. The proof is identical to the one of Theorem 3.4.1 except for the RCD condition for the quotient space X' (we will talk about that in the next section).

Fixing $b := -\frac{\log(u)}{N-1}$, from Proposition 3.5.6 follows that

$$|\mathrm{db}| \equiv 1, \tag{3.5.9}$$

$$\operatorname{Hess} \mathbf{b} = \operatorname{Id} - \mathbf{e}_1 \otimes \mathbf{e}_1, \tag{3.5.10}$$

$$\Delta \mathbf{b} = N - 1. \tag{3.5.11}$$

Defining w_d and w_m as in (2.2.9) and (2.2.10), we have that $w_d(t) = e^t$ and $w_m = e^{(N-1)t}$.

From Theorem 2.2.1 follows that X is isomorphic to $\mathbb{R} \times_w X'$ with $X' := b^{-1}(0)$ and warping functions w_d and w_m . Moreover X' is compact since, if it was not then the space $\mathbb{R} \times_w X'$ would have only one end (the argument is identical to the one in the proof of Theorem 3.4.1).

3.5.2 RCD condition of the quotient space

In this case Theorem 2.5.4 cannot be applied, so we have to argue differently to prove the RCD condition of the quotient space X'.

This case has been studied in [CD+], we recall briefly their argument since it is completely different from ours in Theorem 2.5.4.

As seen in Section 2.5, the first three properties of Definition 1.5.1 are satisfied in the space X'. We prove that the Bochner inequality (1.5.1) holds too.

The first step to prove it is to study how the Laplacian of the warped product space $\mathbb{R} \times_w X'$ behaves with respect to the Laplacian in X'. We denote by Δ the Laplacian in $\mathbb{R} \times_w X'$ and by $\underline{\Delta}$ the Laplacian in X'.

Lemma 3.5.8. Let $\rho \in C_c^{\infty}(\mathbb{R})$ and $f \in D(\underline{\Delta})$. Moreover assume that $w_{\mathfrak{m}} \in C^1(\mathbb{R})$. Indicating with $\overline{f}, \overline{\rho} : \mathbb{R} \times_w X' \to \mathbb{R}$ the functions $\overline{f}(t, x') := f(x')$ and $\overline{\rho}(t, x') := \rho(t)$, it holds $\overline{f}\overline{\rho} \in D(\Delta)$ and $\Delta(\overline{f}\overline{\rho}) \in W^{1,2}(\mathbb{R} \times_w X')$.

Moreover

$$\Delta(f\overline{\rho}) = \rho w_{\mathsf{d}}^{-2} \underline{\Delta} f + (\rho' w_{\mathfrak{m}})' w_{\mathfrak{m}}^{-1} f. \qquad (3.5.12)$$

Remark 3.5.9. In (3.5.12) with a little abuse of notation we omitted the right compositions with the projections $\pi_1(t, x') := t$ and $\pi_2(t, x') := x'$.

Proof. We indicate with X_w the warped product $\mathbb{R} \times_w X'$. By Theorem 1.7.6 follows easily that function $\overline{f}\overline{\rho}$ is in $W^{1,2}(\mathbf{X}_w)$ for every f, ρ as in the assumptions.

Recalling nations (2.2.5) and (2.4.2), let $\tilde{\mathcal{A}}$ be the set of functions f: $X_w \to \mathbb{R}$ such that $f \circ \mathsf{T} \in \mathcal{A}$, and similarly we define $\tilde{\mathcal{G}}$ and $\tilde{\mathcal{H}}$. By Lemma 2.4.1 we have that $\tilde{\mathcal{A}}$ is dense in $W^{1,2}(\mathbf{X}_w)$. Let $\varphi \in \operatorname{Test}(\mathbf{X}_w) \cap \tilde{\mathcal{A}}$ given by $\varphi = \sum_{i=1}^n a_i \overline{h}_i \overline{g}_i$, where $a_i \in \mathbb{R}, \ \overline{h}_i \in \tilde{\mathcal{H}}$

and $\overline{q}_i \in \mathcal{G}$.

Using the Leibniz rule and remembering the argument in Section 2.4, we have that

$$\int \langle \nabla(\overline{f}\overline{\rho}), \nabla\varphi \rangle \,\mathrm{d}\mathfrak{m}_w = \sum_{i=1}^n a_i \int \left(\overline{f}\overline{g}_i \langle \nabla\overline{\rho}, \nabla\overline{h}_i \rangle + \overline{\rho}\overline{h}_i \langle \nabla\overline{f}, \nabla\overline{g}_i \rangle\right) \,\mathrm{d}\mathfrak{m}_w.$$

Noting that, by Theorem 1.7.6 follows that

$$\langle \nabla \overline{f}, \nabla \overline{g}_i \rangle_{\mathcal{X}_w} = w_{\mathsf{d}}^{-2} \langle \nabla f, \nabla g_i \rangle_{\mathcal{X}'} \text{ and } \langle \nabla \overline{\rho}, \nabla \overline{h}_i \rangle_{\mathcal{X}_w} = \rho' h'_i,$$

we conclude that

$$\sum_{i=1}^{n} a_{i} \int \overline{\rho} \overline{h}_{i} \langle \nabla \overline{f}, \nabla \overline{g}_{i} \rangle \,\mathrm{d}\mathfrak{m}_{w} = \sum_{i=1}^{n} a_{i} \int_{\mathbb{R}} f g_{i} w_{\mathsf{d}}^{-2} w_{\mathfrak{m}} \int_{X'} \langle \nabla f, \nabla g_{i} \rangle \,\mathrm{d}\mathfrak{m}' \,\mathrm{d}\mathcal{L}^{1}$$
$$= -\sum_{i=1}^{n} a_{i} \int_{\mathbb{R}} \rho h_{i} w_{\mathsf{d}}^{-2} w_{\mathfrak{m}} \,\mathrm{d}\mathcal{L}^{1} \int_{X'} g_{i} \underline{\Delta} f, \,\mathrm{d}\mathfrak{m}'$$

and

$$\sum_{i=1}^{n} a_{i} \int \overline{f} \overline{g}_{i} \langle \nabla \overline{\rho}, \nabla \overline{h}_{i} \rangle \, \mathrm{d}\mathfrak{m}_{w} = \sum_{i=1}^{n} a_{i} \int_{\mathbb{R}} \rho' h'_{i} w_{\mathfrak{m}} \, \mathrm{d}\mathcal{L}^{1} \int_{\mathrm{X}'} f g_{i} \, \mathrm{d}\mathfrak{m}'$$
$$= -\sum_{i=1}^{n} a_{i} \int_{\mathbb{R}} (\rho' w_{\mathfrak{m}})' h_{i} \, \mathrm{d}\mathcal{L}^{1} \int_{\mathrm{X}'} f g_{i}, \mathrm{d}\mathfrak{m}'.$$

Then for every $\varphi \in \text{Test}(X_w) \cap \tilde{A}$ it holds

$$\int \langle \nabla(\overline{f}\overline{\rho}), \nabla\varphi \rangle \,\mathrm{d}\mathfrak{m}_w = -\int \varphi \Big(\rho w_{\mathsf{d}}^{-2}\underline{\Delta}f + f(w_{\mathfrak{m}}^{-1}(\rho'w_{\mathfrak{m}})'\Big) \,\mathrm{d}\mathfrak{m}_w, \quad (3.5.13)$$

where, as in the previous remark, we omitted the right composition with the projections.

Since $\text{Test}(X_w) \cap \tilde{A}$ is dense in $W^{1,2}(X_w)$, we have that (3.5.13) holds for every $\varphi \in W^{1,2}(X_w)$, then we conclude that $\overline{f\rho} \in D(\Delta)$ and

$$\Delta(\overline{f}\overline{\rho}) = \rho w_{\mathsf{d}}^{-2}\underline{\Delta}f + (\rho'w_{\mathfrak{m}})'w_{\mathfrak{m}}^{-1}f.$$

Starting from the weak Bochner inequality (1.5.1) on the space $\mathbb{R} \times_w X'$ and using (3.5.12) we can see that, in our case, the Bochner inequality holds on X'.

Proposition 3.5.10. Assume that X' and $\mathbb{R} \times_w X'$ are as in Proposition 3.5.7. Then for every $f \in D(\underline{\Delta})$ such that $\underline{\Delta}f \in W^{1,2}(X')$ and all non-negative $g \in D(\underline{\Delta}) \cap L^{\infty}(X')$ such that $\underline{\Delta}g \in L^{\infty}(X')$ the following is satisfied:

$$\frac{1}{2} \int \underline{\Delta} g |\mathrm{d}f|_{\mathrm{X}'}^2 \,\mathrm{d}\mathfrak{m}' \ge \frac{1}{N} \int g(\underline{\Delta}f)^2 \,\mathrm{d}\mathfrak{m}' + \int g \langle \nabla \underline{\Delta}f, \nabla f \rangle_{\mathrm{X}'} \,\mathrm{d}\mathfrak{m}'. \quad (3.5.14)$$

Sketch of the proof. Let $f \in D(\underline{\Delta})$ such that $\underline{\Delta}f \in W^{1,2}(\mathbf{X}')$, and $g \in D(\underline{\Delta}) \cap L^{\infty}(\mathbf{X}')$ such that $g \geq 0$ and $\underline{\Delta}g \in L^{\infty}(\mathbf{X}')$.

Take $\rho \in C_c^{\infty}(\mathbb{R})$. Since f and g satisfy the hypotheses of Lemma 3.5.8, we have that the following weak Bochner inequality holds for $\overline{f\rho}$ and $\overline{g\rho}$:

$$\frac{1}{2} \int \Delta(\overline{g\rho}) |\mathrm{d}(\overline{f\rho})|^2 \,\mathrm{d}\mathfrak{m}_w \geq \frac{1}{N} \int \overline{g\rho} (\Delta(\overline{f\rho}))^2 \,\mathrm{d}\mathfrak{m}_w \\
- (N-1) \int \overline{g\rho} |\mathrm{d}(\overline{f\rho})|^2 \,\mathrm{d}\mathfrak{m}_w \\
+ \int \overline{g\rho} \langle \nabla(\Delta(\overline{f\rho})), \nabla(\overline{f\rho}) \rangle \,\mathrm{d}\mathfrak{m}_w.$$
(3.5.15)

Using (3.5.12), (1.7.1) and the Leibniz rule all the terms of (3.5.15) can be split in products of integrals on \mathbb{R} and on X'. For instance:

$$\begin{split} \int \Delta(\overline{g\rho}) |\mathrm{d}(\overline{f\rho})|^2 \,\mathrm{d}\mathfrak{m}_w &= \int (\rho w_{\mathsf{d}}^{-2} \underline{\Delta}g + (\rho' w_{\mathfrak{m}})' w_{\mathfrak{m}}^{-1}g) \\ &\quad (\rho^2 w_{\mathsf{d}}^{-2} |\mathrm{d}f|_{X'}^2 + f^2(\rho')^2) \,\mathrm{d}\mathfrak{m}_w \\ &= \int_{\mathbb{R}} \rho^3 w_{\mathsf{d}}^{-4} w_{\mathfrak{m}} \,\mathrm{d}\mathcal{L}^1 \int_{X'} |\mathrm{d}f|_{X'}^2 \underline{\Delta}g \,\mathrm{d}\mathfrak{m}' \\ &\quad + \int_{\mathbb{R}} \rho(\rho')^2 w_{\mathsf{d}}^{-2} w_{\mathfrak{m}} \,\mathrm{d}\mathcal{L}^1 \int_{X'} f^2 \underline{\Delta}g \,\mathrm{d}\mathfrak{m}' \\ &\quad + \int_{\mathbb{R}} \rho^2 w_{\mathsf{d}}^{-2} (\rho' w_{\mathfrak{m}})' \,\mathrm{d}\mathcal{L}^1 \int_{X'} g |\mathrm{d}f|_{X'}^2 \,\mathrm{d}\mathfrak{m}' \\ &\quad + \int_{\mathbb{R}} (\rho')^2 (\rho' w_{\mathfrak{m}})' \,\mathrm{d}\mathcal{L}^1 \int_{X'} g f^2 \,\mathrm{d}\mathfrak{m}', \end{split}$$
(3.5.16)

and similarly for the other three terms of (3.5.15).

The idea to conclude is then to find a sequence of functions (ρ_n) such that all the "unwanted terms" go to 0 as $n \to +\infty$ leaving only (3.5.14).

Let $\rho_0 \in C_c^{\infty}(\mathbb{R})$ be a cut-off function which is identically 1 on [-1, 1] and 0 outside [-2, 2] and let $\rho_n(t) := \rho_0(t+n)$. Remembering that $w_d(t) = e^t$ and $w_{\mathfrak{m}}(t) = e^{(N-1)t}$, we can observe that all the "real integrals" in (3.5.16) and in the similar terms derived from (3.5.15) goes to 0 faster than $\int \rho^3 w_{\mathsf{d}}^{-4} w_{\mathsf{m}} \, \mathrm{d}\mathcal{L}^1$:

$$\int \rho^3 w_{\mathsf{d}}^{-4} w_{\mathfrak{m}} \, \mathrm{d}\mathcal{L}^1 \ge \int_{-(n+1)}^{-(n-1)} e^{(N-5)s} \, \mathrm{d}s = C e^{-(N-5)n},$$

$$\int_{\mathbb{R}} \rho(\rho')^2 w_{\mathsf{d}}^{-2} w_{\mathfrak{m}} \, \mathrm{d}\mathcal{L}^1 \le C \int_{-(n+2)}^{-(n-2)} e^{(N-3)s} \, \mathrm{d}s = C e^{-(N-3)n},$$

$$\int_{\mathbb{R}} \rho^2 w_{\mathsf{d}}^{-2} (\rho' w_{\mathfrak{m}})' \, \mathrm{d}\mathcal{L}^1 \le C \int_{-(n+2)}^{-(n-2)} e^{(N-3)s} \, \mathrm{d}s = C e^{-(N-3)n},$$

$$\int_{\mathbb{R}} (\rho')^2 (\rho' w_{\mathfrak{m}})' \, \mathrm{d}\mathcal{L}^1 \le C \int_{-(n+2)}^{-(n-2)} e^{(N-1)s} \, \mathrm{d}s = C e^{-(N-1)n},$$

and so on for the other terms. Dividing (3.5.15) by $\int \rho^3 w_d^{-4} w_m \, \mathrm{d}\mathcal{L}^1$ and sending n to $+\infty$ we obtain (3.5.14).

Thanks to Proposition 3.5.10 we conclude that the space (X', d', \mathfrak{m}') is $\mathsf{RCD}(0, N)$, and this concludes the proof of Theorem 3.5.3.

Bibliography

- [A] L. AMBROSIO, Transport equation and Cauchy problem for BV vector fields, Invent. Math. **158** (2004), 227–26.
- [AB] L. AMBROSIO, J. BERTRAND, *DC calculus*, Math. Z. **288** (2018), no. 3-4, 1037–1080.
- [AGS13] L. AMBROSIO, N. GIGLI, G. SAVARÉ, Density of Lipschitz functions and equivalence of weak gradients in metric measure spaces, Rev. Mat. Iberoam. 29 (2013), no. 3, 969-996.
- [AGS14a] L. AMBROSIO, N. GIGLI, G. SAVARÉ, Calculus and heat flow in metric measure spaces and applications to spaces with Ricci bounds from below, Invent. Math. 195 (2014), no. 2, 289–391.
- [AGS14b] L. AMBROSIO, N. GIGLI, G. SAVARÉ, Metric measure spaces with Riemannian Ricci curvature bounded from below, Duke Math. J. 163 (2014), no. 7, 1405-1490.
- [AGS15] L. AMBROSIO, N. GIGLI, G. SAVARÉ, Bakry-Émery curvaturedimension condition and Riemannian Ricci curvature bounds, The Annals of Probability 43 (2015), no. 1, 339-404.
- [AMS] L. AMBROSIO, A. MONDINO, G. SAVARÉ, On the Bakry-Emery condition, the gradient estimates and the local-to-global property of RCD*(K, N) metric measure spaces, J. Geom. Anal. 26 (2016), no. 1, 24-56.
- [AT14] L. AMBROSIO, D. TREVISAN, Well-posedness of Lagrangian flows and continuity equations in metric measure spaces, Anal. PDE 7 (2014), no. 5, 1179–1234.
- [AT15] L. AMBROSIO, D. TREVISAN, Lecture notes on the DiPerna-Lions theory in abstract measure spaces, Preprint, arXiv:1505.05292 (2015).

- [BaSt] K. BACHER, K.-T. STURM, Localization and tensorization properties of the curvature-dimension condition for metric measure spaces, J. Funct. Anal. **259** (2010), no. 1, 28–56.
- [BB] A. BJORN, J. BJORN, Nonlinear potential theory on metric spaces, EMS Tracts in Mathematics 17, European Mathematical Society (EMS), Zürich (2011).
- [B] V. BOGACHEV, *Measure theory. Vol. I, II*, Springer-Verlag, Berlin (2007).
- [BPS] E. BRUÈ, E. PASQUALETTO, D. SEMOLA, Rectifiability of the reduced boundary for sets of finite perimeter over RCD(K, N) spaces, J. Eur. Math. Soc. **25** (2023), no. 2, pp. 413–465.
- [BS20a] E. BRUÈ, D. SEMOLA, Constancy of the dimension for RCD(K, N) spaces via regularity of Lagrangian flows, Comm. Pure Appl. Math. **73** (2020), no. 6, 1141–1204.
- [BS20b] E. BRUÈ, D. SEMOLA, Regularity of Lagrangian flows over $\operatorname{RCD}^*(K, N)$ spaces, J. Reine Angew. Math. **765** (2020), 171–203.
- [BCM] V. BUFFA, G. E. COMI, M. MIRANDA JR, On BV functions and essentially bounded divergence-measure fields in metric spaces, Rev. Mat. Iberoam. 38 (2021), no. 3, 883-946.
- [BBI] D. BURAGO, Y. BURAGO, S. IVANOV, A course in metric geometry, Graduate studies in mathematics **33**, American Mathematical Society, Providence, RI (2001).
- [CMM] F. CAVALLETTI, F. MAGGI, A. MONDINO, Quantitative isoperimetry à la Levy-Gromov, Comm. Pure Appl. Math. 72 (2019), 1631-1677.
- [CaMi] F. CAVALLETTI, E. MILMAN, The globalization theorem for the curvature-dimension condition, Invent. Math. 226 (2021), no. 1, 1-137.
- [CaMo17] F. CAVALLETTI, A. MONDINO, Sharp and rigid isoperimetric inequalities in metric-measure spaces with lower Ricci curvature bounds, Invent. Math. 208 (2017), no. 7, 803-849.
- [CaMo20] F. CAVALLETTI, A. MONDINO, New formulas for the Laplacian of distance functions and applications, Anal. PDE 13 (2020), no. 7, 2091–2147.

- [C] J. CHEEGER, Differentiability of Lipschitz Functions on Metric Measure Spaces, GAFA, Geom. funct. anal. 9 (1999), 428–517.
- [CC96] J. CHEEGER, T. H. COLDING, Lower bounds on Ricci curvature and the almost rigidity of warped products, Ann. of Math. (2) **144** (1996), no. 1, 189-237.
- [CC97] J. CHEEGER, T. H. COLDING, On the structure of spaces with Ricci curvature bounded below. I, J. Differential Geom. 46 (1997), no. 3, 406-480.
- [CC00a] J. CHEEGER, T. H. COLDING, On the structure of spaces with Ricci curvature bounded below. II, J. Differential Geom. 54 (2000), no. 1, 13-35.
- [CC00b] J. CHEEGER, T. H. COLDING, On the structure of spaces with Ricci curvature bounded below. III, J. Differential Geom. **54** (2000), no. 1, 37-74.
- [CG] J. CHEEGER, D. GROMOLL, The splitting theorem for manifolds of nonnegative Ricci curvature, J. Differential Geom. 6 (1971), 119-128.
- [CY] S. Y. CHENG, S. T. YAU, Differential equations on Riemannian manifolds and thei geometric applications, Comm. Pure Appl. Math. 28 (1975), no. 3, 333-354.
- [C-V] S. COHN-VOSSEN, Totalkrümmung und geodätische Linien auf einfach zusammenhängenden offenen volständigen Flächenstücken, Recueil Math. Moscow, 43 (1936), 139-163.
- [CN] T. H. COLDING, A. NABER, Sharp Hölder continuity of tangent cones for spaces with a lower Ricci curvature bound and applications, Ann. of Math. (2) **176** (2012), no. 2, 1173-1229.
- [CD+] C. CONNELL, X. DAI, J. NÙÑEZ-ZIMBRÒN, R. PERALES, P. SUÀREZ-SERRATO, G. WEI, Maximal volume entropy rigidity for RCD*(-(N-1), N) spaces, Lond. Math. Soc. (2) 104 (2021), no. 4, 1615–1681.
- [CEMS] D. CORDERO-ERAUSQUIN, R. J. MCCANN, M. SCHMUCKEN-SCHLÄGER, A Riemannian interpolation inequality 'a la Borell, Brascamp and Lieb, Invent. Math. **146** (2001), no. 2, 219–257.

- [DPG] G. DE PHILIPPIS, N. GIGLI, From volume cone to metric cone in the nonsmooth setting, Geom. Funct. Anal. 26 (2016), no. 6, 1526-1587.
- [DPMR] G. DE PHILIPPIS, A. MARCHESE, F. RINDLER, On a conjecture of Cheeger, Preprint, arXiv:1607.02554 (2016).
- [DPR] G. DE PHILIPPIS, F. RINDLER, On the structure of *A*-free measures and applications, Annals of Math. **184** (2016), 1017–1039.
- [DGP] C. DEBIN, N. GIGLI, E. PASQUALETTO, Quasi-continuous vector fields on RCD spaces, Potential Anal. 54 (2021), no. 1, 183–211.
- [D] Q. DENG, Hölder continuity of tangent cones in $\mathsf{RCD}(K, N)$ spaces and applications to non-branching, Preprint, arXiv:2009.07956 (2020).
- [DPL] R. J. DI PERNA, P. L. LIONS, Ordinary differential equations, transport theory and Sobolev spaces, Invent. Math. 98 (1989), 511-547.
- [EKS] M. ERBAR, K. KUWADA, K.-T. STURM, On the equivalence of the entropic curvature-dimension condition and Bochner's inequality on metric measure spaces, Invent. Math. 201 (2015), no. 3, 993-1071.
- [EG] L. C. EVANS, R. F. GARIEPY, Measure theory and fine properties of functions, Advanced Mathematical Studies, CRC Press, Boca Raton (1992).
- [G13a] N. GIGLI, The splitting theorem in non-smooth context, Preprint, arXiv:1302.5555 (2013).
- [G13b] N. GIGLI, An overview of the proof of the splitting theorem in spaces with non-negative Ricci curvature, Anal. Geom. Metr. Spaces 2 (2014), no. 1, 169–213.
- [G15] N. GIGLI, On the differential structure of metric measure spaces and applications, Mem. Amer. Math. Soc. **236** (2015), no. 1113.
- [G18a] N. GIGLI, Lecture Notes On Differential Calculus on RCD Spaces, Publ. Res. Inst. Math. Sci. 54 (2018), no. 4, 855–918.

- [G18b] N. GIGLI, Nonsmooth differential geometry an approach tailored for spaces with Ricci curvature bounded from below Mem. Amer. Math. Soc. 251 (2018), no. 1196.
- [GH] N. GIGLI, B. HAN, Sobolev spaces on warped products, J. Funct. Anal. **275** (2018), no. 8, 2059-2095.
- [GKO] N. GIGLI, K. KUWADA, S.-I. OHTA, *Heat flow on Alexandrov* spaces, Comm. Pure Appl. Math. **66** (2013), no. 3, 307–331.
- [GMa] N. GIGLI, F. MARCONI, A general splitting principle on RCD spaces and applications to spaces with positive spectrum, Preprint (2023).
- [GMo] N. GIGLI, A. MONDINO, A PDE approach to nonlinear potential theory in metric measure spaces, J. Math. Pures Appl. **100** (2013), no. 4, 505–534.
- [GMS] N. GIGLI, A. MONDINO, G. SAVARÉ, Convergence of pointed non-compact metric measure spaces and stability of Ricci curvature bounds and heat flows, Proc. Lond. Math. Soc. (3) 111 (2015), no. 5, 1071-1129.
- [GP20] N. GIGLI, E. PASQUALETTO, Lecture on nonsmooth differential geometry, SISSA Springer Series 2, Springer (2020).
- [GP21] N. GIGLI, E. PASQUALETTO, Behaviour of the reference measure on RCD spaces under charts, Comm. Anal. Geom. 29 (2021), no. 6, 1391–1414.
- [GP22] N. GIGLI, E. PASQUALETTO, Equivalence of two different notions of tangent bundle on rectifiable metric measure spaces, Comm. Anal. Geom. 30 (2022), no. 1, 1-51.
- [GRS] N. GIGLI, T. RAJALA, K.-T. STURM, Optimal maps and exponentiation on finite dimensional spaces with Ricci curvature bounded from below, J. Geom. Anal., **26** (2016), no. 4, 2914–2929.
- [GR18] N. GIGLI, C. RIGONI, Recognizing the flat torus among RCD*(0, N) spaces via the study of the first cohomology group, Calc. Var. Partial Differential Equations 57 (2018), no. 4, Paper No. 104.
- [GR19] N. GIGLI, C. RIGONI, A note about the strong maximum principle on RCD spaces, Canad. Math. Bull. **62** (2019), no. 2, 259–266.

[GT18]	N. GIGLI, L. TAMANINI, Second order differentiation formula on compact $RCD^*(K, N)$ spaces, Preprint, arXiv:1701.03932 (2018).
[GT21]	N. GIGLI, L. TAMANINI, Second order differentiation formula on compact $RCD^*(K, N)$ spaces, J. Eur. Math. Soc. (JEMS) 23 (2021), no. 5, 1727–1795.
[GV]	N. GIGLI, I. Y. VIOLO, Monotonicity formulas for harmonic functions in $RCD(0, N)$ spaces, J. Geom. Anal. 33 (2023), no. 3.
[Haj]	P. HAJLASZ, Sobolev spaces on an arbitrary metric space, Potential Anal. 5 (1996), 403-415.
[Han]	B. HAN, Ricci tensor on $\text{RCD}^*(K, N)$ spaces, J. Geom. Anal. 28 (2018), no. 2, 1295–1314.
[HeKo]	J. HEINONEN, P. KOSKELA, Quasiconformal maps in metric spaces with controlled geometry, Acta Math. 181 (1998), 1-61.
[HKX]	B. HUA, M. KELL, C. XIA, Harmonic functions on metric measure spaces, Preprint ArXiv:1308.3607 (2013).
[KM]	M. KELL, A. MONDINO, On the volume measure of non-smooth spaces with Ricci curvature bounded below, Ann. Sc. Norm. Super. Pisa Cl. Sci. (5) 18 (2018), no. 2, 593-610.
[K]	C. KETTERER, Cones over metric measure spaces and the max- imal diameter theorem, Journal de Mathematiques Pures et Ap- pliquees 103 (2015), pp. 1228–1275.
[LW01]	P. LI, J. WANG, Complete manifolds with positive spectrum, J. Diff. Geom. ${\bf 58}~(2001),~{\rm no.}~3,~501{-}534$.

- [LW02] P. LI, J. WANG, Complete manifolds with positive spectrum, II, J. Diff. Geom. 62 (2002), no. 1, 143–162.
- [LV] J. LOTT, C. VILLANI, *Ricci curvature for metric-measure spaces* via optimal transport, Ann. of Math. (2) **169** (2009), 903–991.
- [Mil] A. D. MILKA, Metric structure of a certain class of spaces that contain straight lines, Ukrain. Geometr. Sb. Vyp. 4 (1967), 43-48.
- [Mir] M. MIRANDA JR, Functions of bounded variation on "good" metric spaces, J. Math. Pures Appl. (9) 82 (2003), no. 8, 975–1004.

- [MN] A. MONDINO, A. NABER, Structure theory of metric measure spaces with lower Ricci curvature bounds, J. Eur. Math. Soc. (JEMS) **21** (2019), no. 6, 1809-1854.
- [OV] F. OTTO, C. VILLANI, Generalization of an inequality by Talagrand and links with the logarithmic Sobolev inequality, J. Funct. Anal. **173** (2000), no. 2, 361–400.
- [R12a] T. RAJALA, Interpolated measures with bounded density in metric spaces satisfying the curvature-dimension conditions of Sturm, J. Funct. Anal. 263 (2012), no. 4, 896-924.
- [R12b] T. RAJALA, Local Poincaré inequalities from stable curvature conditions on metric spaces, Calc. Var. Partial Differential Equations 44 (2012), no. 3-4, 477-494.
- [RS] T. RAJALA, K.-T. STURM, Non-branching geodesics and optimal maps in strong $CD(K, \infty)$ -spaces, Calc. Var. Partial Differential Equations **50** (2014), no. 3-4, 831–846.
- [Sha] N. SHANMUGALINGAM, Newtonian spaces: an extension of Sobolev spaces to metric measure spaces, Rev. Mat. Iberoam. 16 (2000), no. 2, 243–279.
- [S94] K.-T. STURM, Analysis on local Dirichlet spaces. I. Recurrence, conservativeness and L^p-Liouville properties, J. Reine und Angew. Math. 456 (1994), 173–196.
- [S06a] K.-T. STURM, On the geometry of metric measure spaces. I, Acta Math. **196** (2006), 65–131.
- [S06b] K.-T. STURM, On the geometry of metric measure spaces. II, Acta Math. **196** (2006), 133–177.
- [SvR] K.-T. STURM, M.-K. VON RENESSE, Transport inequalities, gradient estimates, entropy and Ricci curvature, Comm. Pure Appl. Math. 58 (2005), no. 7, 923–940.
- [T] V. A. TOPONOGOV, Riemannian spaces containing straight lines, Dokl. Akad. Nauk SSSR 127 (1959), 977-979.
- [V] C. VILLANI, Optimal transport: old and new, Grundlehren der Mathematischen Wissenschaften 338, Springer-Verlag, Berlin (2009).

[W]	N. WEAVER, Lipschitz algebras and derivations II. Exterior dif- ferentiation, J. Funct. Anal. 178 (2000), no. 1, 64–112.
[ZZ]	H. C. ZHANG, X. P. ZHU, Local Li-Yau's estimates on $RCD^*(K, N)$ metric measure spaces, Calc. Var. Partial Differential Equations 55, 93 (2016).

Ringraziamenti

La fine di un percorso crea sempre molte riflessioni, sia sul futuro sia sul passato, ed è in questo momento che più ci si rende conto di quante persone sono state al tuo fianco e ti hanno aiutato, persone senza le quali sarebbe stato tutto diverso. Vorrei quindi prendermi qualche riga per ringraziare le persone che più di tutti hanno contribuito a rendere questo percorso un'esperienza indimenticabile.

Per cominciare vorrei ringraziare il professor Nicola Gigli per tutto: in questi anni mi ha insegnato molto e mi ha aiutato a crescere non solo dal punto di vista accademici. Mi ha sempre supportato ed è stato infinitamente disponibile (e paziente). Inoltre mi ha offerto esperienze uniche per le quali non credo riuscirò mai a ringraziarlo abbastanza.

Un ringraziamento anche alla SISSA per l'ottima organizzazione del percorso di dottorato e per l'estrema gentilezza e disponibilità di tutto il personale nell'aiutare gli studenti in qualunque situazione.

I'd like to thank the Fields Institute in Toronto, along with its staff, all the organizers and the participants of the Thematic Program on Nonsmooth Riemannian and Lorentzian Geometry for a wonderful semester. Thanks to Elena, Tauseef and Marilou, I was very lucky to live with such a beatiful international group of friends.

Vorrei ringraziare inoltre tutti i miei colleghi e amici che ho conosciuto in questi quattro anni. In particolare: Martina, Ivan P., Simone C., Paolo, Davide, Francesco R., Francesco C., Simone V. ed Emanuele C.. Un grande grazie anche ad Ivan Y. V. per l'aiuto offerto e l'interesse mostrato nel mio progetto.

Un ringraziamento speciale agli amici con cui passo la maggior parte delle mie serate: Emanuele, Camilla, Alberto, Agno, Giacomo, Andrea e Lorenzo. Grazie di esserci sempre stati, il costante collegamento con voi, il vostro supporto e la vostra compagnia nonostante la distanza mi ha aiutato molto in questi anni.

Ringrazio inoltre la Chattina Bascé per i costanti aggiornamenti sportivi e le sgasate su Heat. Grazie al Coro Vocinsieme di Praticello, che nonostante i dieci anni passati a ore di treno di distanza mi ha sempre accolto calorosamente.

Concludo ringraziando tutta la mia numerosa famiglia, che mi ha accompagnato in tutti questi anni tra una partita a carte e l'altra. In particolare grazie ai miei genitori Loris e Meri per avermi sempre aiutato e sostenuto in tutte le mie scelte, ed Elio per aver condiviso con me sin dalle scuole medie la sua passione per la matematica insegnandomi tantissimo sempre con estrema disponibilità e chiarezza. Come in bicicletta, stare a ruota e prendere la scia rende tutto più facile.

Grazie di cuore a Stefano, Stefania ed Elena, per avermi ospitato un numero infinito di volte e fatto sempre sentire come uno di famiglia.

Per finire, i miei ringraziamenti più sentiti a Francesca, che non solo mi ha sopportato per otto anni, ma ogni giorno continua a darmi più di quanto io possa chiedere. Grazie.